# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# **FORM 10-Q**

(Mark One)			
□ QUARTERLY REPORT PURSUANT TO SECTION I	3 OR 15 (D) OF THE SECURITIES AND EX	CHANGE ACT OF 1934	
	For the quarterly period ended June 30	, 2022	
	or		
□ TRANSITION REPORT PURSUANT TO SECTION	13 OR 15 (D) OF THE SECURITIES AND EX	CHANGE ACT OF 1934	
	For the transition period fromto	)	
MAR	ATHON DIGITAL HOL		
Nevada	001-36555	01-0949984	
(State or other jurisdiction of incorporation)	(Commission File Number)	(IRS Employer Identification No.	)
	ive, Suite 100 Las Vegas, NV	89144	
(Address of princip	al executive offices)	(Zip Code)	
Indicate by check mark whether the registrant (1) has file months (or for such shorter period that the registrant was  Indicate by check mark whether the registrant has sub (§232.405 of this chapter) during the preceding 12 month  Indicate by check mark whether the registrant is a larg company. See the definitions of "large accelerated filer,"	required to file such reports), and (2) has been smitted electronically every Interactive Data F as (or for such shorter period that the registrant value accelerated filer, an accelerated filer, a non	subject to such filing requirements for the past 90 da ile required to be submitted pursuant to Rule 405 was required to submit such files). Yes ⋈ No □ -accelerated filer, smaller reporting company, or a	ys. Yes ⊠ No □  of Regulation S-7  n emerging growtl
Large Accelerated Filer		Accelerated Filer	
Non-accelerated Filer   Emerging growth company		Smaller Reporting Company	
If an emerging growth company, indicate by check mark accounting standards provided pursuant to Section 13(a).	of the Exchange Act. □		or revised financia
Indicate by check mark whether the registrant is a shell co		,	
	Securities registered pursuant to Section 12(b	o) of the Act:	
Title of each class	Trading Symbol(s)	Name of each exchange on which regis	tered
Common Stock	MARA	The Nasdaq Capital Market	
Indicate the number of shares outstanding of each of the outstanding as of August 9, 2022.	issuer's classes of common stock, as of the late	st practicable date, 116,810,405 shares of common s	stock are issued and

# TABLE OF CONTENTS

		Page
	PART I FINANCIAL INFORMATION	
Item 1.	Financial Statements	3
	Consolidated Condensed Balance Sheets as of June 30, 2022 (unaudited) and December 31, 2021	3
	Consolidated Condensed Statements of Operations for the Three and Six Months Ended June 30, 2022 and 2021 (unaudited)	4
	Consolidated Condensed Statements of Stockholders' Equity for the Three and Six Months Ended June 30, 2022 and 2021 (unaudited)	5
	Consolidated Condensed Statements of Cash Flows for the Six Months Ended June 30, 2022 and 2021 (unaudited)	6
	Notes to Unaudited Consolidated Condensed Financial Statements	7
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	18
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	24
Item 4.	Controls and Procedures	25
PART II	- OTHER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	26
Item 1A	<u>Risk Factors</u>	27
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	28
Item 3.	<u>Defaults upon Senior Securities</u>	28
Item 4.	Mine Safety Disclosures	28
Item 5.	Other Information	28
Item 6.	<u>Exhibits</u>	28

# OTHER PERTINENT INFORMATION

Unless specifically set forth to the contrary, "Marathon Digital Holdings, Inc.," "we," "us," "our" and similar terms refer to Marathon Digital Holdings, Inc., a Nevada corporation, and its subsidiaries.

# MARATHON DIGITAL HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED CONDENSED BALANCE SHEETS

		June 30, 2022		December 31, 2021		
ASSETS		(unaudited)				
Current assets:						
Cash and cash equivalents	\$	86,461,467	\$	268,522,019		
Restricted cash	Ψ	3,200,000	Ψ	200,522,017		
Digital currencies		136,836,179		102,805,980		
Digital currencies, restricted		53,558,996		-		
Digital currencies loaned		-		20,437,284		
Digital currencies held in fund		-		223,778,545		
Deposits		40,006,270		34,458,347		
Loan receivable		30,000,000		30,000,000		
Prepaid expenses and other current assets		12,128,736		8,148,016		
Total current assets		362,191,648	_	688,150,191		
Total Carreix assets		302,171,010		000,120,171		
Other assets:						
Property and equipment (net of accumulated depreciation and impairment charges of \$55,390,407 and \$21,311,461,						
respectively)		314,257,284		276,242,794		
Assets held for sale		14,758,386		-		
Advances to vendor		800,204,367		466,254,623		
Investments		16,999,823		3,000,000		
Long term prepaids				13,665,589		
Right-of-use assets		1,166,049		-		
Intangible assets (net of accumulated amortization of \$280,497 at December 31, 2021)		-		931,226		
8						
Total other assets		1 147 385 909		760 094 232		
Total other assets TOTAL ASSETS LIABILITIES AND STOCKHOLDERS' EQUITY	\$	1,147,385,909 1,509,577,557	\$	760,094,232 <b>1,448,244,423</b>		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:	<u>-</u>	1,509,577,557		1,448,244,423		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities: Accounts payable	\$	1,509,577,557 51,302,960	<b>\$</b> \$	1,448,244,423 10,772,523		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities: Accounts payable Accrued expenses	<u>-</u>	1,509,577,557 51,302,960 3,056,538		1,448,244,423 10,772,523		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:    Accounts payable    Accrued expenses    Short term borrowings - revolving credit line	<u>-</u>	51,302,960 3,056,538 35,000,000		1,448,244,423		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:    Accounts payable    Accrued expenses    Short term borrowings - revolving credit line    Operating lease liabilities	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105		1,448,244,423 10,772,523 2,154,616		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917		1,448,244,423 10,772,523 2,154,616 - 867,260		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105		1,448,244,423 10,772,523 2,154,616 - 867,260		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Convertible notes	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399 728,405,922		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities Deferred tax liabilities	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564 28,570,781		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399 728,405,922 - 23,020,721		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399 728,405,922		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities Deferred tax liabilities	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564 28,570,781		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399 728,405,922 - 23,020,721		
LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities Deferred tax liabilities Total long-term liabilities  Commitments and Contingencies	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564 28,570,781		1,448,244,423 10,772,523 2,154,616 		
LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities Deferred tax liabilities Total long-term liabilities  Commitments and Contingencies  Stockholders' Equity: Preferred stock, 0.0001 par value, 50,000,000 shares authorized, no shares issued and outstanding at June 30, 2022 and December 31, 2021, respectively	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564 28,570,781		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399 728,405,922 - 23,020,721		
LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities Deferred tax liabilities Total long-term liabilities  Commitments and Contingencies  Stockholders' Equity: Preferred stock, 0.0001 par value, 50,000,000 shares authorized, no shares issued and outstanding at June 30, 2022	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564 28,570,781		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399 728,405,922 - 23,020,721		
LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:     Accounts payable     Accrued expenses     Short term borrowings - revolving credit line     Operating lease liabilities     Accrued interest     Total current liabilities Long-term liabilities Long-term liabilities Convertible notes Operating lease liabilities Deferred tax liabilities Total long-term liabilities  Commitments and Contingencies  Stockholders' Equity: Preferred stock, 0.0001 par value, 50,000,000 shares authorized, no shares issued and outstanding at June 30, 2022 and December 31, 2021, respectively	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564 28,570,781		1,448,244,423  10,772,523 2,154,616 867,260 13,794,399 728,405,922 23,020,721 751,426,643		
TOTAL ASSETS  LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities: Accounts payable Accrued expenses Short term borrowings - revolving credit line Operating lease liabilities Accrued interest Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities Total long-term liabilities  Convertible notes Operating lease liabilities Total long-term liabilities  Commitments and Contingencies  Stockholders' Equity: Preferred stock, 0.0001 par value, 50,000,000 shares authorized, no shares issued and outstanding at June 30, 2022 and December 31, 2021, respectively Common stock, 0.0001 par value; 200,000,000 shares authorized; 113,865,235 and 102,733,273 issued and outstanding at June 30, 2022 and December 31, 2021, respectively Additional paid-in capital	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564 28,570,781 759,985,038		1,448,244,423  10,772,523 2,154,616 867,260 13,794,399 728,405,922 23,020,721 751,426,643		
LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities: Accounts payable Accrued expenses Short term borrowings - revolving credit line Operating lease liabilities Accrued interest Total current liabilities Long-term liabilities Convertible notes Operating lease liabilities Total long-term liabilities  Convertible notes Operating lease liabilities Total long-term liabilities Commitments and Contingencies	<u>-</u>	51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520 730,347,693 1,066,564 28,570,781 759,985,038		1,448,244,423  10,772,523 2,154,616 867,260 13,794,399 728,405,922 23,020,721 751,426,643		
LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities: Accounts payable Accrued expenses Short term borrowings - revolving credit line Operating lease liabilities Accrued interest Total current liabilities Convertible notes Operating lease liabilities Convertible notes Operating lease liabilities Total long-term liabilities Convertible notes Operating lease liabilities Total long-term liabilities  Commitments and Contingencies  Stockholders' Equity: Preferred stock, 0.0001 par value, 50,000,000 shares authorized, no shares issued and outstanding at June 30, 2022 and December 31, 2021, respectively Common stock, 0.0001 par value; 200,000,000 shares authorized; 113,865,235 and 102,733,273 issued and outstanding at June 30, 2022 and December 31, 2021, respectively Additional paid-in capital	<u>-</u>	1,509,577,557  51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520  730,347,693 1,066,564 28,570,781 759,985,038		1,448,244,423  10,772,523 2,154,616 867,260 13,794,399 728,405,922 23,020,721 751,426,643  10,273 835,693,610 (450,719		
LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities: Accounts payable Accrued expenses Short term borrowings - revolving credit line Operating lease liabilities Accrued interest Total current liabilities Convertible notes Operating lease liabilities Convertible notes Operating lease liabilities Total long-term liabilities  Commitments and Contingencies  Stockholders' Equity: Preferred stock, 0.0001 par value, 50,000,000 shares authorized, no shares issued and outstanding at June 30, 2022 and December 31, 2021, respectively Common stock, 0.0001 par value, 200,000,000 shares authorized; 113,865,235 and 102,733,273 issued and outstanding at June 30, 2022 and December 31, 2021, respectively Additional paid-in capital Accumulated other comprehensive loss	<u>-</u>	1,509,577,557  51,302,960 3,056,538 35,000,000 162,105 622,917 90,144,520  730,347,693 1,066,564 28,570,781 759,985,038		1,448,244,423 10,772,523 2,154,616 - 867,260 13,794,399 728,405,922 - 23,020,721		

Certain prior period amounts have been reclassified to conform to current period presentation.

The accompanying notes are an integral part to these unaudited consolidated condensed financial statements.

# MARATHON DIGITAL HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED CONDENSED STATEMENT OF OPERATIONS (unaudited)

		Three Months	June 30,	Six Months Ended June 30,			une 30,	
		2022		2021		2022		2021
Total Revenues		24,921,816	\$	29,321,857	\$	76,639,534	\$	38,474,672
Costs and expenses								
Cost of revenues								
Cost of revenues - energy, hosting and other		(16,684,759)		(4,056,168)		(29,201,710)		(5,724,646)
Cost of revenues - depreciation and amortization		(24,709,797)		(2,937,666)		(38,586,480)		(3,675,603)
		(41,394,556)		(6,993,834)		(67,788,190)		(9,400,249)
Operating expenses								
General and administrative expenses		(12,641,331)		(6,831,040)		(26,835,089)		(60,175,421)
Impairment of digital currencies		(127,590,231)		(11,078,660)		(147,141,486)		(11,740,859)
Impairment of patents		-		-		(919,363)		-
		(140,231,562)		(17,909,700)		(174,895,938)		(71,916,280)
Other Operating income (expenses)								
Change in fair value of digital currencies held in fund		(79,688,590)		(114,704,596)		(85,016,208)		17,323,121
Gain on sale of equipment		58,181,516		-		58,181,516		-
		(21,507,074)		(114,704,596)		(26,834,692)		17,323,121
Operating income (loss)		(178,211,376)		(110,286,273)		(192,879,286)		(25,518,736)
Non-Operating income (expenses)		165,280		1,400,872		393,973		(7,250)
Interest expense		(3,748,322)		(1,203)		(6,562,358)		(2,406)
Loss before income taxes	\$	(181,794,418)	\$	(108,886,604)	\$	(199,047,671)	\$	(25,528,392)
Income tax (expense) benefit		(9,852,224)		1,984		(5,557,560)		514
Net loss	\$	(191,646,642)	\$	(108,884,620)	\$	(204,605,231)	\$	(25,527,878)
	_		_					
Net loss per share, basic and diluted:	\$	(1.75)	\$	(1.09)	\$	(1.93)	\$	(0.26)
Weighted average shares outstanding, basic and diluted:		109,437,293		99,466,946	_	106,101,762		96,922,964

Certain prior period amounts have been reclassified to conform to current period presentation . The accompanying notes are an integral part to these unaudited consolidated condensed financial statements.

# MARATHON DIGITAL HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED CONDENSED STATEMENTS OF STOCKHOLDERS' EQUITY

(Unaudited)

# For the Six Months Ended June 30, 2021

	Preferr	ed Stocl	k	Commo	n Stoo	ck	Additional Paid-in	Accumulated	cumulated Other nprehensive	Total Stockholders'
	Number	Am	ount	Number	A	mount	Capital	Deficit	Loss	Equity
Balance as of December 31, 2020		\$	_	81,974,619	\$	8,197	\$ 428,242,763	\$ (116,055,277)	\$ (450,719)	\$ 311,744,964
Stock based compensation, net of tax										
withholding	-		-	4,800,962		480	51,907,098	-	-	51,907,578
Issuance of common stock, net of offering										
costs/At-the-market offering	-		-	12,500,000		1,250	237,428,370	-	-	237,429,620
Options exercised for cash	-		-	23,500		3	(3)	-	-	-
Warrant exercised for cash	-		-	170,904		17	160,145	-	-	160,162
Common stock issued for cashless exercise of										
warrants	-			2,044		-	-			-
Common stock issued for service and license										
agreements	-			162,094		16	4,804,823			4,804,839
Net loss	-		-	-		-	-	(25,527,878)	-	(25,527,878)
Balance as of June 30, 2021		\$	_	99,634,123	\$	9,963	\$ 722,543,196	\$ (141,583,155)	\$ (450,719)	\$ 580,519,285

# For the Six Months Ended June 30, 2022

										Ac	cumulated		
								Additional			Other	Total	
	Preferre	ed Stoc	k	Common Sto		n Stock		Paid-in	Accumulated	Comprehensive		Stockholders'	
	Number	Am	ount	Number	A	mount		Capital	Deficit		Loss	Equity	
Balance as of December 31, 2021		\$	_	102,733,273	\$	10,273	\$	835,693,610	\$ (152,229,783)	\$	(450,719)	\$ 683,023,381	
Stock based compensation, net of tax													
withholding	-		-	375,730		38		15,407,538	-		-	15,407,576	
Issuance of common stock, net of offering													
costs/At-the-market offering	-		-	10,556,232		1,056		161,041,218	-		-	161,042,274	
Common stock issued for long term service													
contract	-		-	200,000		20		4,579,979	-		-	4,579,999	
Net loss	-		-	-		-		-	(204,605,231)		-	(204,605,231)	
Balance as of June 30, 2022	_	\$	-	113,865,235	\$	11,387	\$	1,016,722,345	\$ (356,835,014)	\$	(450,719)	\$ 659,447,999	

# For the Three Months Ended June 30, 2021

									Ac	cumulated		
							Additional			Other	Total	
	Preferre	ed Stock	ζ.	Common Stock		k	Paid-in	Accumulated	Comprehensive		Stockholders'	
	Number	Am	ount	Number	A	mount	Capital	Deficit		Loss	Equity	
Balance as of March 31, 2021		\$	-	99,370,465	\$	9,937	\$ 716,862,400	\$ (32,698,535)	\$	(450,719)	\$ 683,723,083	
Stock based compensation, net of tax												
withholding	-		-	99,520		10	875,973	-		-	875,983	
Common stock issued for cashless exercise of												
warrants	-		-	2,044		-	-	-		-	-	
Common stock issued for service and license												
agreements	-		-	162,094		16	4,804,823	-		-	4,804,839	
Net loss	-		-	-		-	-	(108,884,620)		-	(108,884,620)	
Balance as of June 30, 2021		\$	-	99,634,123	\$	9,963	\$ 722,543,196	\$ (141,583,155)	\$	(450,719)	\$ 580,519,285	

# For the Three Months Ended June 30, 2022

	Preferre		Commo			Additional Paid-in	Accumulated	 Other mprehensive	Total Stockholders'
	Number	Amount	Number	P	Amount	Capital	Deficit	 Deficit	Equity
Balance as of March 31, 2022	-	\$	- 106,051,713	\$	10,605	\$ 939,741,806	\$ (165,188,372)	\$ (450,719)	\$ 774,113,320
Stock based compensation, net of tax									
withholding	-		- 256,934		26	6,132,198	-	-	6,132,224
Issuance of common stock, net of offering									
costs/At-the-market offering	-		- 7,556,588		756	70,848,341	-	-	70,849,097
Common stock issued for long term service									
contract	-				-	-	-	-	-
Net loss	-				-	-	(191,646,642)	-	(191,646,642)
Balance as of June 30, 2022	-	\$	- 113,865,235	\$	11,387	\$ 1,016,722,345	\$ (356,835,014)	\$ (450,719)	\$ 659,447,999

The accompanying notes are an integral part to these unaudited consolidated condensed financial statements.

# MARATHON DIGITAL HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED CONDENSED STATEMENTS OF CASH FLOWS

(unaudited)

		Six Months Ended June 30,				
		2022		2021		
CASH FLOWS FROM OPERATING ACTIVITIES						
Net income (loss)	\$	(204,605,231)	\$	(25,527,878)		
Adjustments to reconcile net income (loss) to net cash used in operating activities:						
Depreciation and amortization		38,586,480		3,675,603		
Amortization of prepaid service contract		15,533,541		1,122,000		
Gain on sale of assets		(58,181,516)		-		
Deferred tax expense		5,550,060		-		
Change in fair value of digital currencies held in fund		85,016,208		(17,323,121)		
Impairment of digital currencies		147,141,486		11,740,859		
Stock based compensation		15,451,474		55,717,561		
Amortization of bond issuance costs		1,941,771		-		
Impairment of patents		919,363		-		
Other adjustments from operations, net		498,324		859,212		
Changes in operating assets and liabilities:						
Digital currencies		(76,449,636)		(38,474,672)		
Deposits		(5,547,923)		-		
Prepaid expenses and other assets		(1,268,673)		(167,906)		
Accounts payable and accrued expenses		(5,180,641)		1,626,500		
Accrued interest		(244,343)		-		
Net cash used in operating activities		(40,839,256)		(6,751,842)		
CASH FLOWS FROM INVESTING ACTIVITIES		(10,000,000)		(0,100,000)		
Advances to vendor		(393,991,125)		(55,935,273)		
Purchase of property and equipment		(13,751,596)		(66,566,839)		
Sale of property and equipment		87,240,000		(00,500,057)		
Purchase of digital currencies held in fund		07,240,000		(150,000,000)		
Purchase of equity investments		(13,999,823)		(150,000,000)		
Sale of digital currencies in investment fund		482,872				
Net cash used in investing activities		(334,019,672)		(272.502.112)		
		(334,019,672)		(272,502,112)		
CASH FLOWS FROM FINANCING ACTIVITIES		161 040 054		212 106 015		
Proceeds from issuance of common stock, net of issuance costs		161,042,274		312,196,845		
Net change in revolving credit agreement borrowings		35,000,000		-		
Value of shares withheld for taxes		(43,898)		(3,809,983)		
Proceeds received on exercise of options and warrants		<u>-</u>		160,163		
Net cash provided by financing activities		195,998,376		308,547,025		
Net (decrease) increase in cash, cash equivalents and restricted cash		(178,860,552)		29,293,071		
Cash, cash equivalents and restricted cash — beginning of period		268,522,019		141,322,776		
Cash, cash equivalents and restricted cash — end of period	\$	89,661,467	\$	170,615,847		
Supplemental schedule of non-cash investing and financing activities:						
Receivable due to share issuance	\$	4,720,197	\$	-		
Options exercised into common stock	\$	-	\$	3		
Unpaid advances to vendor	\$	46,613,000	\$			
ī	<b>3</b>	, ,				
Operating lease assets obtained in exchange for new operating lease liabilities	\$	1,420,370	\$	-		
Common stock issued for service and license agreements	\$	4,579,999	\$	4,804,839		
	·	<del></del>	_			

Certain prior period amounts have been reclassified to conform to current period presentation.

The accompanying notes are an integral part to these unaudited consolidated condensed financial statements.

## NOTE 1 - ORGANIZATION AND DESCRIPTION OF BUSINESS

Marathon Digital Holdings, Inc. (the "Company") was incorporated in the State of Nevada on February 23, 2010 under the name Verve Ventures, Inc. On December 7, 2011, the Company changed its name to American Strategic Minerals Corporation and was engaged in exploration and potential development of a minerals business. In June 2012, the Company discontinued the minerals business and began to invest in real estate properties in Southern California. In October 2012, the Company discontinued its real estate business and the Company commenced IP licensing operations, at which time the Company's name was changed to Marathon Patent Group, Inc. The Company changed its name to Marathon Digital Holdings, Inc. on March 1, 2021. In 2018, the Company began its bitcoin mining operations by purchasing cryptocurrency mining machines and establishing a data center in Canada to mine digital assets. The Company ceased operating in Canada in 2020 and relocated all owned mining equipment out of Canada to the US. The Company has since expanded its activities in the mining of bitcoin. As of June 30, 2022, the Company no longer holds any legacy IP assets and is solely focused on the mining of bitcoin and ancillary opportunities within the bitcoin ecosystem.

# NOTE 2 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The accompanying condensed consolidated financial statements are unaudited and have been prepared in accordance with the rules and regulations of the SEC. They include all adjustments that we consider necessary for a fair statement of the results for the interim periods presented. Such adjustments consisted only of normal recurring items unless otherwise disclosed. The June 30, 2022, Condensed Consolidated Balance Sheet was derived from audited financial statements but does not include all footnote disclosures from the annual financial statements.

These financial statements should be read in conjunction with the financial statements and related notes included in the Company's 2021 Annual Report.

## **Basis of Presentation and Principles of Consolidation**

The accompanying unaudited consolidated condensed financial statements, including the accounts of the Company's subsidiaries, Marathon Crypto Mining, Inc., Crypto Currency Patent Holding Company and Soems Acquisition Corp. have been prepared by the Company, without audit, pursuant to the rules and regulations of the Securities and Exchange Commission (SEC). Certain information and disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States of America (GAAP) have been condensed or omitted pursuant to such rules and regulations. These consolidated condensed financial statements reflect all adjustments (consisting only of normal recurring adjustments) which, in the opinion of management, are necessary to present fairly the financial position, the results of operations and cash flows of the Company for the periods presented. It is suggested that these consolidated condensed financial statements be read in conjunction with the consolidated financial statements and the notes thereto included in the Company's most recent Annual Report on Form 10-K. The results of operations for the interim periods are not necessarily indicative of the results to be expected for the full year ended December 31, 2022.

#### **Use of Estimates and Assumptions**

The preparation of financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. Significant estimates made by management include, but are not limited to, estimating the useful lives of fixed assets, the assumptions used to calculate fair value of options granted, realization of long-lived assets, deferred income taxes, unrealized tax positions and the realization of digital currencies.

# Restricted Cash

Restricted cash principally represents those cash balances that support commercial letters of credit and are restricted from withdrawal. The following table provides a reconciliation of the total cash, cash equivalents and restricted cash reported on the Condensed Consolidated Balance Sheets to the corresponding amounts reported on the Condensed Consolidated Statements of Cash Flows.

		As of	As of
	Ju	ne 30, 2022	June 30, 2021
Cash and cash equivalents	\$	86,461,467	\$ 170,615,847
Restricted cash		3,200,000	-
Cash, cash equivalents and restricted cash	\$	89,661,467	\$ 170,615,847

#### Reclassifications and corrections

For purposes of comparability, certain prior-period amounts have been reclassified to conform to the current-period presentation, including corrections of immaterial errors in prior periods.

## Digital Currencies, Digital currencies, restricted and Digital currencies loaned

Digital currencies, Digital currencies, restricted and Digital currencies loaned are included in current assets in the consolidated balance sheets. Digital currencies are recorded at cost less impairment.

An intangible asset with an indefinite useful life is not amortized but assessed for impairment annually, or more frequently, when events or changes in circumstances occur indicating that it is more likely than not that the indefinite-lived asset is impaired. Impairment exists when the carrying amount exceeds its fair value. In testing for impairment, the Company has the option to first perform a qualitative assessment to determine whether it is more likely than not that an impairment exists. If it is determined that it is not more likely than not that an impairment exists, a quantitative impairment test is not necessary. If the Company concludes otherwise, it is required to perform a quantitative impairment test. To the extent an impairment loss is recognized, the loss establishes the new cost basis of the asset. Subsequent reversal of impairment losses is not permitted.

The following table presents the activities of the digital currencies for the six months ended June 30, 2022:

Digital currencies, Digital currencies, restricted and Digital currencies loaned at December	
31, 2021*	\$ 123,243,264
Additions of digital currencies	76,449,636
Digital currencies transferred from fund	137,843,761
Impairment of digital currencies	(147,141,486)
Digital currencies, Digital currencies, restricted and Digital currencies loaned at June 30,	
2022	\$ 190,395,175

<sup>\*</sup> Includes a loan of digital currencies of 600 bitcoin (\$20,437,284). On June 14, 2022 the Company terminated the loan and there are no loans of digital assets outstanding as of June 30, 2022.

At June 30, 2022, we held approximately 10,055 bitcoin with a carrying value of \$190.4 million and carried on the balance sheet as digital currencies (\$136.8 million) and digital currencies, restricted (\$53.6 million). The fair market value of the bitcoin as of June 30, 2022 was approximately \$198.9 million.

Halving – The bitcoin blockchain and the cryptocurrency reward for solving a block is subject to periodic incremental halving. Halving is a process designed to control the overall supply and reduce the risk of inflation in cryptocurrencies using a Proof-of-Work consensus algorithm. At a predetermined block, the mining reward is cut in half, hence the term "Halving". The last halving for bitcoin occurred on May 12, 2020. For example, the current fixed reward on the bitcoin network for solving a new block is six and one quarter (6.25) bitcoins per block, which decreased from twelve and a half (12.5) bitcoins per block in May 2020. It is estimated that the number of bitcoins per block will halve again in about four (4) years. Many factors influence the price of bitcoin and potential increases or decreases in prices in advance of or following a future halving is unknown.

## **Digital Currencies Held in Fund**

In 2016, the FASB issued Accounting Standards Update (ASU) 2016-01, Financial Instruments — Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, that requires entities to generally measure investments in equity securities at fair value and recognize changes in fair value in net income.

On January 25, 2021, the Company entered into a limited partnership agreement with NYDIG Digital Assets Fund III, LP ("Fund") whereas the Fund purchased 4,812.66 bitcoin in an aggregate purchase price of \$150 million. The Company owns 100% of the limited partnership interest and consolidates the Fund under a voting interest model. The consolidated assets in the investment fund are included in current assets in the consolidated balance sheets under the caption "Digital currencies held in investment fund."

The Fund qualifies and operates as an investment company for accounting purposes pursuant to the accounting and reporting guidance under ASC 946, Financial Services – Investment Companies, which requires fair value measurement of the Fund's investments in digital assets. The digital assets held by the Fund are traded on a number of active markets globally, including the over the counter ("OTC") market and digital asset exchanges. A fair value measurement under ASC 820 for an asset assumes that the asset is exchanged in an orderly transaction between market participants either in the principal market for the asset or, in the absence of a principal market, the most advantageous market for the asset (ASC 820-10-35-5). The fair value of the assets within the Fund are determined at the end of each reporting period based on pricing obtained from CoinDesk Bitcoin Price Index at approximately 4pm New York time. Any changes in the fair value of the assets are recorded in the Consolidated Statement of Operations under the caption "Change in fair value of investment in NYDIG fund." The Company transferred all of its bitcoin holdings from the Fund to its own account on June 10, 2022.

Digital currencies held in fund at December 31, 2021	\$ 223,778,545
Sale of digital currencies	(482,872)
Change in fair value of digital currencies held in fund	(85,016,208)
Management expenses incurred by fund	(435,704)
Digital currencies transferred out of fund	(137,843,761)
Digital currencies held in fund at June 30, 2022	\$ -

# Investments

Investments, which may be made from time to time for strategic reasons (and not to engage in the business of investments) are included in non-current assets in the consolidated balance sheets. Investments are recorded at cost and the Company analyzes these investments value on a quarterly basis. As part of the Company's policy to maximize return on strategic investment opportunities, while preserving capital and limiting downside risk, the Company may at times enter into equity investments or SAFE agreements. The nature and timing of the Company's investments will depend on available capital at any particular time and the investment opportunities identified and available to the Company.

On December 21, 2021 and December 30, 2021, the Company entered into two separate Simple Agreement for Future Equity ("SAFE") agreements classified on the balance sheet as non-current assets. The SAFE agreements are accounted for as equity securities without readily determinable fair value at cost minus impairment, as adjusted for observable price changes in orderly transactions for identical or similar investment of the same issue pursuant to Topic 321 *Investments – Equity Securities*. The investment in SAFE agreements is presented on the balance sheet at June 30, 2022 and December 31, 2021 as a component of the-caption "Investments" at a collective carrying value of \$6.5 million, equal to their purchased amounts with no noted impairments or adjustments.

## Fair Value of Financial Instruments

The Company measures at fair value certain of its financial and non-financial assets and liabilities by using a fair value hierarchy that prioritizes the inputs to valuation techniques used to measure fair value. Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date, essentially an exit price, based on the highest and best use of the asset or liability. The levels of the fair value hierarchy are:

- Level 1: Observable inputs such as quoted market prices in active markets for identical assets or liabilities
- Level 2: Observable market-based inputs or unobservable inputs that are corroborated by market data
- Level 3: Unobservable inputs for which there is little or no market data, which require the use of the reporting entity's own assumptions.

The carrying amounts reported in the consolidated balance sheet for cash, accounts receivable, accounts payable, and accrued expenses, approximate their estimated fair market value based on the short-term maturity of these instruments. The carrying value of notes payable and other long-term liabilities approximate fair value as the related interest rates approximate rates currently available to the Company.

Financial assets and liabilities are classified in their entirety within the fair value hierarchy based on the lowest level of input that is significant to their fair value measurement. The Company measures the fair value of its marketable securities and investments by taking into consideration valuations obtained from third-party pricing sources. The pricing services utilize industry standard valuation models, including both income and market-based approaches, for which all significant inputs are observable, either directly or indirectly, to estimate fair value. These inputs included reported trades of and broker-dealer quotes on the same or similar securities, issuer credit spreads, benchmark securities and other observable inputs.

The following tables present information about the Company's assets and liabilities measured at fair value on a recurring basis and the Company's estimated level within the fair value hierarchy of those assets and liabilities as of June 30, 2022 and December 31, 2021, respectively:

	Fair value measured at June 30, 2022				
		Total carrying value at June 30, 2022	Quoted prices in active markets (Level 1)	Significant other observable inputs (Level 2)	Significant unobservable inputs (Level 3)
Assets		_			
Money Market Accounts	\$	82,884,715	\$ 82,884,715	\$ -	\$ -
		Fair	r value measured at l	December 31, 2021	
		carrying value at December 31, 2021	Quoted prices in active markets (Level 1)	Significant other observable inputs (Level 2)	Significant unobservable inputs (Level 3)
Assets					
Money Market Accounts	\$	266,635,158	\$ 266,635,158	\$ -	\$ -

There were no transfers among Levels 1, 2 or 3 during the three and six months ended June 30, 2022.

Digital currencies held in fund

On June 10, 2022 the company withdrew approximately 4,769 bitcoin from its investment in NYDIG Digital Assets Fund III, LP, the ("Investment Fund") and transferred the bitcoin directly into the Company's account. As a result, the Company will no longer receive "mark-to-market" accounting for the bitcoin formerly held in the Investment Fund and the 4,769 bitcoin will now be classified as "Digital currencies" on the balance sheet and subject to impairment analysis as a indefinite-lived intangible.

223,778,545

223,778,545

## Net Income and Basic and Diluted Net Income per Share

Net income per common share is calculated in accordance with ASC Topic 260: Earnings Per Share ("ASC 260"). Basic income per share is computed by dividing net income (loss) by the weighted average number of shares of common stock outstanding during the period. For the six month period ending June 30, 2022, the Company incurred a loss position and as such the computation of diluted net income (loss) per share does not include dilutive common stock equivalents in the weighted average shares outstanding, as they would be anti-dilutive.

Computation of potential shares for the diluted earnings (loss) per share calculation at June 30, 2022 and 2021 are as follows:

	As of June	30,
	2022	2021
Warrants to purchase common stock	324,375	457,837
Restricted stock	1,063,410	199,038
Options to purchase common stock	-	81,120
Convertible notes to exchange common stock	9,812,955	-
Total	11,200,740	737,995

The following table sets forth the computation of basic and diluted income (loss) per share:

	For the Three Months Ended June 30,			For the Six Months Ended June 30,				
	<u></u>	2022		2021		2022		2021
Net loss attributable to common shareholders	\$	(191,646,642)	\$	(108,884,620)	\$	(204,605,231)	\$	(25,527,878)
Denominator:								
Weighted average common shares - basic and diluted		109,437,293		99,466,946		106,101,762		96,922,964
Loss per common share - basic and diluted	\$	(1.75)	\$	(1.09)	\$	(1.93)	\$	(0.26)

#### NOTE 3 – REVENUES FROM CONTRACTS WITH CUSTOMERS

The Company recognizes revenue under ASC 606, Revenue from Contracts with Customers. The core principle of the revenue standard is that a company should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the company expects to be entitled in exchange for those goods or services. The following five steps are applied to achieve that core principle:

- Step 1: Identify the contract with the customer
- Step 2: Identify the performance obligations in the contract
- Step 3: Determine the transaction price
- Step 4: Allocate the transaction price to the performance obligations in the contract
- Step 5: Recognize revenue when the Company satisfies a performance obligation

In order to identify the performance obligations in a contract with a customer, a company must assess the promised goods or services in the contract and identify each promised good or service that is distinct. A performance obligation meets ASC 606's definition of a "distinct" good or service (or bundle of goods or services) if both of the following criteria are met:

- The customer can benefit from the good or service either on its own or together with other resources that are readily available to the customer (i.e., the good or service is capable of being distinct), and
- the entity's promise to transfer the good or service to the customer is separately identifiable from other promises in the contract (i.e., the promise to transfer the good or service is distinct within the context of the contract).

If a good or service is not distinct, the good or service is combined with other promised goods or services until a bundle of goods or services is identified that is distinct.

The transaction price is the amount of consideration to which an entity expects to be entitled in exchange for transferring promised goods or services to a customer. The consideration promised in a contract with a customer may include fixed amounts, variable amounts, or both. When determining the transaction price, an entity must consider the effects of all of the following:

- Variable consideration
- Constraining estimates of variable consideration
- The existence of a significant financing component in the contract
- Noncash consideration
- Consideration payable to a customer

Variable consideration is included in the transaction price only to the extent that it is probable that a significant reversal in the amount of cumulative revenue recognized will not occur when the uncertainty associated with the variable consideration is subsequently resolved. The transaction price is allocated to each performance obligation on a relative standalone selling price basis. The transaction price allocated to each performance obligation is recognized when that performance obligation is satisfied, at a point in time or over time as appropriate.

Providing computing power in bitcoin transaction verification services to the network is the only performance obligation under our arrangements with the network. The transaction consideration the Company receives, if any, is noncash consideration, which the Company measures at fair value on the date received, which is not materially different than the fair value at the time of contract inception. The consideration is all variable. Because it is not probable that a significant reversal of cumulative revenue will not occur, the consideration is constrained until the Company successfully places a block (by being the first to solve an algorithm) and the Company receives confirmation of the consideration it will receive, at which time revenue is recognized. There is no significant financing component in these transactions.

Fair value of the digital asset award received is determined using the daily closing U.S. dollar spot rate of the related digital currency on the date of receipt.

Expenses associated with running the digital currency mining business, such as rent and electricity cost are also recorded as cost of revenues. Depreciation on digital currency mining equipment is recorded as a component of cost of revenues.

#### **Block rewards**

Block rewards earned by a bitcoin miner are recognized as revenue, but the evaluation is required to determine if the block rewards earned should be recognized as revenue from contracts with customers under FASB ASC 606 or as other revenue.

The Company evaluated whether its mining activities represent a contract with a customer to provide services and, determined it should recognize block rewards it receives from the network as revenue from a customer under FASB ASC 606. All relevant facts and circumstances, including the network's protocols, were considered in determining (1) whether the Company has a contract with a customer under FASB ASC 606-10-25-2 and (2) whether its mining activities on the network meet all the criteria in FASB ASC 606-10-25-1.

The inflow of bitcoin as a result of the block reward would meet the definition of revenue because it gives rise to economic benefits to the miner from rendering services or carrying out activities.

Therefore, the Company may account for the block reward as revenue.

Block rewards are the Company's most significant source of revenue. Block rewards included in revenues on the statements of operations were approximately \$24.5 million and \$26.6 million, respectively for the three months ended June 30, 2022 and June 30, 2021. Block rewards included in revenues on the statements of operations were approximately and \$75.6 and \$34.8 million for the six months ended June 30, 2022 and June 30, 2021.

## **Transaction Fees**

Transaction fees earned by the Company are recognized as revenue from customers in accordance with FASB ASC 606 and pursuant to AICPA Practice Guide "Accounting for and Auditing Digital Assets". The transaction fees are specified in each transaction request and paid by the requester to the Company, acting as the successful miner, in exchange for the successful processing of the transaction.

The requester meets the definition of a customer in FASB ASC 606 because it has contracted with the miner to obtain a service (successful mining) that is an output of the miner's ordinary activities in exchange for consideration. A contract with a customer exists at the point when the miner successfully validates a requesting customer's transaction to the distributed ledger. At this point, the performance obligation has been satisfied in accordance with FASB ASC 606-10-25-30. Because of this, the additional criteria in FASB ASC 606-10-25-1 would be met as follows:

- Both the requester (a customer) and the miner have approved the contract and are committed to the transaction at the point of successfully validating and adding the transaction to the distributed ledger.
- Each party's rights, the consideration to be transferred, and the payment terms are clear.
- The transaction has commercial substance (that is, the risk, timing, or amount of the miner's future cash flows is expected to change as a result of the contract).
- Collection of the fees is probable because it is completed as part of closing a successful block.

By successfully mining a block, the miner satisfies its performance obligation to the requester and, thus, should recognize revenue at that point in time.

The payment of transaction fees in bitcoin constitutes non-cash consideration under FASB ASC 606-10-32-21. This non-cash consideration is measured at its estimated fair value at contract inception - that is, the date that the criteria in FASB ASC 606-10-25-1 are met. If fair value cannot be reasonably estimated in accordance with FASB ASC 606-10-32-22, the consideration should be measured indirectly by reference to the stand-alone selling price of the miner's services.

Transaction fees were approximately \$1.0 million and \$0.3 million for the six and three months ended June 30, 2022, respectively and \$3.6 million and \$2.7 million for the six and three months ended June 30, 2021, respectively.

#### **Pool Fees**

The Company is a pool operator and acts as an agent, and not as a principal. The Company did not have control over any third party contributing hashrate to its pool. It merely facilitated the contribution of hash rate by third party pool participants who could choose to join or leave a pool as they wish. As the pool operator, the Company recognized 100% of all pool fees generated by such pool as fee revenue and not mining revenue. The Company therefore concluded that in its capacity as the pool operator it was an agent, and not a principal.

From May 2021 until April 30, 2022, the Company operated a mining pool that included certain third parties. Pool fees included in revenues on the statements of operations were approximately \$76 thousand and \$89 thousand, respectively for the three months ended June 30, 2022 and June 30, 2021. Pool fees included in revenues on the statements of operations were approximately \$331 thousand and \$89 thousand, respectively for the six months ended June 30, 2022 and June 30, 2021. As of April 30, 2022, third party miners were no longer participating in the Company's mining pool. As such, the Company will no longer recognize pool fees.

#### NOTE 4 – ADVANCES TO VENDORS AND DEPOSITS

The Company contracts with bitcoin mining server manufacturers in procuring equipment necessary for the operation of its bitcoin mining operations. A typical agreement calls for a certain percentage of the total order to be paid in advance at specific intervals, usually (1) within several days of execution of a specific contract (2) approximately six months before each shipment date and (3) approximately one month before each shipment date. We account for these payments as Advances to vendor on the balance sheet.

As of June 30, 2022 and December 31, 2021, such advances totalled approximately \$800.2 million and \$466.3 million, respectively. At June 30, 2022, the company had a payable of \$46.6 million related to the accrual of an advance to a vendor that was subsequently approved for payment and paid in early July.

In addition, the Company contracts with other service providers for hosting of its equipment and operational support in data centers where the company's equipment is deployed. These arrangements also call for advance payments to be made to vendors in conjunction with the contractual obligations associated with these services. We classify these payments as deposits on the balance sheet.

#### NOTE 5 - PROPERTY AND EQUIPMENT

The components of property and equipment as of June 30, 2022 and December 31, 2021 are:

	Useful life (Years)	June 30, 2022	December 31, 2021
Website	7	273,122	121,787
Mining equipment	5	186,608,915	163,868,283
Construction in Progress	N/A	182,765,654	133,565,908
Mining patent	17	-	1,210,000
Gross property, equipment and intangible assets		369,647,691	298,765,978
Less: Accumulated depreciation and amortization		(55,390,407)	(21,591,958)
Property, equipment and intangible assets, net		\$ 314,257,284	\$ 277,174,020

The Company's depreciation expense related to property and equipment for the three and six months ended June 30, 2022 and June 30, 2021 was \$24,701,111 and \$38,565,242, and \$2,937,666 and \$3,675,603, respectively. Amortization expense for the three and six months ended June 30, 2022 and June 30, 2021 was \$8,686 and \$21,238, and \$17,794 and \$35,588, respectively.

## NOTE 6 - ASSETS HELD FOR SALE

On December 2, 2021, we entered into an agreement with DCRBN Ventures Development and Acquisition LLC ("DCRBN") in which the Company agreed to sell certain equipment to DCRBN starting in April 2022, in conjunction with the development of commercial activities at the King Mountain wind farm in McCamey, TX. During the three months ended June 30, 2022, the Company sold equipment for cash proceeds totalling \$87.2 million and realized a pre-tax gain on the sale of such assets of \$58.2 million. There were no such sales in the prior-year period. As of June 30, 2022, the third and final batch of equipment was to be sold subsequent to quarter end and as such, classified as assets held for sale on the balance sheet.

# NOTE 7 - STOCKHOLDERS' EQUITY

## Common Stock

Shelf Registration Statements on Form S-3 and At The Market Offering Agreements

On February 11, 2022, we entered into an At The Market Offering Agreement, or sales agreement, with H.C. Wainwright & Co., LLC relating to shares of our common stock. In accordance with the terms of the sales agreement, we may offer and sell shares of our common stock having an aggregate offering price of up to \$750,000,000 from time to time through Wainwright acting as our sales agent. As of June 30, 2022, the Company had sold 10,556,232 shares of common stock for an aggregate purchase price of \$161.0 million net of offering costs pursuant to this At The Market Offering Agreement.

## Series B Convertible Preferred Stock

As of June 30, 2022, there were no shares of Series B Convertible Preferred Stock outstanding.

# Series E Preferred Stock

There were no shares of Series E Convertible Preferred Stock outstanding as of June 30, 2022.

## **Common Stock Warrants**

A summary of the status of the Company's outstanding stock warrants and changes during the six months ended June 30, 2022 is as follows:

	Number of Warrants	eighted Average Exercise Price	Weighted Average Remaining Contractual Life (in years)
Outstanding as of December 31, 2021	326,779	\$ 25.54	3.5
Issued		\$ -	-
Expired	(2,404)	\$ 52.00	-
Exercised	-	\$ -	-
Outstanding as of June 30, 2022	324,375	\$ 25.00	3.5
Warrants exercisable as of June 30, 2022	324,375	\$ 25.00	3.5
The aggregate intrinsic value of warrants outstanding and exercisable at June 30, 2022 was		\$ -	

# **Common Stock Options**

As of June 30, 2022 and December 31, 2021, there were no stock options outstanding.

## **Restricted Stock**

A summary of the restricted stock award activity (represented by restricted stock units (RSUs) for the six months ended June 30, 2022 as follows:

# Restricted Stock Units

		Wei	ghted Average Grant
	Number of Units		Date Fair Value
Nonvested at December 31, 2021	642,094	\$	35.93
Granted	797,046	\$	36.79
Vested	(375,730)	\$	66.74
Nonvested at June 30, 2022	1,063,410	\$	25.69

During the second quarter of 2022, the Compensation Committee issued grants that will vest over the next four years and result in total stock compensation expense of approximately \$20.5 million.

## NOTE 8 - DEBT, COMMITMENTS AND CONTINGENCIES

#### Debt

On October 1, 2021, the Company entered into a Revolving Credit and Security Agreement (the "Agreement") with Silvergate Bank pursuant to which Silvergate has agreed to loan the Company up to \$100 million on a revolving basis. At June 30, 2022 and December 31, 2021 there were amounts of \$35,000,000 and \$0 outstanding under this facility. This facility was refinanced on July 28, 2022 (see Note 9 - Subsequent Events).

On November 18, 2021, the Company issued \$650 million principal amount of its 1.00% Convertible Senior Notes due 2026 (the "Notes"). The Notes were issued pursuant to, and are governed by, an indenture dated as of November 18, 2021, between the Company and U.S. Bank National Association, as trustee. Pursuant to the purchase agreement between the Company and the initial purchasers of the Notes, the Company also granted the initial purchasers an option to purchase up to an additional \$97,500,000 principal amount of Notes. This option was exercised and an additional \$97,500,000 principal amount of Notes were issued on November 23, 2021.

As of June 30, 2022 and December 31, 2021, notes outstanding, net of unamortized discounts of approximately \$17.2 million and \$19.1 million, respectively, were \$730.3 million and \$728.4 million, respectively.

#### Leases

In February 2016, the FASB issued ASU No. 2016-02, Leases (Topic 842), and has since issued amendments thereto, related to the accounting for leases (collectively referred to as "ASC 842"). ASC 842 establishes a right-of-use, or ROU, model that requires a lessee to record a ROU asset and a lease liability on the balance sheet for all leases with terms longer than 12 months. Leases will be classified as either finance or operating, with classification affecting the pattern of expense recognition in the income statement. Effective January 1, 2019, the Company adopted ASU 842. The Company determines if an arrangement contains a lease at inception based on whether or not the Company has the right to control the asset during the contract period and other facts and circumstances.

The Company leases office space in the United States under operating lease agreements. Office space is the Company's only material underlying asset class under operating lease agreements. The Company has no material finance leases.

Effective June 1, 2018, the Company rented its corporate office at 1180 North Town Center Drive, Suite 100, Las Vegas, Nevada 89144, on a month to month basis.

Effective February 14, 2022, the Company rented an office located at Tower 101, 101 NE Third Avenue, Fort Lauderdale, Florida, 33301, for a term of 63 months,

Effective March 1, 2022, the Company rented an office located at 300 Spectrum Center Drive, Irvine CA, 92618, for a term of 24 months.

Effective May 1, 2022, the Company rented warehouse space located at 3306 5th Street SE, East Wenatchee, Washington, 98802, for a term of 24 months.

As of June 30, 2022, the Company's right-of-use ("ROU") assets and total lease liabilities were \$1.2 million and \$1.2 million, respectively for leases in the United States. As of December 31, 2021, the Company's ROU assets and total lease liabilities were nil. The Company has made payments and amortized the right-of-use assets totalling \$28,790 and \$47,555, respectively, for the three and six month periods ending June 30, 2022.

Operation lease costs are recorded on a straight-line basis within operating expenses. The Company's total lease expense is comprised of the following:

		For the Six Months Ended			
	_	June 30, 2022		June 30, 2021	
Operating leases	_				
Operating lease cost	\$	100,808	\$	97,407	
Operating lease expense		100,808		97,407	
Short-term lease rent expense		14,298		14,289	
Total rent expense	\$	115,106	\$	111,696	

	For the Three	e Months Ended
	June 30, 2022	June 30, 2021
Operating leases		
Operating lease cost	\$ 74,676	\$ -
Operating lease expense	74,676	
Short-term lease rent expense	7,158	5,126
Total rent expense	\$ 81,834	\$ 5,126

Additional information regarding the Company's leasing activities as a lessee is as follow:

	For the Six Months Ended June 30, 2022 June 30, 2021				
	 June 30, 2022		June 30, 2021		
Operating cash flows from operating leases	\$ 27,376	\$		-	
Weighted-average remaining lease term – operating leases	4.4			-	
Weighted-average discount rate – operating leases	5.0%			0.0%	

As of June 30, 2022, contractual minimum lease payments are as follows for the next five years.

Year	Amount
2022 (remaining)	175,719
2023	357,651
2024	259,934
2025	236,696
2026	240,991
Thereafter	101,824
Total	1,372,815

### **Legal Proceedings**

#### Ho Matter

On January 14, 2021, Plaintiff Michael Ho ("Plaintiff" or "Ho") filed a Civil Complaint for Damages and Restitution ("Complaint") against the Company and 10 Doe Defendants. The Complaint alleges six causes of action against the Company, (1) Breach of Written Contract; (2) Breach of Implied Contract; (3) Quasi-Contract; (4) Services Rendered; (5) Intentional Interference with Prospective Economic Relations; and (6) Negligent Interference with Prospective Economic Relations, which is the one plead against "all Defendants" and is most likely to involve later named defendants. The claims arise from the same set of facts, Ho alleges that the Company profited from commercially-sensitive information he shared with the Company and then it refused to compensate him for his role in securing the acquisition of a supplier of energy for the Company. On February 22, 2021, the Company responded to Mr. Ho's Complaint with a general denial and the assertion of applicable affirmative defenses. Then, on February 25, 2021, the Company removed the action to the United States District Court in the Central District of California, where the action remains pending. The Company filed a motion for summary judgment/adjudication of all causes of action. On February 11, 2022, the Court granted the motion and dismissed Ho's 2nd, 5th and 6th causes of action. Discovery is closed. The Court held a pre-trial conference on February 24, 2022, where it vacated the March 3, 2022 trial date and ordered the parties to meet and confer on a new trial date. The Court discussed the various theories of damages amintained by the parties. In its ruling on the summary judgment motion and at the pre-trial conference on February 24, 2022, the Court noted that a jury is more likely to accept \$150,000 as an appropriate damages amount if liability is found, as opposed to the various theories espoused by Ho that result in multi-million dollar recoveries. Due to outstanding issues of fact and law, it is impossible to predict the outcome at this time; however, after

## Information Subpoena

On October 6, 2020, the Company entered into a series of agreements with multiple parties to design and build a data center for up to 100-megawatts in Hardin, MT. In conjunction therewith, the Company filed a Current Report on Form 8-K on October 13, 2020. The 8-K discloses that, pursuant to a Data Facility Services Agreement, the Company issued 6,000,000 shares of restricted Common Stock, in transactions exempt from registration under Section 4(a)(2) of the Securities Act of 1933, as amended. During the quarter ended September 30, 2021, the Company and certain of its executives received a subpoena to produce documents and communications concerning the Hardin, Montana data center facility described in our Form 8-K dated October 13, 2020. We understand that the SEC may be investigating whether or not there may have been any violations of the federal securities law. We are cooperating with the SEC.

## Putative Class Action Complaint

On December 17, 2021, a putative class action complaint was filed in the United States District Court for the District of Nevada, against the Company and present and former senior management. The complaint alleges securities fraud related to the disclosure of an SEC investigation previously made by the Company on November 15, 2021. Plaintiff Tad Schlatre served the complaint on the Company on March 1, 2022. Multiple alleged shareholders have moved for appointment as lead plaintiff. Those motions remain pending before the Court.

## Derivative Complaints

On February 18, 2022, a shareholder derivative complaint was filed in the United States District Court for the District of Nevada, against current and former members of the Company's board of directors and senior management. The complaint is based on allegations substantially similar to the allegations in the December 2021 putative class action complaint, related to the Company's disclosure of an SEC investigation previously made by the Company on November 15, 2021. On March 4, 2022, the complaint was served on the Company. On April 4, 2022, the defendants moved to dismiss the complaint.

On May 5, 2022, a second shareholder derivative complaint was filed in the United States District Court for the District of Nevada, against current and former members of the Company's board of directors and senior management. The second shareholder derivative complaint is based on allegations substantially similar to the allegations in the February 18, 2022 derivative complaint. On May 11, 2022, the defendants moved to dismiss the second shareholder derivative complaint.

On June 1, 2022, the Court entered an order consolidating the two derivative actions. A June 13, 2022 scheduling order provides for plaintiffs to file a consolidated complaint and for renewed motions to dismiss the consolidated shareholder derivative complaint. The consolidated complaint has not yet been filed.

In the opinion of management, after consulting legal counsel, the ultimate disposition of these five matters will not have a material adverse effect on the Company and its related entities combined financial position, results of operations, or liquidity.

#### **NOTE 9 – SUBSEQUENT EVENTS**

On July 5, 2022, the Company expanded certain hosting arrangements to include an additional 42 megawatts of hosting capacity at a facility near Granbury, Texas. The Company expects to have an additional 14,000 miners installed at this facility, bringing the total number of miners installed near Granbury to 26,000 or approximately 3.6 EH/s. Based on current construction schedules these miners are expected to be installed before the end of 2022.

On July 12, 2022, the Company entered into an agreement to secure approximately 200 megawatts of hosting capacity for the Company's previously purchased miners, including 90 megawatts of hosting capacity in Texas and at least 110 megawatts of hosting capacity in North Dakota. The Company expects to have 66,000 miners, representing approximately 9.2 EH/s, hosted across these facilities. Based on current construction schedules, installations of the Company's miners are expected to begin at these facilities during the fourth quarter of 2022 with all miners installed by approximately mid-year 2023. As part of this agreement, the Company has an option to increase hosting capabilities utilizing up to an additional 70 megawatts in North Dakota. The Company also secured an additional 12 megawatts of hosting capacity with a variety of other providers and expects to install approximately 4,000 miners, representing approximately 0.8 EH/s, with these hosting providers, starting in August 2022.

On July 15, 2022 the Federal Energy Regulatory Commission found that King Mountain Upton Wind, LLC (King Mountain) would retain its status as an exempt wholesale generator notwithstanding a proposal to share ownership of the Interconnection Facilities as tenants-in-common with a retail energy customer. This action enabled the energization of a modular data center adjacent to the Generating Facility. Approximately 69,000 of the Company's bitcoin mining machines are located at this data center and energization enabled this equipment to come online starting on August 5, 2022.

On July 19, 2022, the Company sold its final shipment of equipment in accordance with its April agreement with DCRBN. The equipment was sold to DCRBN in conjunction with the development of commercial activities at the King Mountain wind farm in McCamey, TX. The Company recorded cash proceeds totalling \$43.6 million and realized a pre-tax gain on the sale of such assets of \$28.8 million during the month of July 2022.

On July 28, 2022 the Company terminated its power purchase agreements and commenced the acceleration of its exit from Hardin. As a result, the Company further accelerated the cost of a prepaid service contract (\$7.2 million in cost of revenue – Energy, hosting and other) and the remaining depreciation (\$13.1 million in cost of revenue – depreciation and amortization) related to the infrastructure assets at Hardin during the month of July. The data center infrastructure assets and the prepaid service contract have therefore been fully depreciated or amortized as of July 31, 2022. The bitcoin mining servers that are on site are in the process of being inventoried and removed from the facility and will be sold or redeployed to other locations in the near future.

On July 28, 2022, the Company entered into a Revolving Credit and Security Agreement (the "Agreement") with Silvergate Bank (the "Bank") pursuant to which Silvergate has agreed to loan the Company up to \$100,000,000 on a revolving basis pursuant to the terms of the Agreement and the \$100,000,000 principal amount revolving credit note issued by the Company in favor of the Bank under the Agreement ("Note"). The terms of the facility ("RLOC") set forth in the Agreement and Note are as follows:

**Initial Term:** Termination is on August 5, 2024.

Availability: The RLOC shall be made available from time to time to the Company for periodic draws (provided no event of default then exists) from

its closing date up to and including the termination date of the Agreement.

Origination Fee: 0.35% of the Loan Commitment to the Bank (or \$350,000); due at RLOC closing (and on each anniversary if the RLOC continues for

more than one year).

**Unused Commitment Fee:** 0.25% per annum of the portion of the unused Loan Commitment, payable monthly in arrears.

Renewal: The RLOC may be renewed annually by agreement between the Bank and the Company, subject to (without limitation): (i) Company

makes a request for renewal, in writing, no less than sixty (60) days prior to the then current maturity date, (ii) no event of default then exists, (iii) Company provides all necessary documentation to extend the RLOC, (iv) Company has paid all applicable fees related to the loan renewal, and (v) the Bank has approved such extension request according to its internal credit policies as determined by the Bank in

its sole and absolute discretion.

Interest Rate and Payments: Interest only to be paid monthly, with principal all due at maturity. The interest rate is defined as the higher of (i) the Floor Rate and (ii)

Prime Rate plus the Applicable Margin. "Floor Rate" shall mean, as of any date of determination: (a) five and one-quarter percent (5.25%) for any days during an Interest Period the LTV Ratio is less than forty percent (40%), (b) six percent (6.00%) for any days during an Interest Period the LTV Ratio is greater than or equal to forty percent (40%) and less than fifty-five percent (55%), and (c) six and three-quarter percent (6.75%) for any day. The Applicable Margin means at any time: (a) one and one-quarter percent (1.25%) for any days during an Interest Period the LTV Ratio is less than forty (40%), (b) two percent (2.00%) for any days during an Interest Period the LTV Ratio is greater than or equal to 40% and less than fifty-five percent (55%), and (c) two and three-quarter percent (2.75%) for any days

during an Interest Period the LTV Ratio is greater than or equal to fifty-five percent (55%).

Collateral: The RLOC will be secured by a pledge of a sufficient amount of Company's right, title and interest in and to bitcoin stored in a custody

account for the benefit of the Bank (the "Collateral Account"). the Bank will establish a Collateral Account with a regulated custodial entity (the "Custodian") that has been approved by the Bank. the Bank and Custodian will have a custodial agreement to perfect the security interest in the pledged Collateral Account which, among other things, allows for 1) the Bank to monitor the balance of the Collateral Account and 2) allows the Bank to have exclusive control over the Collateral Account including liquidation of the collateral in

the event of Company's default under the terms of the RLOC. the Bank may also file a UCC financing statement on the pledged collateral.

At origination, the Company must ensure the Collateral Account balance has sufficient bitcoin to cause a Loan to Value (the "LTV") ratio of 65% (or less) ("Minimum Advance Rate") on the unpaid principal balance of the RLOC. If at any time the LTV ratio exceeds 75%, the

Company must bring the rate of advance to the Minimum Advance Rate.

Covenants: The Company must maintain a minimum adjusted net worth of \$350,000,000. The Company must maintain a minimum liquidity of

\$25,000,000.

On that same date, the Company entered into a Term Credit and Security Agreement ("Term Loan Agreement") and Term Credit Note with the Bank with the following terms:

**Initial Term:** Termination is on August 5, 2024.

**Minimum Advance Rate:** 

Availability: Up to \$100,000,000.00 with \$50,000,000.00 to be made as of the Closing Date (the "Initial Draw"), and \$50,000,000.00 to be made, at

Borrower's request, on or before April 25, 2023 (the "Delayed Draw"), and subject to satisfaction of the conditions set forth in the Term

Loan Agreement.

Fees: An origination fee of \$150,000.00 and a contingent draw fee in the amount of \$250,000.00 (the, "Contingent Draw Fee") upon the

execution of the Term Loan Agreement. This Contingent Draw Fee will be refunded to the Company if it borrows the Delayed Draw by

no later than November 25, 2022.

Interest Rate and Payments: Interest, which shall be due on the principal amount of the loan, at the higher of 5.75% and the Prime Rate plus 1.75%, only to be paid

monthly, with principal all due at maturity.

Collateral: The Term Loan will be secured by a pledge of a sufficient amount of Company's right, title and interest in and to bitcoin stored in a

custody account for the benefit of the Bank (the "Collateral Account"). the Bank will establish a Collateral Account with a regulated custodial entity (the "Custodian") that has been approved by the Bank, the Bank and Custodian will have a custodial agreement to perfect the security interest in the pledged Collateral Account which, among other things, allows for 1) the Bank to monitor the balance of the Collateral Account and 2) allows the Bank to have exclusive control over the Collateral Account including liquidation of the collateral in the event of Company's default under the terms of the Term Loan, the Bank may also file a UCC financing statement on the pledged

collateral.

Covenants: The Company must maintain a minimum adjusted net worth of \$350,000,000. The Company must maintain a minimum liquidity of

\$25,000,000.

### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

This report on Form 10-Q ("Report") and other written and oral statements made from time to time by us may contain so-called "forward-looking statements," all of which are subject to risks and uncertainties. Forward-looking statements can be identified by the use of words such as "expects," "plans," "will," "forecasts," "projects," "intends," "estimates," and other words of similar meaning. One can identify them by the fact that they do not relate strictly to historical or current facts. These statements are likely to address our growth strategy, financial results and product and development programs. One must carefully consider any such statement and should understand that many factors could cause actual results to differ from our forward-looking statements. These factors may include inaccurate assumptions and a broad variety of other risks and uncertainties, including some that are known and some that are not. No forward-looking statement can be guaranteed and actual future results may vary materially.

Information regarding market and industry statistics contained in this Report is included based on information available to us that we believe is accurate. It is generally based on industry and other publications that are not produced for purposes of securities offerings or economic analysis. We have not reviewed or included data from all sources and cannot assure investors of the accuracy or completeness of the data included in this Report. Forecasts and other forward-looking information obtained from these sources are subject to the same qualifications and the additional uncertainties accompanying any estimates of future market size, revenue and market acceptance of products and services. We do not assume any obligation to update any forward-looking statements. As a result, investors should not place undue reliance on these forward-looking statements.

The following discussion and analysis is intended as a review of significant factors affecting our financial condition and results of operations for the periods indicated. The discussion should be read in conjunction with our consolidated financial statements and the notes presented herein. In addition to historical information, the following Management's Discussion and Analysis of Financial Condition and Results of Operations contains forward-looking statements that involve risks and uncertainties. Our actual results could differ significantly from those expressed, implied or anticipated in these forward-looking statements as a result of certain factors discussed herein and any other periodic reports filed and to be filed with the Securities and Exchange Commission.

#### Cautionary Note Regarding Forward-Looking Statements

This report and other documents that we file with the Securities and Exchange Commission contain forward-looking statements that are based on current expectations, estimates, forecasts and projections about our future performance, our business, our beliefs and our management's assumptions. Statements that are not historical facts are forward-looking statements. Words such as "expect," "outlook," "forecast," "would," "could," "should," "project," "intend," "plan," "continue," "sustain", "on track", "believe," "seek," "estimate," "anticipate," "may," "assume," and variations of such words and similar expressions are often used to identify such forward-looking statements, which are made pursuant to the safe harbor provisions of the Private Securities Litigation Reform Act of 1995. These forward-looking statements are not guarantees of future performance and involve risks, assumptions and uncertainties, including, but not limited to, those described in our reports that we file or furnish with the Securities and Exchange Commission. Should one or more of these risks or uncertainties materialize, or should underlying assumptions prove incorrect, actual results may vary materially from those indicated or anticipated by such forward-looking statements. Accordingly, you are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date they are made. Except to the extent required by law, we undertake no obligation to update publicly any forward-looking statements after the date they are made, whether as a result of new information, future events, changes in assumptions or otherwise.

# **Business of the Company**

The Company was incorporated in the State of Nevada on February 23, 2010 under the name Verve Ventures, Inc. On December 7, 2011, the Company changed its name to American Strategic Minerals Corporation and were engaged in exploration and potential development of uranium and vanadium minerals business. In June 2012, the Company discontinued the minerals business and began to invest in real estate properties in Southern California. In October 2012, the Company discontinued its real estate business and the Company commenced IP licensing operations, at which time the Company's name was changed to Marathon Patent Group, Inc. The Company commenced mining bitcoin in 2018 and changed its name to Marathon Digital Holdings, Inc. on March 1, 2021. As of June 30, 2022, the Company no longer holds any legacy IP assets and is solely focused on the mining of bitcoin and ancillary opportunities within the bitcoin ecosystem under the name Marathon Digital Holdings, Inc.

#### **Recent developments**

During the three-month period ended June 30, 2022, deteriorating macroeconomic conditions contributed to a significant downturn in financial markets. These conditions were more pronounced in businesses exposed to digital assets, including bitcoin mining. Many digital asset companies executed cost savings measures, reduced expansion plans and capital expenditures, sold digital assets and in some cases executed layoffs of staff. Holders of digital assets including bitcoin experienced a significant decrease in the value of their digital asset holdings during the period. The price of bitcoin dropped from \$45,539 on April 1, 2022 to a low of \$19,018 on June 18, 2022, and was \$19,785 on June 30, 2022. The Company faced these same challenges, along with operational issues at our Hardin, MT facility and delays in the energization of a bitcoin mining facility in Texas. The Company did not sell any bitcoin during the period as a means of raising cash, although we did execute a previously-contracted sale of equipment during the quarter, details of which are included below. Despite the economic and operational challenges experienced during the quarter, the Company ended the period with \$89.7 million in cash on hand and continues to expect to have sufficient liquidity sources in the future to support ongoing operations. Our primarily sources of liquidity are expected to be cash on hand, available borrowing capacity with our Revolving and Term Loan facilities with Silvergate Bank, our ATM facility and our bitcoin holdings.

A brief discussion of some of the more significant recent events impacting the Company's operations follows.

On June 11, 2022, a severe storm passed through Hardin damaging the power generating facility that supplies the data center with power. As a result, the Company's bitcoin production at the plant was significantly reduced. Anticipated repairs to the plant were persistently delayed until July 14, at which point the plant resumed operations at reduced power levels and operating capacity. Additional outages continued to occur at the plant throughout July, and the Company decided to accelerate its exit from Hardin, moving the date up from the planned date of August 15 to July 28. As a result, the Company further accelerated the cost of a prepaid service contract and the remaining depreciation and amortization related to the infrastructure assets at Hardin during the month of July. The data center infrastructure assets and the prepaid service contract have therefore been fully depreciated or amortized as of July 31, 2022. The bitcoin mining servers that are on site are in the process of being inventoried and removed from the facility and will be sold or redeployed to other locations in the near future.

On June 10, 2022 the Company withdrew approximately 4,769 bitcoin from its investment in NYDIG Digital Assets Fund III, LP, the ("Investment Fund") and transferred the bitcoin directly into the Company's account. As a result, the Company will no longer receive "mark-to-market" accounting for the bitcoin formerly held in the Investment Fund and the 4,769 bitcoin will now be classified as "Digital currencies" on the balance sheet and subject to impairment analysis as a indefinite-lived intangible.

On June 14, 2022 the Company terminated its loan of 600 bitcoin with NYDIG. The Company decided to terminate the bitcoin loan in response to recent market conditions and its desire to hold all of its bitcoin directly so it could fully utilize these holdings for corporate purposes as needed, including as collateral for credit facilities.

On July 5, 2022, the Company expanded certain hosting arrangements to include an additional 42 megawatts of hosting capacity at a facility near Granbury, Texas. The Company expects to have an additional 14,000 miners installed at this facility, bringing the total number of miners installed near Granbury to 26,000 or approximately 3.6 EH/s. Based on current construction schedules these miners are expected to be installed before the end of 2022.

On July 12, 2022, the Company entered into an agreement to secure approximately 200 megawatts of hosting capacity for the Company's previously purchased miners, including 90 megawatts of hosting capacity in Texas and at least 110 megawatts of hosting capacity in North Dakota. The Company expects to have 66,000 miners, representing approximately 9.2 EH/s, hosted across these facilities. Based on current construction schedules, installations of the Company's miners are expected to begin at these facilities during the fourth quarter of 2022 with all miners installed by approximately mid-year 2023. As part of this agreement, the Company has an option to increase hosting capabilities utilizing up to an additional 70 megawatts in North Dakota. The Company also secured an additional 12 megawatts of hosting capacity with a variety of other providers and expects to install approximately 4,000 miners, representing approximately 0.8 EH/s, with these hosting providers, starting in August 2022.

On July 15, 2022 the Federal Energy Regulatory Commission found that King Mountain Upton Wind, LLC would retain its status as an exempt wholesale generator notwithstanding a proposal to share ownership of the Interconnection Facilities as tenants-in-common with a retail energy customer. King Mountain had filed a petition on April 5, 2022 seeking a declaratory order to confirm its status as an exempt wholesale generator. In the Petition, King Mountain stated that it proposed to share ownership of interconnection facilities that are currently eligible facilities within the meaning of section 32(a)(2) of the Public Utility Holding Company Act as tenants-in common with a retail energy customer. King Mountain stated that it intended to sell wholesale electricity from the Generating Facility to a third party, who would then sell electricity at retail to the owner of a modular data center which would operate adjacent to the Generating Facility and supply it with renewable energy. This action enabled the energization a modular data center adjacent to the Generating Facility. Approximately 69,000 of the Company's bitcoin mining machines are located at this data center and energization enabled this equipment to come online starting on August 5, 2022.

#### **Non-GAAP Financial Measures**

We provide investors with a reconciliation from net income to the non-GAAP measure known as Adjusted EBITDA as a component of Management's Discussion and Analysis. For each period in question, we define "Adjusted EBITDA" as (a) GAAP net income (or loss) plus (b) adjustments to add back the impacts of (1) depreciation and amortization, (2) interest expense, (3) income tax expense and (4) adjustments for non-cash and non-recurring items (which currently include (i) stock compensation expense, (ii) net of withholding taxes and (iii) impairments of patents (if any).

Adjusted EBITDA is not a measurement of financial performance under GAAP and, as a result, this measure may not be comparable to similarly titled measures of other companies. Non-GAAP financial measures are subject to material limitations as they are not in accordance with, or a substitute for, measurements prepared in accordance with GAAP. Adjusted EBITDA is not meant to be considered in isolation and should be read only in conjunction with our Quarterly Reports on Form 10-Q and our Annual Reports on Form 10-K as filed with the Securities and Exchange Commission. Management uses both Adjusted EBITDA and the supplemental information provided herein as a means of understanding, managing and evaluating business performance and to help inform operating decision making. We rely primarily on our Consolidated Condensed Financial Statements to understand, manage, and evaluate our financial performance and use the non-GAAP financial measures only supplementally.

## **Recent Issued Accounting Standards**

See Note 2 to our Consolidated Condensed Financial Statements for a discussion of recent accounting standards and pronouncements.

## **Results of Operations**

For the Three Months ended June 30, 2022 and 2021

	Three Months Ended June 30,				Favorable		
	2022			2021	J)	(Unfavorable)	
Revenues	\$	24,921,816	\$	29,321,857	\$	(4,400,041	
Cost of revenues - energy, hosting and other	Ψ	(16,684,759)	Ψ	(4,056,168)	Ψ	(12,628,591	
Cost of revenues - depreciation and amortization		(24,709,797)		(2,937,666)		(21,772,131	
Total margin	_	(16,472,740)		22,328,023		(38,800,763	
Gain on sale of equipment		58,181,516		-		58,181,516	
General and administrative expenses		(12,641,331)		(6,831,040)		(5,810,291	
General and administrative expenses		(12,041,331)		(0,831,040)		(3,610,291	
Changes in carrying value of digital assets:		(70 (00 500)		(114.704.506)		25.016.006	
Change in fair value of digital currencies held in fund		(79,688,590)		(114,704,596)		35,016,006	
Impairment of digital currencies		(127,590,231)		(11,078,660)		(116,511,571	
		(207,278,821)		(125,783,256)		(81,495,565	
Non-operating income		165,280		1,400,872		(1,235,592	
Net loss		(191,646,642)		(108,884,620)		(82,762,022	
Bitcoin ("BTC") production during the period, in BTC		707		654		53	
Reconciliation to Adjusted EBITDA							
Net loss	\$	(191,646,642)	\$	(108,884,620)	\$	(82,762,022	
Exclude: Interest expense		3,748,322		1,203		3,747,119	
Exclude: Income tax expense (benefit)		9,852,224		(1,984)		9,854,208	
EBIT		(178,046,096)		(108,885,401)		(69,160,695	
Exclude: Depreciation and amortization		24,709,797		2,937,666		21,772,131	
EBITDA		(153,336,299)		(105,947,735)		(47,388,564	
Exclude: Stock compensation expense, net of withholding tax		6,132,224		875,971		5,256,253	
Adjusted EBITDA	\$	(147,204,075)	\$	(105,071,764)	\$	(42,132,311	

## **Revenues and Total Margin**

We generated revenues of \$24.9 million during the three months ended June 30, 2022 compared with \$29.3 million during the three months ended June 30, 2021. This \$4.4 million decrease in revenue was driven by lower revenue per bitcoin mined (\$6.8 million) resulting from lower market prices for bitcoin in the current-year period when compared with the prior-year period. This decrease was partially offset by an 8% increase in bitcoin production activity (a \$2.4 million increase in revenues) from the prior-year period. Cost of revenues – energy, hosting and other during the three months ended June 30, 2022 amounted to \$16.7 million compared with \$4.1 million in the prior-year period. This \$12.6 million increase was driven by accelerated cost recognition associated with the early exit from Hardin (\$9.4 million) and to a lesser extent higher costs per bitcoin mined. Total margin, which we define as revenues less cost of revenues – energy, hosting and other and cost of revenues – depreciation and amortization, totalled a loss of \$16.5 million compared with an income position of \$22.3 million in the prior-year period. This \$38.8 million decrease in total margin was driven primarily by the impact of accelerated costs related to the Hardin exit and the lower revenue per bitcoin mined.

Notwithstanding the increased mining activities vs. the prior-year period, our production of bitcoin during the three months ended June 30, 2022 was negatively impacted by ongoing maintenance issues and the storm at our Hardin, MT facility as well as the delays in energizing our bitcoin mining equipment at the King Mountain data center in Texas.

#### Gain on sale of assets

On December 2, 2021, we entered into an agreement with DCRBN Ventures Development and Acquisition LLC ("DCRBN") in which the Company agreed to sell certain equipment to DCRBN starting in April 2022, in conjunction with the development of commercial activities at the King Mountain wind farm in McCamey, TX. During the three months ended June 30, 2022, the Company sold equipment for cash proceeds totalling \$87.2 million and realized a pre-tax gain on the sale of such assets of \$58.2 million. There were no such sales in the prior-year period.

#### General and administrative expenses

General and administrative expenses were \$12.6 million for the three months ended June 30, 2022, an increase of \$5.8 million from the prior-year period. Our general and administrative expenses increased primarily as a result of higher stock-based (non-cash) compensation expense, which increased to \$6.2 million from \$0.9 million in the prior-year period; and higher costs associated with increased business activities.

#### Changes in carrying value of digital assets:

- <u>Impairment of digital currencies recorded in operating expenses</u>: We incurred significant impairment of digital assets during the three months ended June 30, 2022 as the price of bitcoin declined to a low of \$19,018 on June 18, 2022. Total impairment expense was \$127.6 million for the three months ended June 30, 2022 compared with an impairment expense of \$11.1 million for the prior-year period.
- Change in fair value of digital currencies recorded in operating income (expense): On June 10, 2022 the company withdrew 4,769 bitcoin from its investment fund. Total changes in the fair value of investment fund from April 1 through the June 10 withdrawal date resulted in a loss of \$79.7 million in the current year period. During the prior-year quarter, the change in fair value of the bitcoin held in the investment fund was a loss of \$114.9 million.

#### Non-operating income

Non-operating income decreased primarily due to changes in the fair value of a stock warrant liability recorded in the prior-year period.

# **Depreciation and amortization**

Depreciation and amortization, which we classify as "Cost of revenues – depreciation and amortization" in our statements of operations, increased significantly when compared to the prior-year period primarily due to the acceleration of depreciation related to our exit of the Hardin, MT facility (a \$15.8 million increase in depreciation) and, to a lesser extent increased depreciation costs associated with a higher number of mining servers in operation (\$4.7 million).

## Interest expense

Interest expense increased \$3.7 million from the prior-year as a result interest related to the convertible notes issued in November 2021 (\$2.8 million) and interest on borrowings outstanding under the Company's revolving credit agreement (\$0.9 million).

# Income tax expense

Income tax expense was \$9.8 million for the period ended June 30, 2022. We recorded tax expense despite a pre-tax loss from operations due to a valuation adjustment related to the certain deferred tax benefits.

#### Net loss

We recorded a net loss of \$(191.6) million in the current year period compared with net loss of \$(108.9) million in the prior period. This \$82.7 million decline was primarily driven by the impact of declines in the carrying value of our digital assets (\$81.5 million), higher depreciation expense (\$21.8 million), lower total margin \$(17.0 million), increased income tax expense (\$9.8 million), higher operating expenses (\$5.8 million) increased interest expense (\$3.7 million) partially offset by the gain on the sale of equipment of \$58.1 million.

# Adjusted EBITDA

Adjusted EBITDA was a loss of \$(147.2) million compared with a loss of \$(105.1) million in the prior-year period.

This \$42.1 million decline was primarily driven by the impact of declines in the carrying value of our digital assets (\$81.5 million) and lower total margin \$(17.0 million) partially offset by the gain on the sale of equipment (\$58.1 million).

## **Results of Operations**

For the Six Months ended June 30, 2022 and 2021

		Six Months E	nded Jun	ne 30,		Favorable	
	2022			2021	(Unfavorable)		
Revenues	\$	76,639,534	\$	38,474,672	S	38,164,862	
Cost of revenues - energy, hosting and other		(29,201,710)		(5,724,646)		(23,477,064)	
Cost of revenues - depreciation and amortization		(38,586,480)		(3,675,603)		(34,910,877)	
Total margin		8,851,344		29,074,423		(20,223,079)	
Gain on sale of equipment	_	58,181,516		<u>-</u> _		58,181,516	
General and administrative expenses		(26,835,089)		(60,175,421)		33,340,332	
Changes in carrying value of digital assets:							
Change in fair value of digital currencies held in fund		(85,016,208)		17,323,121		(102,339,329)	
Impairment of digital currencies		(147,141,486)		(11,740,859)		(135,400,627)	
		(232,157,694)		5,582,262		(237,739,956)	
Non-operating income (expenses)		393,973		(7,250)		401,223	
Net loss		(204,605,231)		(25,527,878)		(179,077,353)	
Bitcoin ("BTC") production during the period, in BTC		1,966		846		1,119	
Reconciliation to Adjusted EBITDA							
Net loss	\$	(204,605,231)	\$	(25,527,878)	\$	(179,077,353)	
Exclude: Interest expense		6,562,358		2,406		6,559,952	
Exclude: Income tax expense (benefit)		5,557,560		(514)		5,558,074	
EBIT		(192,485,313)		(25,525,986)		(166,959,327)	
Exclude: Depreciation and amortization		38,586,480		3,675,603		34,910,877	
EBITDA	·	(153,898,833)		(21,850,383)		(132,048,450)	
Exclude: Stock compensation expense, net of withholding tax		15,407,576		51,907,111		(36,499,535)	
Exclude: Impairment of patents		919,363		_		919,363	
Adjusted EBITDA	\$	(137,571,894)	\$	30,056,728	\$	(167,628,622)	

# **Revenues and Total Margin**

We generated revenues of \$76.6 million during the six months ended June 30, 2022 compared with \$38.5 million during the six months ended June 30, 2021. This increase in revenue was driven by a 132% increase in bitcoin production (\$50.9 million) partially offset by lower revenue per bitcoin mined (\$12.7 million) resulting from lower market prices for bitcoin in the current-year period when compared with the prior-year period. Cost of revenues – energy, hosting and other during the six months ended June 30, 2022 amounted to \$29.2 million compared with \$5.7 million in the prior-year period. This \$23.5 million increase was driven by higher costs per bitcoin mined (\$15.9 million, including the impact of accelerated costs related to the exit from Hardin) and increased costs associated with higher bitcoin production (\$7.6 million). Total margin, which we define as revenues less cost of revenues – energy, hosting and other and cost of revenues – depreciation and amortization, totalled \$8.9 million compared with \$29.1 million in the prior-year period. This \$20.2 million decrease in total margin was driven primarily by the impact of accelerated costs related to the Hardin exit partially offset by the increase in bitcoin production.

Notwithstanding the increased mining activities vs. the prior-year period, our production of bitcoin during the six months ended June 30, 2022 was negatively impacted by ongoing maintenance issues and the storm at our Hardin, MT facility as well as the delays in energizing our bitcoin mining equipment at the King Mountain data center in Texas.

### Gain on sale of assets

On December 2, 2021, we entered into an agreement with DCRBN Ventures Development and Acquisition LLC ("DCRBN") in which the Company agreed to sell certain equipment to DCRBN starting in April 2022, in conjunction with the development of commercial activities at the King Mountain wind farm in McCamey, TX. During the six months ended June 30, 2022, the Company sold equipment for cash proceeds totalling \$87.2 million and realized a pre-tax gain on the sale of such assets of \$58.2 million. There were no such sales in the prior-year period.

#### General and administrative expenses

General and administrative expenses were \$26.8 million for the six months ended June 30, 2022 compared with \$60.2 million for the prior year period, a decrease of \$33.4 million from the prior-year period. This decrease was primarily the result of a \$36.5 million decrease in stock-based (non-cash) compensation expense partially offset by higher costs associated with increased business activities.

## Changes in carrying value of digital assets:

- Impairment of digital currencies recorded in operating expenses: We incurred significant impairment of digital assets during the six months ended June 30, 2022 as the
  price of bitcoin hit new lows in June 2022. Total impairment expense was \$147.1 million for the six months ended June 30, 2022 compared with an impairment expense of
  \$11.7 million for the prior-year period.
- Change in fair value of digital currencies recorded in operating income (expense): During the month of June the company withdrew 4,769 bitcoin from its investment fund. Total year to day changes in the fair value of investment fund through the June 10 withdrawal date resulted in a loss of \$85.0 million in the current year period. During the prior-year period, the change in fair value of the bitcoin held in the investment fund was an increase in fair value of \$17.3 million.

## Non-operating income (loss)

Non-operating income increased primarily due to changes in the fair value of a stock warrant liability recorded in the prior-year period.

## **Depreciation and amortization**

Depreciation and amortization, which we classify as "Cost of revenues – depreciation and amortization" in our statements of operations, increased \$34.9 million when compared to the prior-year period primarily due to the acceleration of depreciation related to our exit of the Hardin, MT facility (a \$19.9 million) and increased depreciation costs associated with a higher number of mining servers in operation when compared with the prior year period (\$10.8 million).

#### Interest expense

Interest expense increased \$6.6 million from the prior-year as a result interest related to the convertible notes issued in November 2021 (\$5.7 million) and interest on borrowings outstanding under the Company's revolving credit agreement (\$0.9 million).

#### Income tax expense (benefit)

Income tax expense was \$5.5 million for the six months ended June 30, 2022 compared with a small tax benefit in the prior year. We recorded tax expense despite a pre-tax loss from operations due to a valuation adjustment related to the certain deferred tax benefits record in the current year period.

## Net loss

We recorded a net loss of \$(204.6) million in the current year period compared with net loss of \$(25.5) million in the prior period. This \$179.1 million decline was primarily driven by the impact of declines in the carrying value of our digital assets (\$237.7 million), higher depreciation expense (\$34.9 million), and to a lesser extent higher interest expense and income tax expense. Partially offsetting these unfavorable variances was the gain on the sale of equipment (\$58.1 million), lower general and administrative expenses (\$33.3 million), and higher total margin (\$14.7 million).

# Adjusted EBITDA

Adjusted EBITDA was a loss of \$(137.6) million compared with positive Adjusted EBITDA of \$30.1 million in the prior year period. This \$167.6 million decline was primarily driven by the impact of declines in the carrying value of our digital assets (\$232.2 million) partially offset by the gain on the sale of equipment (\$58.2 million), higher total margin (\$14.7 million).

## Financial Condition and Liquidity

Cash, cash equivalents and restricted cash totalled \$89.7 million at June 30, 2022, a decrease of \$178.9 million from December 31, 2021. The decrease in cash, cash equivalents and restricted cash was primarily driven by a \$334.0 million use of cash from investing activities resulting primarily from significant levels of advances to vendors related to bitcoin mining server orders (\$394.0 million) and, to a lesser extent, purchases of property and equipment (\$13.8 million) and equity investments (\$14.0 million) partially offset by proceeds from assets sales (\$87.2 million).

Cash flows from financing activities resulted in a source of cash of \$196.0 million, primarily from proceeds from the issuance of common stock (\$161.0 million) and proceeds from borrowings outstanding under the Company's \$100 million revolving credit agreement (\$35.0 million).

Cash flows from operating activities resulted in a use of funds of \$40.8 million. Positive cash flow impacts of operating activities before the impact of changes in operating assets and liabilities (a \$47.9 million source of funds) were more than offset by a \$88.7 million use of funds from changes in operating assets and liabilities, primarily due to changes in digital currencies (a \$76.5 million use of funds).

We had \$35 million outstanding under its revolving credit agreement at June 30, 2022. The maximum borrowings outstanding under the credit agreement during the six months ended June 30, 2022, was \$70 million.

The Company expects to have sufficient liquidity, including cash on hand and available borrowing capacity to support ongoing operations. We will continue to seek to fund the growth in our business activities through the capital markets, including both debt and equity issuances.

#### **Bitcoin Holdings**

At June 30, 2022, we held approximately 10,055 bitcoin with a total carrying value of \$190.4 million on the balance sheet. Approximately 2,820 bitcoin were being utilized as collateral for revolving credit borrowings and were classified as "digital currencies, restricted". The remaining bitcoin were classified as "Digital currencies" on the balance sheet. The fair market value of our bitcoin holdings at June 30, 2022 was approximately \$198.9 million and the value of a single bitcoin was approximately \$19,785.

At June 30, 2021 we held a total of 5,784 bitcoin with a total carrying value of \$195.9 million on the balance sheet. The fair market value of our bitcoin holdings at June 30, 2021 was approximately \$202.7 million and the value of a single bitcoin was approximately \$35,041.

We expect to increase our bitcoin holdings over time primarily through mining activities. As our mining activities increase, we may sell a portion of bitcoin produced in future periods to fund monthly operations, for treasury management purposes or for general corporate purposes.

#### **Off-balance Sheet Arrangements**

We have not entered into any other financial guarantees or other commitments to guarantee the payment obligations of any third parties. We have not entered into any derivative contracts that are indexed to our shares and classified as stockholder's equity or that are not reflected in our consolidated condensed financial statements. Furthermore, we do not have any retained or contingent interest in assets transferred to an unconsolidated entity that serves as credit, liquidity or market risk support to such entity.

## Item 3. Quantitative and Qualitative Disclosures About Market Risk.

As of June 30, 2022, our exposure to market risk was primarily from our At The Market Facility. During the quarter the price at which we sold our common stock per share fluctuated from \$5.34 to \$28.92 with an average price per share of \$13.38. We would have risk on our commercial credit facility had we drawn down upon it as the interest rate changes at the greater of 6% or the prime rate plus 2.75%. We have no other floating debt obligations. Our interest rate exposure will be primarily due to differences between our floating rate debt obligations compared to our floating rate short-term investments. Our ability to borrow under our Revolving Line of Credit is based upon a floating formula regarding the value of collateral which is our owned bitcoin, thus decreases in the market price for bitcoin limit our ability to borrow under the facility.

There have been no other material changes in our primary risk exposures or management of market risks as of this quarter.

#### Item 4. Controls and Procedures.

#### Disclosure Controls and Procedures.

Our management is responsible for establishing and maintaining adequate internal control over financial reporting as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act. Our management is also required to assess and report on the effectiveness of our internal control over financial reporting in accordance with Section 404 of the Sarbanes-Oxley Act of 2002 ("Section 404"). Our internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes of accounting principles generally accepted in the United States. Management assessed the effectiveness of our internal control over financial reporting as of June 30, 2022. In making this assessment, we used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission (COSO) in Internal Control - Integrated Framework in the 2013 COSO framework. Based on this assessment, management concluded that our disclosure controls and procedures were not effective as of June 30, 2022 for the reasons stated in our Annual Report on Form 10-K for the year ended December 31, 2021.

A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected on a timely basis. A significant deficiency is a deficiency, or a combination of deficiencies, in internal control over financial reporting that is less severe than a material weakness, yet important enough to merit attention by those responsible for oversight of the company's financial reporting.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies and procedures may deteriorate

As part of our ongoing program to implement changes and further improve our internal controls and in conjunction with our Code of Ethics, our independent directors have been working with management to include protocols and measures aimed at ensuring quality of our internal controls. Among those measures is the implementation of a whistle blower hotline, which allows third parties to anonymously report noncompliant activity. The hotline may be accessed as follows:

To file a report, use the Client Code "MarathonPG" and pick one of the following options:

Call: 1-877-647-3335

Click: <a href="http://www.RedFlagReporting.com">http://www.RedFlagReporting.com</a>

#### Changes in Internal Controls.

There have been changes in our internal control over financial reporting during the quarter ended June 30, 2022 that have materially affected, or are reasonably likely to materially affect, our internal controls over financial reporting.

We have created a position of Assistant Controler to support our Chief Accounting Officer and his staff which position was filled in July 2022. We are also undertaking an exhaustive review process of our outside internal controls consultants and bringing in additional resources to support our efforts to continue remediation of our internal controls.

#### PART II - OTHER INFORMATION

## Item 1. Legal Proceedings

Ho Matter

On January 14, 2021, Plaintiff Michael Ho ("Plaintiff" or "Ho") filed a Civil Complaint for Damages and Restitution ("Complaint") against the Company and 10 Doe Defendants. The Complaint alleges six causes of action against the Company, (1) Breach of Written Contract; (2) Breach of Implied Contract; (3) Quasi-Contract; (4) Services Rendered; (5) Intentional Interference with Prospective Economic Relations; and (6) Negligent Interference with Prospective Economic Relations, which is the one plead against "all Defendants" and is most likely to involve later named defendants. The claims arise from the same set of facts, Ho alleges that the Company profited from commercially-sensitive information he shared with the Company and then it refused to compensate him for his role in securing the acquisition of a supplier of energy for the Company. On February 22, 2021, the Company responded to Mr. Ho's Complaint with a general denial and the assertion of applicable affirmative defenses. Then, on February 25, 2021, the Company removed the action to the United States District Court in the Central District of California, where the action remains pending. The Company filed a motion for summary judgment/adjudication of all causes of action. On February 11, 2022, the Court granted the motion and dismissed Ho's 2nd, 5th and 6th causes of action. Discovery is closed. The Court held a pre-trial conference on February 24, 2022, where it vacated the March 3, 2022 trial date and ordered the parties to meet and confer on a new trial date. The Court discussed the various theories of damages maintained by the parties. In its ruling on the summary judgment motion and at the pre-trial conference on February 24, 2022, the Court noted that a jury is more likely to accept \$150,000 as an appropriate damages amount if liability is found, as opposed to the various theories espoused by Ho that result in multi-million dollar recoveries. Due to outstanding issues of fact and law, it is impossible to predict the outcome at this time; however, after

#### Information Subpoena

On October 6, 2020, the Company entered into a series of agreements with multiple parties to design and build a data center for up to 100-megawatts in Hardin, MT. In conjunction therewith, the Company filed a Current Report on Form 8-K on October 13, 2020. The 8-K discloses that, pursuant to a Data Facility Services Agreement, the Company issued 6,000,000 shares of restricted Common Stock, in transactions exempt from registration under Section 4(a)(2) of the Securities Act of 1933, as amended. During the quarter ended September 30, 2021, the Company and certain of its executives received a subpoena to produce documents and communications concerning the Hardin, Montana data center facility described in our Form 8-K dated October 13, 2020. We understand that the SEC may be investigating whether or not there may have been any violations of the federal securities law. We are cooperating with the SEC.

#### Putative Class Action Complaint

On December 17, 2021, a putative class action complaint was filed in the United States District Court for the District of Nevada, against the Company and present and former senior management. The complaint alleges securities fraud related to the disclosure of an SEC investigation previously made by the Company on November 15, 2021. Plaintiff Tad Schlatre served the complaint on the Company on March 1, 2022. Multiple alleged shareholders have moved for appointment as lead plaintiff. Those motions remain pending before the Court.

## Derivative Complaints

On February 18, 2022, a shareholder derivative complaint was filed in the United States District Court for the District of Nevada, against current and former members of the Company's board of directors and senior management. The complaint is based on allegations substantially similar to the allegations in the December 2021 putative class action complaint, related to the Company's disclosure of an SEC investigation previously made by the Company on November 15, 2021. On March 4, 2022, the complaint was served on the Company. On April 4, 2022, the defendants moved to dismiss the complaint.

On May 5, 2022, a second shareholder derivative complaint was filed in the United States District Court for the District of Nevada, against current and former members of the Company's board of directors and senior management. The second shareholder derivative complaint is based on allegations substantially similar to the allegations in the February 18, 2022 derivative complaint. On May 11, 2022, the defendants moved to dismiss the second shareholder derivative complaint.

On June 1, 2022, the Court entered an order consolidating the two derivative actions. A June 13, 2022 scheduling order provides for plaintiffs to file a consolidated complaint and for renewed motions to dismiss the consolidated shareholder derivative complaint. The consolidated complaint has not yet been filed.

In the opinion of management, after consulting legal counsel, the ultimate disposition of these five matters will not have a material adverse effect on the Company and its related entities combined financial position, results of operations, or liquidity.

Other than as disclosed herein, we know of no other material, active or pending legal proceedings against us, nor are we involved as a plaintiff in any material proceedings or pending litigation other than in the normal course of business.

#### Item 1A. Risk Factors.

There are no updates or changes to the risk factors set forth in our Annual Report on Form 10-K for the year ended December 31, 2021 except as set forth below.

## Our business could be harmed by prolonged power and internet outages, shortages, or capacity constraints and deployment delays.

Our operations require a significant amount of electrical power and access to high-speed internet to be successful. If we are unable to secure sufficient electrical power, or if we lose internet access for a prolonged period, we may be required to reduce our operations or cease them altogether. We are also dependent upon our third party energy providers to power miners upon installation, and there may be delays in deployment and implementation. If any of these scenarios occurs, our business and results of operations may be materially and adversely affected.

## We are subject to risks associated with our need for significant electrical power.

Our operations have required significant amounts of electrical power, and, as we continue to expand our mining fleet, we anticipate our demand for electrical power will continue to grow. If we are unable to continue to obtain sufficient electrical power on a cost-effective basis, we may not realize the anticipated benefits of our significant capital investments.

Additionally, our operations could be materially adversely affected by prolonged power outages. Therefore, we may have to reduce or cease our operations in the event of an extended power outage, or as a result of the unavailability or increased cost of electrical power. If this were to occur, our business and results of operations could be materially and adversely affected.

# Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

None.

Item 3. Defaults Upon Senior Securities.

None

Item 4. Mine Safety Disclosures.

Not applicable.

Item 5. Other Information.

Not applicable.

# Item 6. Exhibits.

10.1	Forms of Revolving Credit Agreement, Revolving Credit Note, Term Loan Agreement and Term Loan Note
31.1	Certification pursuant to Section 302 of the Sarbanes-Oxley Act of 2002*
31.2	Certification pursuant to Section 302 of the Sarbanes-Oxley Act of 2002*
32.1	Certification pursuant to Section 906 of the Sarbanes-Oxley Act of 2002*
32.2	Certification pursuant to Section 906 of the Sarbanes-Oxley Act of 2002*
101.ins	XBRL Instance Document**
101.sch	XBRL Taxonomy Schema Document**
101.cal	XBRL Taxonomy Calculation Document**
101.def	XBRL Taxonomy Linkbase Document**
101.lab	XBRL Taxonomy Label Linkbase Document**
101.pre	XBRL Taxonomy Presentation Linkbase Document**

104 Inline XBRL

<sup>\*</sup> Furnished herewith

<sup>\*\*</sup> Filed herein

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: August 9, 2022

# MARATHON DIGITAL HOLDINGS, INC.

By: /s/ Fred Thiel
Name: Fred Thiel

Title: Chief Executive Officer (Principal Executive Officer)

By: /s/Simeon Salzman
Name: Simeon Salzman
Title: Chief Accounting Officer

(Principal Financial and Accounting Officer)

# REVOLVING CREDIT AND SECURITY AGREEMENT

**AMONG** 

SILVERGATE BANK

(AS LENDER)

AND

MARATHON DIGITAL HOLDINGS, INC.

(AS BORROWER)

July 28, 2022

### REVOLVING CREDIT AND SECURITY AGREEMENT

This Revolving Credit and Security Agreement is dated as of July 28, 2022 among Marathon Digital Holdings, Inc., a Nevada corporation (the "Borrower") and SILVERGATE BANK, a California-chartered commercial bank ("Lender").

IN CONSIDERATION of the mutual covenants and undertakings herein contained, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Borrower and Lender hereby agree as follows:

#### DEFINITIONS.

- 1.1. Accounting Terms. As used in this Agreement, the Loan Documents or any certificate, report or other document made or delivered pursuant to this Agreement, accounting terms not defined in Section 1.2 or elsewhere in this Agreement and accounting terms partly defined in Section 1.2 to the extent not defined shall have the respective meanings given to them under GAAP.
- 1.2. <u>General Terms</u>. For purposes of this Agreement the following terms shall have the following meanings:
- "Account" shall mean, collectively, any deposit accounts maintained by Borrower with Lender.
- "Account Control Agreement," shall mean that certain Account Control Agreement, dated on or about the date hereof, between Borrower, Lender and Custodian, as the same may be amended, modified, extended, restated, replaced, or supplemented from time to time, including, without limitation, with a successor Custodian.
  - "Advance Date" shall mean the date on which an Advance is made by Lender.
  - "Advance Rate" shall mean sixty-five percent (65%).
  - "Advances" shall mean advances under the Revolving Commitment.
- "Affiliate" of any Person shall mean (a) any Person which, directly or indirectly, is in control of, is controlled by, or is under common control with such Person, or (b) any Person who is a director, manager, member, managing member, general partner or officer (i) of such Person, (ii) of any Subsidiary of such Person or (iii) of any Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the power, direct or indirect, (x) to vote ten percent (10%) or more of the Equity Interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for any such Person, or (y) to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.
- "Agreement" shall mean this Revolving Credit and Security Agreement, as the same may be amended, amended and restated, replaced and restated, extended, supplemented and/or otherwise modified from time to time.

1

"Anti-Terrorism Laws" shall mean any Laws relating to terrorism, trade sanctions programs and embargoes, import/export licensing, money laundering or bribery, and any regulation, order, or directive promulgated, issued or enforced pursuant to such Laws, all as amended, supplemented or replaced from time to time.

"Applicable Law" shall mean all Laws applicable to the Person, conduct, transaction, covenant, Loan Document or contract in question, all provisions of all applicable state, federal and foreign constitutions, statutes, rules, regulations, treaties, directives and orders of any Governmental Body, and all orders, judgments and decrees of all courts and arbitrators.

"Applicable Margin" shall mean, as of any date of determination: (a) one and one-quarter percent (1.25%) for any days during an Interest Period the LTV Ratio is less than forty (40%), (b) two percent (2.00%) for any days during an Interest Period the LTV Ratio is greater than or equal to 40% and less than fifty-five percent (55%), and (c) two and three-quarter percent (2.75%) for any days during an Interest Period the LTV Ratio is greater than or equal to fifty-five percent (55%).

"Approved Electronic Communication" shall mean each notice, demand, communication, information, document and other material transmitted, posted or otherwise made or communicated by e-mail, E-Fax, or any other equivalent electronic service agreed to by Lender, whether owned, operated or hosted by Lender, that any party is obligated to, or otherwise chooses to, provide to Lender pursuant to this Agreement or any Loan Document, including any financial statement, financial and other report, notice, request, certificate and other information material; provided that Approved Electronic Communications shall not include any notice, demand, communication, information, document or other material that Lender specifically instructs a Person to deliver in physical form.

"Availability Period" means the time period during which Borrower is entitled to request Advances.

"Base Rate" is the rate of interest per annum from time to time published in the money rates section of The Wall Street Journal or any successor publication thereto as the "prime rate" then in effect; provided that if such rate of interest, as set forth from time to time in the money rates section of The Wall Street Journal, becomes unavailable for any reason as determined by Lender, the "Prime Rate" shall mean the rate of interest per annum announced by Lender as its prime rate in effect at its principal office in the State of California (such Lender announced Prime Rate not being intended to be the lowest rate of interest charged by Lender in connection with extensions of credit to debtors).

"Beneficial Owner" shall mean, for Borrower, each of the following: (a) each individual, if any, who, directly or indirectly, owns 25% or more of such Borrower's Equity Interests; and (b) a single individual with significant responsibility to control, manage, or direct such Borrower.

"Bitcoin" means a digital commodity (or "digital asset") based on the decentralized, open source protocol of the peer-to-peer Bitcoin computer network. By common convention, Bitcoin with a capital "B" typically refers to the Bitcoin Network as a whole, whereas bitcoin with a lowercase "b" refers to the digital asset native to the Bitcoin Network. For purposes of Article IV

of this Agreement, the term bitcoin shall be deemed to include Forked Assets, which will be considered proceeds of the original bitcoin for which they share a common public address.

"Bitcoin Network" means the peer-to-peer Bitcoin computer network;

"Blockchain" means the public transaction ledger of the Bitcoin Network;

"Borrower" shall have the meaning set forth in the preamble to this Agreement and shall extend to all permitted successors and assigns of such Persons.

"Borrowing Base" means, as of any date of determination, the value in Dollars of bitcoin held in the Collateral Account, in each case as reasonably determined by Lender as of 1:00 p.m. Pacific Time ("PT") on the Business Day of valuation ("Valuation Time") under the Valuation Method. In the absence of fraud or manifest error, any valuations prepared by Custodian and approved by Lender shall be deemed reasonable for purposes of this paragraph. In determining the Borrowing Base, Lender may exclude from the calculation of the Borrowing Base any bitcoin or other assets that Lender reasonably determines creates an unreasonable risk relating to OFAC, BSA/AML compliance or similar matters based on Lender's blockchain analytics analysis and other diligence.

"Business Day" shall mean any day other than Saturday or Sunday or a legal holiday on which commercial banks are authorized or required by Law to be closed for business in California.

"Certificate of Beneficial Ownership" shall mean, for Borrower, a certificate in form and substance acceptable to Lender (as amended or modified by Lender from time to time in its reasonable discretion), certifying, among other things, the Beneficial Owner of Borrower.

"Change in Law" shall mean the occurrence, after the Closing Date, of any of the following: (a) the adoption or taking effect of any Applicable Law; (b) any change in any Applicable Law or in the administration, implementation, interpretation or application thereof by any Governmental Body; or (c) the making or issuance of any request, rule, guideline or directive (whether or not having the force of Law) by any Governmental Body; provided that notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, regulations, guidelines, interpretations or directives thereunder or issued in connection therewith (whether or not having the force of Applicable Law) and (y) all requests, rules, regulations, guidelines, interpretations or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities (whether or not having the force of Law), in each case pursuant to Basel III, shall in each case be deemed to be a Change in Law regardless of the date enacted, adopted, issued, promulgated or implemented.

"Change of Control" shall mean: (a) the occurrence of any event (whether in one or more transactions) which results in a transfer of control of Borrower to a Person; (b) any merger, consolidation or sale of substantially all of the property or assets of Borrower; or (c) removal of Fred Thiel as an officer and director of the Borrower. For purposes of this definition, "control of Borrower" shall mean the power, direct or indirect, in a single transaction, not involving a public

offering(x) to vote more than fifty percent (50%) of the Equity Interests having ordinary voting power for the election of directors (or the individuals performing similar functions) of Borrower or (y) to direct or cause all or substantially all of the direction of the management and policies of Borrower by contract or otherwise.

"Closing Date" shall mean the date on which the conditions precedent set forth in Section 8.1 shall be satisfied or waived, in accordance with the terms hereof or such other date as may be agreed to in writing by the parties hereto.

"Code" shall mean the Internal Revenue Code of 1986, as the same may be amended or supplemented from time to time, and any successor statute of similar import, and the rules and regulations thereunder, as from time to time in effect.

"Collateral" shall mean and include all of Borrower's rights, title and interest, in and to any assets, including bitcoin, deposited, or required to be deposited, in the Collateral Account, together with all security entitlements (as such term is defined in Article 8 of the Uniform Commercial Code as adopted in the State of New York), in and to any assets, including bitcoin, held in the Collateral Account, together with all substitutions, replacements and proceeds arising out of any of the foregoing.

"Collateral Account" means the account in the name of the Borrower established to hold bitcoin collateral pledged by Borrower to the Lender, as security for the Indebtedness owed by Borrower to Lender, and made subject to the Account Control Agreement.

"Consents" shall mean all filings and all licenses, permits, consents, approvals, authorizations, qualifications and orders of Governmental Bodies and other third parties, domestic or foreign, necessary to carry on Borrower's business or necessary (including to avoid a conflict or breach under any agreement, instrument, other document, license, permit or other authorization) for the execution, delivery or performance of this Agreement, or the Loan Documents, including any Consents required under all applicable federal, state or other Applicable Law.

"Controlled Group" shall mean, at any time, Borrower and all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control and all other entities which, together with Borrower, are treated as a single employer under Section 414 of the Code or 29 U.S.C. 1301(b)(1).

"Covered Entity" shall mean (a) Borrower, each of Borrower's Subsidiaries, all Guarantors and all pledgors of Collateral and (b) each Person that, directly or indirectly, is in control of a Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the direct or indirect (x) ownership of, or power to vote, 25% or more of the issued and outstanding Equity Interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for such Person, or (y) power to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.

"<u>Custodian</u>" shall mean NYDIG Trust Company LLC, or such other successor custodian that Lender may approve from time to time, which approval may be granted or withheld in Lender's reasonable discretion.

"<u>Default</u>" shall mean an event, circumstance or condition which, with the giving of notice or passage of time or both, would constitute an Event of Default.

"Dollar" and the sign "\$" shall mean lawful money of the United States of America.

"Effective Date" means the date indicated in a document or agreement to be the date on which such document or agreement becomes effective, or, if there is no such indication, the date of execution of such document or agreement.

"<u>Eligible Contract Participant</u>" has the meaning ascribed to such term under Section 1(a)(12) of the Commodity Exchange Act, as amended and which may further be amended from time to time, including as amended by the Commodity Futures Modernization Act of 2000.

"Equity Interests" means shares of capital stock, partnership interests, membership interests in a limited liability company, beneficial interests in a trust or other equity ownership interests in a Person, and any warrants, options or other rights entitling the holder thereof to purchase or acquire any such equity interest.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as the same may be amended or supplemented from time to time and the rules and regulations promulgated thereunder.

"Event of Default" shall have the meaning set forth in Article X hereof.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

"Floor Rate" shall mean, as of any date of determination: (a) five and one-quarter percent (5.25%) for any days during an Interest Period the LTV Ratio is less than forty percent (40%), (b) six percent (6.00%) for any days during an Interest Period the LTV Ratio is greater than or equal to forty percent (40%) and less than fifty-five percent (55%), and (c) six and three-quarter percent (6.75%) for any days during an Interest Period the LTV Ratio is greater than or equal to fifty-five percent (55%).

" $\underline{Forked\ Assets}$ " shall mean digital assets that result from any Hard Fork occurring after the Effective Date.

"GAAP" shall mean generally accepted accounting principles in the United States of America in effect from time to time.

"Governmental Body" shall mean any nation or government, any state or other political subdivision thereof or any entity, authority, agency, division or department exercising the executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to a government (including any supra-national bodies such as the European Union or

the European Central Bank) and any group or body charged with setting financial accounting or regulatory capital rules or standards (including, without limitation, the Financial Accounting Standards Board, the Bank for International Settlements or the Basel Committee on Banking Supervision or any successor or similar authority to any of the foregoing).

"<u>Hard Fork</u>" means a substantial software modification to the Blockchain which results in two or more competing and incompatible Blockchain implementations, one running the premodification software program and the other running the modified version (i.e., a second "Bitcoin network").

"Indebtedness" shall mean, as to any Person at any time, any and all indebtedness, obligations or liabilities for borrowed money (whether matured or unmatured, liquidated or unliquidated, direct or indirect, absolute or contingent, or joint or several) of such Person. For the avoidance of doubt, the issuance of commercial letters of credit and the financing of mining equipment shall both be considered obligations in the Ordinary Course of Business.

"Indemnified Taxes" shall mean all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Body, including any interest, additions to tax or penalties applicable thereto.

"Insolvency Event" shall mean, with respect to any Person, including without limitation any Lender, such Person or such Person's direct or indirect parent company (a) becomes the subject of a bankruptcy or insolvency proceeding (including any proceeding under Title 11 of the United States Code), or regulatory restrictions, , or (b) has had a receiver, conservator, trustee, administrator, custodian, assignee for the benefit of creditors or similar Person charged with the reorganization or liquidation of its business appointed for it or has called a meeting of its creditors, or (c) admits in writing its inability, or be generally unable, to pay its debts as they become due or cease operations of its present business.

"Interest Payment Date" shall mean the first day of each month, or if such day is not a Business Day, the next succeeding Business Day, commencing on the first day of the first full calendar month after the initial Advance Date; provided that the final Interest Payment Date shall be the Maturity Date.

"Interest Period" shall mean, with respect to any Interest Payment Date, the period of time from and including the first day of the month of the immediately preceding Interest Payment Date through the last day of such month; provided that the Interest Period for the initial Interest Payment Date shall be the period from and including the initial Advance Date to, and including, the last day of the month of the Advance Date, and the final Interest Period shall end on the Maturity Date.

"Interest Rate" means a per annum rate equal to the greater of (i) the Base Rate plus the Applicable Margin or (ii) the Floor Rate.

"Law(s)" shall mean any law(s) (including common law and equitable principles), constitution, statute, treaty, regulation, rule, ordinance, opinion, issued guidance, code, release, ruling, order, executive order, injunction, writ, decree, bond, judgment, authorization or approval,

lien or award of or any settlement arrangement, by agreement, consent or otherwise, with any Governmental Body, foreign or domestic.

"Lien" shall mean any mortgage, deed of trust, pledge, hypothecation, assignment, security interest, lien (whether statutory or otherwise), charge, claim or encumbrance, or preference, priority or other security agreement or preferential arrangement held or asserted in respect of any Collateral of any kind or nature whatsoever including any conditional sale or other title retention agreement, any lease having substantially the same economic effect as any of the foregoing, and the filing of, or agreement to give, any financing statement under the Uniform Commercial Code or comparable law of any jurisdiction.

"Loan" shall mean the extension of credit made pursuant to this Agreement.

"Loan Documents" shall mean, collectively, this Agreement, the Account Control Agreement, the Note and any and all other documents, instruments and agreements made by Borrower in favor of Lender in connection with the Loan, as the same may be hereafter amended, modified or supplemented from time to time.

"Loan Party" shall mean, at each relevant time of determination, (a) Borrower, and (b) any other Person that is now or hereafter becomes a party to this Agreement as a "Borrower" or as a "Guarantor"; and "Loan Parties" shall mean collectively all such Persons.

"LTV Ratio" means, as of any date of determination, the quotient expressed as a percentage of (x) the outstanding Advances divided by (y) the Borrowing Base.

"Material Adverse Effect" shall mean (a) a material adverse effect upon, the operations, business, assets or financial condition of Borrower, including, without limitation, any material disruption of the Bitcoin Network and/or the marketplaces where bitcoin is exchanged for U.S. Dollars; (b) a material impairment of the ability of Borrower to perform its obligations under any Loan Document to which it is a party; or (c) a material adverse effect upon the legality, validity and/or enforceability of this Agreement or any of the Loan Documents; in each case, when taken as a whole, as determined by Lender in its commercially reasonable discretion.

"Maturity Date" means the earlier to occur of (a) the last day of the Term (as defined in Section 13.1 below), and (b) the date on which this Loan becomes due in accordance with Article XI of this Agreement.

"Maximum Rate" means, on any day, the highest rate of interest (if any) permitted by applicable law on such day.

"Note" shall have the meaning given to that term in Section 2.1, as amended, restated, extended, supplemented or otherwise modified from time to time.

"Obligations" shall mean and include any and all loans (including without limitation, all Advances), advances, debts, liabilities, obligations, covenants and duties owing by Borrower or any Subsidiary of Borrower under this Agreement or any Loan Document (and any amendments, extensions, renewals or increases thereto), to Lender (or to any other direct or indirect subsidiary

or affiliate of Lender) of any kind or nature, present or future (including any interest or other amounts accruing thereon, any fees accruing under or in connection therewith, any costs and expenses of any Person payable by Borrower and any indemnification obligations payable by Borrower arising or payable after maturity, or after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding relating to Borrower, whether or not a claim for post-filing or post-petition interest, fees or other amounts is allowable or allowed in such proceeding), whether direct or indirect (including those acquired by assignment or participation), absolute or contingent, joint or several, due or to become due, now existing or hereafter arising, contractual or tortious, liquidated or unliquidated, regardless of how such indebtedness or liabilities arise including all costs and expenses of Lender incurred in the documentation, negotiation, modification, enforcement, collection or otherwise in connection with any of the foregoing, including but not limited to reasonable attorneys' fees and expenses and all material obligations of Borrower to Lender to perform acts or refrain from taking any action.

"Ordinary Course of Business" shall mean, with respect to Borrower, the ordinary course of such Borrower's business as conducted on the Closing Date and reasonable extensions thereof.

"Organizational Documents" shall mean, with respect to any Person, any charter, articles or certificate of incorporation, certificate of organization, registration or formation, certificate of partnership or limited partnership, bylaws, operating agreement, limited liability company agreement, or partnership agreement of such Person and any and all other applicable documents relating to such Person's formation, organization or entity governance matters (including any shareholders' or equity holders' agreement or voting trust agreement) and specifically includes, without limitation, any certificates of designation for preferred stock or other forms of preferred equity.

"Other Taxes" shall mean all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies arising from any payment made hereunder or under any Loan Document or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any Loan Document.

"<u>Participant</u>" shall mean each Person who shall be granted the right by any Lender to participate in any of the Advances and who shall have entered into a participation agreement in form and substance satisfactory to such Lender.

"<u>Payment Office</u>" shall mean such office of Lender or account instructions, which Lender designates to receive payments from Borrower of amounts due under this Agreement.

"PBGC" shall mean the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA or any successor.

"Person" shall mean any individual, sole proprietorship, partnership, corporation, business trust, joint stock company, trust, unincorporated organization, association, limited liability company, limited liability partnership, institution, public benefit corporation, joint venture, entity or Governmental Body (whether federal, state, county, city, municipal or otherwise, including any instrumentality, division, agency, body or department thereof).

"<u>Plan</u>" shall mean any employee benefit plan within the meaning of Section 3(3) of ERISA maintained by Borrower or any member of the Controlled Group or to which Borrower or any member of the Controlled Group is required to contribute.

"Reportable Compliance Event" shall mean that any Covered Entity becomes a Sanctioned Person, or is charged by indictment, criminal complaint or similar charging instrument, arraigned, or custodially detained in connection with any Anti-Terrorism Law or any predicate crime to any Anti-Terrorism Law, or has knowledge of facts or circumstances to the effect that it is reasonably likely that any aspect of its operations is in actual or probable violation of any Anti-Terrorism Law.

"Revolving Commitment" shall mean the obligation of Lender to make Revolving Advances, in an aggregate principal and/or face amount not to exceed \$100,000,000.00.

"Sanctioned Country" shall mean a country subject to a sanctions program maintained under any Anti-Terrorism Law.

"Sanctioned Person" shall mean any individual person, group, regime, entity or thing listed or otherwise recognized as a specially designated, prohibited, sanctioned or debarred person, group, regime, entity or thing, or subject to any limitations or prohibitions (including but not limited to the blocking of property or rejection of transactions), under any Anti-Terrorism Law.

"Servicing and Custody Agreements" shall mean one or more agreements entered into by Lender and the Custodian relating to the Collateral.

"Securities Act of 1933, as amended.

"Subsidiary" of any Person shall mean a corporation or other entity of whose Equity Interests having ordinary voting power (other than Equity Interests having such power only by reason of the happening of a contingency) to elect a majority of the directors of such corporation, or other Persons performing similar functions for such entity, are owned, directly or indirectly, by such Person.

" $\underline{\text{Uniform Commercial Code}}$ " or "UCC" shall have the meaning set forth in Section 1.3 hereof.

"<u>USA PATRIOT Act</u>" shall mean the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

"Valuation Method" shall mean the price of bitcoin at the specified Valuation Time using the XBX index ("Index") published on TradeBlock website available at https://tradeblock.com/markets/index/xbx ("Valuation Source"). If at any time the Index or Valuation Source becomes unavailable Lender may designate an alternate Index or Valuation

Source as determined by Lender in its reasonable discretion. Lender agrees to provide Borrower with prompt notice of any change in the Index or Valuation Source.

- 1.3. <u>Uniform Commercial Code Terms</u>. All terms used herein and defined in the Uniform Commercial Code as adopted in the State of New York from time to time (the "<u>Uniform Commercial Code</u>") shall have the meaning given therein unless otherwise defined herein. To the extent the definition of any category or type of collateral is expanded by any amendment, modification or revision to the Uniform Commercial Code, such expanded definition will apply automatically as of the date of such amendment, modification or revision.
- Certain Matters of Construction. The terms "herein", "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular section, paragraph or subdivision. All references herein to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, this Agreement. Any pronoun used shall be deemed to cover all genders. Wherever appropriate in the context, terms used herein in the singular also include the plural and vice versa. All references to statutes and related regulations shall include any amendments of same and any successor statutes and regulations. Unless otherwise provided, all references to any instruments or agreements to which Lender is a party, including references to any of the Loan Documents, shall include any and all modifications, supplements or amendments thereto, any and all restatements or replacements thereof and any and all extensions or renewals thereof. Except as otherwise expressly provided for herein, all references herein to the time of day shall mean the time in San Diego, California. Whenever the words "including" or "include" shall be used, such words shall be understood to mean "including, without limitation" or "include, without limitation". A Default or an Event of Default shall be deemed to exist at all times during the period commencing on the date that such Default or Event of Default occurs to the date on which such Default or Event of Default is waived in writing pursuant to this Agreement or, in the case of a Default, is cured within any period of cure expressly provided for in this Agreement; and an Event of Default shall "continue" or be "continuing" until such Event of Default has been waived in writing pursuant to this Agreement. Any Lien referred to in this Agreement or any of the Loan Documents as having been created in favor of Lender, any agreement entered into by Lender pursuant to this Agreement or any of the Loan Documents, any payment made by or to or funds received by Lender pursuant to or as contemplated by this Agreement or any of the Loan Documents, or any act taken or omitted to be taken by Lender, shall, unless otherwise expressly provided, be created, entered into, made or received, or taken or omitted, for the benefit or account of Lender. All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an exception to, or otherwise within the limitations of, another covenant shall not avoid the occurrence of a default if such action is taken or condition exists. In addition, all representations and warranties hereunder shall be given independent effect so that if a particular representation or warranty proves to be incorrect or is breached, the fact that another representation or warranty concerning the same or similar subject matter is correct or is not breached will not affect the incorrectness of a breach of a representation or warranty hereunder.

### II. ADVANCES, PAYMENTS.

- 2.1. <u>Revolving Advances</u>. Subject to the terms and conditions set forth in this Agreement, Lender will make Revolving Advances to Borrower in aggregate amounts outstanding at any time equal to the lesser of (x) the Revolving Commitment or (y) an amount equal to the product of (i) the Advance Rate multiplied by (ii) the Borrowing Base (the "<u>Maximum Advance Amount</u>"). Lender may determine the Borrowing Base at any Valuation Time, in its reasonable discretion, following its receipt of a Request for Advance (as defined below) in accordance with the Valuation Method. The Revolving Advances shall be evidenced by the Revolving Credit Note, of even date herewith, made by Borrower in favor of Lender (the "<u>Note</u>").
- Procedures for Requesting Revolving Advances. To make a request for an Advance, the Borrower shall deliver to the Lender and Custodian two (2) Business Days prior to the proposed Advance Date, a request for advance in an amount not less than \$500,000.00 in the form of Exhibit A hereto (a "Request for Advance"), and be limited to no more than one (1) draw per Business Day. Notwithstanding the foregoing, Lender and Custodian may, in their discretion, accept a Request for Advance with less than two (2) Business Days prior written notice. Lender shall be entitled to rely on any Request for Advance signed by the individual signing this Agreement on behalf of Borrower or otherwise designated by Borrower, in writing, as authorized to make requests for Advances. Upon the satisfaction or waiver of each of the conditions precedent set forth in Section 8.2, the Lender shall disburse such Advance to Borrower's deposit account with Lender. If Borrower requests that an Advance be deposited into another account of Borrower, Borrower shall deliver complete wire instructions for that account when it submits its Request for Advance, and Borrower shall be responsible for any wire transfer fees imposed by Lender in accordance with its fee schedule governing Borrower's bank transactions. Any wire requests and/or instructions submitted after 10:00 a.m. PT on the Advance Date may not be deposited until the next Business Day.

# 2.3. Manner and Repayment of Advances.

- (a) The Revolving Advances shall be due and payable in full on the last day of the Term subject to earlier prepayment as herein provided. Notwithstanding the foregoing, all Advances shall be subject to earlier repayment upon (x) acceleration upon the occurrence of an Event of Default under this Agreement or (y) termination of this Agreement.
- (b) All payments of principal, interest and other amounts payable hereunder, or under any of the Loan Documents shall be made to Lender at the Payment Office not later than 3:00 p.m. PT on the due date therefor in Dollars in federal funds or other funds immediately available to Lender. Lender shall have the right to effectuate payment of any and all Obligations due and owing hereunder by charging Borrower's Account.
- (c) Except as expressly provided herein, all payments (including prepayments) to be made by Borrower on account of principal, interest, fees and other amounts payable hereunder shall be made without deduction, setoff or counterclaim and shall be made to Lender, in each case on or prior to 3:00 p.m. PT, in Dollars and in immediately available funds.

## 2.4. Repayment of Excess Advances.

- (a) If, on any day, the LTV Ratio is equal to or greater than seventy-five percent (75%) (a "Collateral Shortfall"), the Borrower shall have one (1) Business Day to cause the LTV Ratio to be less than or equal to sixty-five percent (65%). If the LTV Ratio is not less than or equal to sixty-five percent (65%) by 1 p.m. PST on the first (1st) Business Day after the occurrence of a Collateral Shortfall, an Event of Default shall be deemed to exist, in which case Lender shall be entitled to exercise any and all rights and remedies under Article XI and/or applicable law.
- (b) If, on any day, the LTV Ratio is less than sixty-five percent (65%), the Borrower may, in its sole discretion, increase the LTV Ratio up to sixty-five percent (65%) by either (i) after giving at least two (2) Business Day's prior notice to the Lender, withdrawing bitcoin from the Collateral Account in accordance with Section 2.6 below, which after giving effect to such withdrawal would not increase the LTV Ratio to be greater than sixty-five percent (65%), and/or (ii) requesting additional Advances as set forth in Section 2.2.
- 2.5 <u>Use of Proceeds.</u> Borrower shall apply the proceeds of Advances to (i) provide financing for capital expenditures and general corporate expenses related to the Borrower's bitcoin mining business (collectively, the "<u>Transactions</u>"), and (ii) pay fees and expenses relating to the Transactions.
- 2.6 Return of Excess Collateral. From time to time, Borrower may have more Collateral in the Account than is required by Section 2.4(b) based on the then current valuation of the Collateral (as determined by Lender) and the then principal amount of all outstanding Advances (such excess Collateral, the "Excess Collateral"). Borrower may, but not more frequently than once per Business Day and limited to no more than two (2) requests per work week (i.e. Monday through Friday), request a return of some or all of the Excess Collateral. Requests for a return of Excess Collateral must be in writing (which may be delivered by email) and be for a minimum amount of at least \$1,000,000.00 (or equivalent amount of BTC). Borrower shall deliver written requests to the Custodian. Lender shall use commercially reasonable efforts to authorize the Custodian to process such requests within two (2) Business Days after Lender's receipt of the request. In no event shall Borrower be entitled to a return of any Collateral, if, after giving effect to the return of that Collateral, the principal amount of outstanding Advances (together with any accrued but unpaid interest and any other amounts owed to Lender) would exceed the Maximum Advance Amount.

## III. INTEREST AND FEES.

- 3.1. Interest. Subject to the last sentence of this paragraph, the Obligations shall bear interest at the Interest Rate. Interest on Advances shall be payable in arrears on each Interest Payment Date; provided that all accrued and unpaid interest shall be due and payable at the end of the Term. Upon and after the occurrence of an Event of Default, and during the continuation thereof, at the option of Lender, or in the case of any Event of Default under Article X, immediately and automatically upon the occurrence of any such Event of Default without the requirement of any affirmative action by any party, the Obligations shall bear interest at the lesser of: (a) the Interest Rate, plus six percent (6%); or (b) the Maximum Rate (the "Default Rate").
- 3.2. <u>Origination Fee.</u> Borrower shall pay to Lender an origination fee of \$350,000.00, being paid upon the execution of this Agreement. In addition, to the extent the Term of this facility

has been extended pursuant to Section 13.1, on the first Business Day immediately preceding the date that is twelve (12) months after the Effective Date, and on that date each year thereafter until the Loan is repaid in full and the Revolving Commitment terminated, Borrower shall pay to Lender an annual fee in the amount of \$350,000.00.

- 3.3. <u>Unused Commitment Fee.</u> Borrower shall pay to Lender a fee equal to one-quarter of one percent (1/4%) per annum (computed on the basis of a 360-day year, actual days elapsed) on the daily unused amount of the Revolving Commitment, which fee shall be calculated on a monthly basis by Lender and shall be due and payable by Borrower in arrears on the first day of each month, commencing on the first day of the first full calendar month after the Effective Date.
- 3.4. <u>Computation of Interest and Fees</u>. Interest and fees hereunder shall be computed on the basis of a year of 360 days and for the actual number of days elapsed. If any payment to be made hereunder becomes due and payable on a day other than a Business Day, the due date thereof shall be extended to the next succeeding Business Day and interest thereon shall be payable at the applicable Interest Rate for the outstanding Advances during such extension.
- Maximum Charges. In no event whatsoever shall interest and other charges charged hereunder exceed the highest rate permissible under Applicable Law. In the event interest and other charges as computed hereunder would otherwise exceed the highest rate permitted under Applicable Law: (a) the interest rates hereunder will be reduced to the maximum rate permitted under Applicable Law; (b) such excess amount shall be first applied to any unpaid principal balance owed by Borrower; and (c) if the then remaining excess amount is greater than the previously unpaid principal balance, Lender shall promptly refund such excess amount to Borrower and the provisions hereof shall be deemed amended to provide for such permissible rate. Notwithstanding anything to the contrary contained in this Agreement or in any Loan Document, all agreements which either now are or which shall become agreements among Loan Parties and Lender are hereby limited so that in no contingency or event whatsoever shall the total liability for payments in the nature of interest, additional interest and other charges exceed the applicable limits imposed by any applicable usury laws. If any payments in the nature of interest, additional interest and other charges made under this Agreement or any Loan Document are held to be in excess of the limits imposed by any applicable usury laws, it is agreed that any such amount held to be in excess shall be considered payment of principal hereunder, and the indebtedness evidenced hereby shall be reduced by such amount so that the total liability for payments in the nature of interest, additional interest and other charges shall not exceed the applicable limits imposed by any applicable usury laws, in compliance with the desires of Loan Parties and Lender. The foregoing provisions shall never be superseded or waived and shall control every other provision of this Agreement or any Loan Document and all agreements among Borrower and Lender, or their respective successors and assigns. If the applicable state or federal law is amended in the future to allow a greater rate of interest to be charged under this Agreement than is presently allowed by applicable state or federal law, then the limitation of interest hereunder shall be increased to the maximum rate of interest allowed by applicable state or federal law as amended, which increase shall be effective hereunder on the effective date of such amendment, and all interest charges owing to Lender by reason thereof shall be payable in accordance with Section 3.1 of this Agreement.

- 3.6. <u>Increased Costs.</u> In the event that any Applicable Law or any Change in Law or compliance by Lender with any request or directive (whether or not having the force of law) from any central bank or other financial, monetary or other authority, shall:
- (a) subject Lender to any tax of any kind whatsoever with respect to this Agreement, or change the basis of taxation of payments to Lender in respect;
- (b) impose, modify or deem applicable any reserve, special deposit, assessment, special deposit, compulsory loan, insurance charge or similar requirement against assets held by, or deposits in or for the account of, advances or loans by, or other credit extended by, any office of Lender, including pursuant to Regulation D of the Board of Governors of the Federal Reserve System; or
- impose on Lender any other condition, loss or expense (other than Taxes) affecting this Agreement or any Loan Document or any Advance made by Lender; and the result of any of the foregoing is to increase the cost to Lender of making, converting to, continuing, renewing or maintaining its Advances hereunder by an amount that Lender deems to be material or to reduce the amount of any payment (whether of principal, interest or otherwise) in respect of any of the Advances by an amount Lender deems to be material, then, in any case Borrower shall promptly pay Lender, upon its demand, such additional amount as will compensate Lender for such additional cost or such reduction, as the case may be; provided that the foregoing shall not apply to increased costs which are reflected in the Base Rate, as the case may be. Lender shall certify the amount of such additional cost or reduced amount to Borrower, and such certification shall be conclusive absent manifest error. For purposes of this Section 3.6, the Dodd-Frank Wall Street Reform and Consumer Protection Act and any and all rules, regulations, orders, requests, guidelines, and directives adopted, promulgated or implemented in connection therewith are deemed to have been introduced and adopted after the date of this Agreement. Any such surcharge shall be limited to the direct increase in taxes or other fees caused thereby, and no convenience fee or other surcharges may be added by Lender.
- 3.7. <u>Application of Payments</u>. So long as no Event of Default then exists, all payments from Borrower received by Lender shall first be applied to accrued but unpaid interest; second, to the principal balance outstanding; and third, to any late charges and costs of collection. Upon the occurrence of an Event of Default, Lender may apply payments in such order and amounts as Lender elects, in its reasonable discretion.

### 3.8. Taxes.

(a) Any and all payments by or on account of any Obligations hereunder or under any Loan Document shall be made free and clear of and without reduction or withholding for any Indemnified Taxes or Other Taxes; provided that if Borrower shall be required by Applicable Law to deduct any Indemnified Taxes (including any Other Taxes) from such payments, then (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section) Lender receives an amount equal to the sum it would have received had no such deductions been made, (ii) Borrower shall make such deductions and (iii) Borrower shall timely pay the full amount deducted to the relevant Governmental Body in accordance with Applicable Law.

- (b) Without limiting the provisions of Section 3.8(a) above, Borrower shall timely pay any Other Taxes to the relevant Governmental Body in accordance with Applicable Law.
- (c) Borrower shall indemnify Lender, within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section) paid by Lender and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Body. A certificate as to the amount of such payment or liability delivered to Borrower by Lender shall be conclusive absent manifest error.
- (d) As soon as practicable after any payment of Indemnified Taxes or Other Taxes by Borrower to a Governmental Body, Borrower shall deliver to Lender the original or a certified copy of a receipt issued by such Governmental Body evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to Lender.
- (e) If Lender determines, in its reasonable discretion, that it has received a refund of any Indemnified Taxes or Other Taxes as to which it has been indemnified by Borrower or with respect to which Borrower have paid additional amounts pursuant to this Section, it shall pay to Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by Borrower under this Section with respect to the Indemnified Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses of the Lender and without interest (other than any interest paid by the relevant Governmental Body with respect to such refund); provided that Borrower, upon the request of Lender, agrees to repay the amount Governmental Body) to Lender in the event Lender is required to repay such refund to such Governmental Body. This Section shall not be construed to require Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to Borrower or any other Person.

### IV. COLLATERAL: GENERAL TERMS.

- 4.1. <u>Security Interest in the Collateral</u>. To secure the prompt payment and performance to Lender of the Obligations, Borrower hereby assigns, pledges and grants to Lender, a continuing security interest in and to and Lien on all of the Collateral, whether now owned or existing or hereafter created, acquired or arising and wheresoever located.
- 4.2. <u>Perfection of Security Interest</u>. Borrower shall take all action that may be necessary or commercially reasonably desirable, or that Lender may commercially reasonably request, so as at all times to maintain the validity, perfection, enforceability and priority of Lender's security interest in and Lien on the Collateral or to enable Lender to protect, exercise or enforce its rights hereunder and in the Collateral, including, but not limited to, (a) immediately discharging all Liens on the Collateral other than Lender's security interest, (b) entering into custodial arrangements satisfactory to Lender, and (c) executing and delivering financing statements, control agreements, instruments of pledge, notices and assignments, in each case in form and substance satisfactory to

Lender, relating to the creation, validity, perfection, maintenance or continuation of Lender's security interest and Lien under the UCC or other Applicable Law. By its signature hereto, Borrower hereby authorizes Lender to file against such Borrower, one or more financing, continuation or amendment statements pursuant to the UCC in form and substance satisfactory to Lender, on the condition that the description of the collateral in any such statement only is limited to only the Collateral. Subject to any expense limitation set forth in Section 13.9 all charges, expenses and fees Lender may incur in doing any of the foregoing, and any local taxes relating thereto, shall be charged to Borrower's Account as a Revolving Advance and added to the Obligations, or, at Lender's option, shall be paid by Borrower to Lender immediately upon demand.

- 4.3. Preservation of Collateral. Following the occurrence of a Default or Event of Default, in addition to the rights and remedies set forth in Section 11.1 hereof, Lender: (a) may at any time take such steps as Lender deems necessary to protect Lender's interest in and to preserve the Collateral, including the hiring of cyber and security consultants or implementing other security protection measures as Lender may deem appropriate; (b) may purchase or lease cold storage vaults to preserve the Collateral and to otherwise do all acts necessary to protect Lender's interests in the Collateral; and (c) may acquire one or more hardware wallets where Lender may store the private keys associated with all or part of the Collateral. Borrower shall cooperate fully with all of Lender's efforts to preserve the Collateral and will take such actions to preserve the Collateral as Lender may direct. All of Lender's expenses of preserving the Collateral, including any expenses relating to the bonding or insuring of a custodian, shall be charged to Borrower's Account as a Revolving Advance and added to the Obligations. Nothing contained in this Section 4.3 is intended to impose any burden or obligation on Lender to preserve or hold any portion of the Collateral, and Lender may immediately, with or without notice, liquidate some or all of the Collateral following an Event of Default.
- 4.4. <u>Ownership and Location of Collateral</u>. With respect to the Collateral, at the time the Collateral becomes subject to Lender's security interest, Borrower shall be the sole owner of and fully authorized and able to sell, transfer, pledge and/or grant a first priority security interest in each and every item of its respective Collateral to Lender; and the Collateral shall be free and clear of all Liens whatsoever.
- 4.5. <u>Defense of Lender's Interests.</u> Until (a) payment and performance in full of all of the Obligations and (b) termination of this Agreement, Lender's interests in the Collateral shall continue in full force and effect. During such period Borrower shall not, without Lender's prior written consent, pledge, sell, assign, transfer, create or suffer to exist a Lien upon or encumber or allow or suffer to be encumbered in any way, any part of the Collateral. Borrower shall defend Lender's interests in the Collateral against any and all Persons whatsoever. At any time, following demand by Lender for payment of all Obligations during the continuance of an Event of Default, Lender shall have the right to take control of the Collateral in whatever manner stored, including hardware wallets, software wallets, whether encrypted or unencrypted, and all medium upon which the private keys associated with the Collateral are retrievable. In addition, with respect to all Collateral, Lender shall be entitled to all of the rights and remedies set forth herein and further provided by the Uniform Commercial Code or other Applicable Law.

4.6. <u>Financing Statements</u>. Except with respect to the financing statements filed by Lender, no financing statement covering any of the Collateral or any proceeds thereof is or will be on file in any public office.

### 4.7 Custodian

- (a) Lender shall have no liability to Borrower or other person for Custodian's failure to comply with any of its contractual obligations with respect to the Collateral or any other acts or omissions of Custodian, including, without limitation, any losses or damages arising out of the theft or misappropriation of the Collateral or claims that Custodian failed to properly safeguard the Collateral. Any diligence performed by Lender with respect to the qualifications and suitability of Custodian is solely for Lender's benefit. Lender makes no representations or statements about the Custodian's experience or qualifications to perform its obligations under the Collateral Covenants. Borrower is engaged in commercial activities involving virtual currency and digital assets, has significant experience buying, selling and storing virtual currency and other digital assets, including bitcoin, and has done its own evaluation and diligence concerning Custodian. By entering into this Agreement and accepting the Advances, Borrower represents and warrants to Lender that it has performed independent diligence on the Custodian and determined Custodian to be qualified to custody the Collateral. Borrower is under no obligation to enter into the transactions evidenced by this Agreement and does so only after making the determination in the immediately preceding sentence.
- (b) In the event Borrower suffers any loss or damages as a result of Custodian's acts or omissions, including without limitation, any theft or misappropriation of the Collateral, Borrower shall look solely to Custodian for any such loss or damages. Borrower hereby waives and releases Lender from any and all claims, liabilities, damages, losses or otherwise caused by Custodian or Custodian's employees, agents or contractors, including, without limitation, any claims or causes of action based on the theory that Borrower sustained damages or losses as a result of Lender conditioning the making of the Loan on the requirement that a Lender-approved custodian control the Collateral during the term of the Loan.
- (c) Without limiting the generality of the provisions in subparagraph (a) above, Borrower also acknowledges that it shall have no right to offset any amounts owed under this Agreement or any Loan Documents by any claims against Custodian, including, without limitation, claims arising out of the theft or misappropriation of any of the Collateral, it being understood that Borrower's obligations hereunder are absolute and irrespective of any such losses or claims.
- (d) Borrower acknowledges that Lender may have entered into separate agreements with Custodian and/or one or more Affiliates of Custodian pursuant to which Lender compensates Custodian or Affiliates of Custodian, including, without limitation, for performing certain loan servicing functions with respect to the Collateral and/or the Loan. The obligations of the Custodian or the Affiliates of Custodian under those agreements to service certain aspects of the Loan and provide trade execution services are for the sole benefit of Lender and Borrower is not an intended third-party beneficiary under any such agreements.
- (e) Lender shall not be liable to Borrower for any delays in the return of any Collateral to Borrower resulting from any delays caused by the Custodian or any other cause outside the

reasonable control of Lender, including, without limitation, any disruption of the bitcoin network or governmental directive preventing the lawful transfer of the Collateral to Borrower.

- (f) Custodian shall constitute a "securities intermediary" as that term is defined in Article 8 of the Uniform Commercial Code of the State of New York ("Article 8"), or an agent thereof. Borrower and Lender acknowledge and agree that (i) the Collateral Account is considered a "securities account" under Article 8, and (iii) the Collateral in the Collateral Account shall be considered "financial assets" under Article 8. The treatment of Collateral as a financial asset under Article 8 does not determine the characterization or treatment of those assets under any other law or rule.
- Lender may from time to time require Borrower to establish the Collateral Account with a substitute Custodian (each, a "Replacement Custodian"). Upon Borrower's receipt of Lender's written request to establish a Collateral Account with a substitute Custodian, Borrower shall, within thirty (30) Business Days after receiving Lender's request, (i) cause such replacement Collateral Account to be established, (ii) transfer the Collateral to the replacement Collateral Account, and (iii) enter into, and cause the substitute Custodian to enter into, an account control agreement in form and substance acceptable to Lender. Borrower's failure to perform the foregoing obligations within that thirty (30) Business Days period shall entitle Lender to immediately (x) cause the liquidation of some or all of the Collateral and the application of the proceeds to repay all outstanding Advances, and/or (y) direct the Custodian to transfer some or all of the Collateral to another account under the exclusive control of the Lender, which other account may be maintained with another custodian or subject to self-custody (the actions in clauses (x) and (y), collectively, "Protective Measures". Notwithstanding anything to the contrary contained herein, in the event Lender determines a material adverse change has occurred with respect to the then current Custodian, Lender may immediately take such Protective Measures as Lender deems necessary or prudent, in its reasonable discretion, without any prior notice to, or consent from, Borrower. In the event Lender exercises any right to liquidate Collateral under this Section 4.7(g), Borrower shall remain liable for any remaining outstanding Advances after application of the proceeds from the liquidated Collateral and Lender shall have no obligation to make further Advances. Subject to Borrower's timely compliance with the foregoing provisions of this Section 4.7(g), in the event Lender previously caused the Collateral to be transferred to another custodian or to self-custody the Collateral, Borrower may thereafter request that Lender custody the Collateral with a substitute Custodian mutually acceptable to both Lender and Borrower. If, after thirty (30) days from the date Borrower provides such written request to Lender, Borrower and Lender are unable to agree upon a substitute Custodian, Borrower may, without any prepayment fees, repay the outstanding principal balance of the Loan, together with all accrued and unpaid interest and all other amounts due under the Loan Documents. Borrower acknowledges that the provisions of this Section 4.7 are not intended to condition the effectiveness of Lender's designation of a substitute Custodian on Borrower's consent or approval.

## V. REPRESENTATIONS AND WARRANTIES.

Borrower represents and warrants as follows:

5.1. <u>Authority</u>. Borrower has full power, authority and legal right to enter into this Agreement and the Loan Documents to which it is a party and to perform all its respective

Obligations hereunder and thereunder. This Agreement and the Loan Documents to which it is a party have been duly executed and delivered by Borrower, and this Agreement and the Loan Documents to which it is a party constitute the legal, valid and binding obligation of such Borrower enforceable in accordance with their terms, except as such enforceability may be limited by any applicable bankruptcy, insolvency, moratorium or similar laws affecting creditors' rights generally. The execution, delivery and performance of this Agreement and of the Loan Documents to which it is a party (a) are within such Borrower's corporate or company powers, as applicable, have been duly authorized by all necessary corporate or company action, as applicable, are not in contravention of law or the terms of such Borrower's Organizational Documents or to the conduct of such Borrower's business or of any material contract or undertaking to which such Borrower is a party or by which such Borrower is bound, (b) will not conflict with or violate any law or regulation, or any judgment, order or decree of any Governmental Body, (c) will not require the Consent of any Governmental Body, any party to a material contract or any other Person, and (d) will not conflict with, nor result in any breach in any of the provisions of or constitute a default under or result in the creation of any Lien upon any asset of such Borrower (other than the liens created hereunder in favor of Lender) under the provisions of any agreement, instrument, or other document to which such Borrower is a party or by which it or its property is a party or by which it may be bound.

- 5.2. <u>Formation and Qualification</u>. Borrower is duly incorporated or formed, as applicable, and in good standing under the laws of the state in which it is organized and is qualified to do business and is in good standing in the states where it conducts business which constitute all states in which qualification and good standing are necessary for Borrower to conduct its business and own its property and where the failure to so qualify could reasonably be expected to have a Material Adverse Effect on such Borrower. Borrower has delivered to Lender true and complete copies of its Organizational Documents and will promptly notify Lender of any amendment or changes thereto.
- 5.3. <u>Survival of Representations and Warranties</u>. All representations and warranties of such Borrower contained in this Agreement and the Loan Documents to which it is a party shall be true at the time of such Borrower's execution of this Agreement and the Loan Documents to which it is a party, and shall survive the execution, delivery and acceptance thereof by the parties thereto and the closing of the transactions described therein or related thereto.
- 5.4. <u>Tax Returns</u>. Borrower has filed all federal, state and local tax returns, to the extent required, and other reports each is required by law to file and has paid all taxes, assessments, fees and other governmental charges that are due and payable. The provision for taxes on the books of Borrower is adequate for all years not closed by applicable statutes, and for its current fiscal year, and no Borrower has any knowledge of any deficiency or additional assessment in connection therewith not provided for on its books.
- 5.5. <u>Financial Statements</u>. The December 31, 2020, December 31, 2021 and year-to-date March 31, 2022 balance sheet of Borrower (the "<u>Balance Sheet</u>") furnished to Lender is accurate, complete and correct in all material respects and fairly reflects the financial condition of Borrower as of the Closing Date, and has been prepared in accordance with GAAP, consistently applied. The Balance Sheet has been certified as accurate, complete and correct in all material respects by the Chief Financial Officer of Borrower. All financial statements referred to in this

section 5.5, including the related schedules and notes thereto, have been prepared in accordance with GAAP, except as may be disclosed in such financial statements.

5.6. <u>Entity Names</u>. Except as disclosed to Lender, in writing, during the application process, Borrower has not been known by any other company or corporate name, as applicable, in the past five (5) years, nor has Borrower been the surviving corporation or company, as applicable, of a merger or consolidation or acquired all or substantially all of the assets of any Person during the preceding five (5) years.

## 5.7. Solvency; No Litigation, Violation, Indebtedness or Default; ERISA Compliance.

- (a) (i) Borrower is, and after giving effect to the Loan, Borrower will be solvent, able to pay its debts as they mature, and has, and after giving effect to the Loan, will have capital sufficient to carry on its business and all businesses in which it is about to engage, (ii) as of the Closing Date, the fair present saleable value of its assets, calculated on a going concern basis, is in excess of the amount of its liabilities, and (iii) subsequent to the Closing Date, the fair saleable value of its assets (calculated on a going concern basis) will be in excess of the amount of its liabilities.
- (b) Except as disclosed by Borrower to Lender in writing prior to the Closing Date, no Borrower has any pending or threatened litigation, arbitration, actions or proceedings. No Borrower has any outstanding Indebtedness other than the Obligations, except for (i) Indebtedness reflected on the Borrower's financial statements delivered to Lender, and (ii) Indebtedness incurred in the ordinary course of Borrower's business that is not secured by any lien, security interest other charge in or against the Collateral.
- (c) Borrower is not in violation of any applicable statute, law, rule, regulation or ordinance in any respect which could reasonably be expected to have a Material Adverse Effect, nor is Borrower in violation of any order of any court, Governmental Body or arbitration board or tribunal
- (d) The Borrower and each member of the Controlled Group have fulfilled their obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and have not incurred any liability to the PBGC or a Plan under Title IV of ERISA in connection with the termination of a Plan, and each Plan maintained by the Borrower is in compliance with the presently applicable provisions of ERISA and the Code.
- 5.8. <u>Licenses and Permits</u>. Borrower (a) is in compliance with and (b) has procured and is now in possession of, all material licenses or permits required by any applicable federal, state or local law, rule or regulation for the operation of its business in each jurisdiction wherein it is now conducting or proposes to conduct business and where the failure to procure such licenses or permits could reasonably be expected to have a Material Adverse Effect.
- 5.9. <u>Default of Indebtedness</u>. No Borrower is in default in the payment of the principal of or interest on any Indebtedness or under any instrument or agreement under or subject to which any Indebtedness has been issued and no event has occurred under the provisions of any such

instrument or agreement which with or without the lapse of time or the giving of notice, or both, constitutes or would constitute an event of default thereunder.

- 5.10. No Default. No Borrower is in default in the payment or performance of any of its contractual obligations and no Default or Event of Default has occurred.
- 5.11. <u>Disclosure</u>. No representation or warranty made by Borrower in this Agreement or in any financial statement, report, certificate or any other document furnished in connection herewith contains any untrue statement of a material fact or omits to state any material fact necessary to make the statements herein or therein not misleading. There is no fact known to Borrower or which reasonably should be known to such Borrower which such Borrower has not disclosed to Lender in writing with respect to the transactions contemplated by or this Agreement which could reasonably be expected to have a Material Adverse Effect.
- 5.12. <u>Business and Property of Borrower</u>. Upon and after the Closing Date, Borrower does not propose to engage in any business other than the business it currently operates as of the Effective Date, and activities incidental thereto or otherwise necessary to conduct the foregoing. On the Closing Date, Borrower will own all the property and possess all of the rights and Consents necessary for the conduct of the business of such Borrower.
- 5.13. <u>Certificate of Beneficial Ownership</u>. The Certificate of Beneficial Ownership executed and delivered to Lender for Borrower on or prior to the Closing Date, as updated from time to time in accordance with this Agreement, is accurate, complete and correct as of the date hereof and as of the date any such update is delivered. The Borrower acknowledges and agrees that the Certificate of Beneficial Ownership is one of the Loan Documents.
- 5.14 <u>Eligible Contract Participant</u>. Borrower qualifies as an Eligible Contract Participant.
  - 5.15 <u>Intentionally omitted.</u>

## VI. AFFIRMATIVE COVENANTS

Borrower shall, until payment in full of the Obligations and termination of this Agreement:

- 6.1. <u>Compliance with Laws</u>. Comply with all Applicable Laws with respect to the Collateral or any part thereof or to the operation of such Borrower's business the non-compliance with which could reasonably be expected to have a Material Adverse Effect (except to the extent any separate provision of this Agreement shall expressly require compliance with any particular Applicable Law(s) pursuant to another standard). Borrower may, however, contest or dispute any Applicable Laws in any reasonable manner, provided that any related Lien is inchoate or stayed and sufficient reserves are established to the reasonable satisfaction of Lender to protect Lender's Lien on or security interest in the Collateral.
- 6.2. <u>Conduct of Business and Maintenance of Existence and Assets.</u> (a) Conduct continuously and operate actively its business according to good business practices; (b) keep in full force and effect its existence and comply in all material respects with the laws and regulations

governing the conduct of its business where the failure to do so could reasonably be expected to have a Material Adverse Effect; and (c) make all such reports and pay all such franchise and other taxes and license fees and do all such other acts and things as may be lawfully required to maintain its rights, licenses, leases, powers and franchises under the laws of the United States or any political subdivision thereof where the failure to do so could reasonably be expected to have a Material Adverse Effect.

- 6.3. <u>Books and Records</u>. Keep proper books of record and account in which full, true and correct entries will be made of all dealings or transactions of or in relation to its business and affairs (including without limitation accruals for taxes), all in accordance with, or as required by, GAAP consistently applied in the opinion of such independent public accountant as shall then be regularly engaged by Borrower.
- 6.4. Payment of Taxes. Pay, when due, all taxes, assessments and other charges lawfully levied or assessed upon such Borrower or any of the Collateral, including personal property taxes, assessments and charges and all franchise, income, employment, social security benefits, withholding, and sales taxes. If any tax by any Governmental Body is or may be imposed on or as a result of any transaction between Borrower and Lender which Lender may be required to withhold or pay or if any taxes, assessments, or other charges remain unpaid after the date fixed for their payment, or if any claim shall be made which, in Lender's 's opinion, may possibly create a valid Lien on the Collateral, Lender may without notice to Borrower pay the taxes, assessments or other charges and Borrower hereby indemnifies and holds Lender harmless in respect thereof. The amount of any payment by Lender under this Section 6.4 shall be charged to Borrower' Account as an Advance and added to the Obligations and, until Borrower shall furnish Lender with an indemnity therefor (or supply Lender with evidence satisfactory to Lender that due provision for the payment thereof has been made), Lender may hold without interest any balance standing to Borrower' credit and Lender shall retain its security interest in and Lien on any and all Collateral held by Lender.

## 6.5. Financial Covenants

- (a) Adjusted Net Worth. Borrower shall maintain at all times an Adjusted Net Worth in an amount not less than \$350,000,000.00. Lender shall determine compliance quarterly based on Borrower's consolidated financial statements filed with the United States Securities and Exchange Commission (SEC). As used herein, the term "Adjusted Net Worth" means for Borrower the excess of its total assets over its total liabilities, with consideration for bitcoin at fair market value based on the date of such financial statement.
- (b) <u>Minimum Liquidity</u>. Borrower shall maintain minimum unrestricted and unencumbered cash of at least \$25,000,000.00, which shall be verified quarterly based on consolidated financial statements filed with the SEC for the preceding quarter.

## 6.6. Insurance.

(a) (i) Borrower shall maintain, at a minimum, commercial general liability insurance in an amount satisfactory to Lender, and such other insurance coverages, as is customary in the case of companies engaged in businesses similar to Borrower, and having deductibles

consistent with customary practice. Borrower shall require the Custodian to maintain coverage that insures against the theft, or the accidental or intentional loss or destruction of, the private keys associated with the Collateral, subject to certain exclusions from coverage that are generally applicable to all digital assets in the custody of the Custodian (the "Required Digital Assets Coverage"). During the time in which the Collateral remains in the custody of the Custodian and the Custodian maintains the Required Digital Asset Coverage, Borrower is not required to maintain separate coverage for losses covered by the Required Digital Assets Coverage.

- (b) With respect to any Required Digital Assets Coverage maintained by a Custodian, Borrower acknowledges that such coverages may not be sufficient to cover the entire loss in the event some or all of the Collateral is lost or stolen. Lender's acceptance of such insurance should not be considered as a representation, opinion or indication that the coverages provided by the Custodian are suitable for Borrower. In some cases, the amount of proceeds available on account of a loss may only be sufficient to repay the Advances and may not be sufficient to cover some or all of Borrower's equity in the Collateral. As such, Borrower should evaluate with its risk management personnel what insurance coverage it should maintain independent of those maintained by the Custodian.
- 6.7. <u>Payment of Indebtedness</u>. Pay, discharge or otherwise satisfy at or before maturity (subject, where applicable, to specified grace periods) all its Indebtedness, except when the failure to do so would not reasonably be expected to have a Material Adverse Effect.
- 6.8. <u>Standards of Financial Statements</u>. Cause all financial statements referred to in Section 9.2 as to which GAAP is applicable to be complete and correct in all material respects (subject, in the case of interim financial statements, to normal year-end audit adjustments) and to be prepared in reasonable detail and in accordance with GAAP applied consistently throughout the periods reflected therein (except as disclosed therein and agreed to by such reporting accountants or officer, as applicable).
- 6.9. <u>Execution of Supplemental Instruments.</u> Execute and deliver to Lender from time to time, upon demand, such supplemental agreements, statements, assignments and transfers, or instructions or documents relating to the Collateral, and such other instruments as Lender may request, in order that the full intent of this Agreement may be carried into effect.
- 6.10. Certificate of Beneficial Ownership and Other Additional Information. Provide to Lender: (i) confirmation of the accuracy of the information set forth in the most recent Certificate of Beneficial Ownership provided to the Lender; (ii) a new Certificate of Beneficial Ownership, in form and substance satisfactory to Lender when the individual(s) to be identified as a Beneficial Owner have changed; and (iii) such other information and documentation as may reasonably be requested by Lender from time to time for purposes of compliance by Lender with applicable laws (including without limitation the USA PATRIOT ACT and other "know your customer" and antimoney laundering rules and regulations), and any policy or procedure implemented by Lender or such Lender to comply therewith.
- 6.11 <u>Operational Security</u>. Borrower shall follow and maintain operational security best practices throughout the Term of the Loan, including, without limitation, ensuring that all communications initiated by Borrower to Lender and/or Custodian involving private or public keys

or other sensitive information are accomplished through secure means, such as encrypted email communication. Borrower shall employ adequate security measures to safeguard any passwords, personal identification numbers and other credentials that can be used to access information about the Loan, the Collateral and any deposits and/or withdrawals of digital assets involving the Loan.

## VII. NEGATIVE COVENANTS.

Borrower shall not, until satisfaction in full of the Obligations and termination of this Agreement:

- 7.1. <u>Sale of Collateral</u>. Sell, lease, transfer or otherwise dispose of any Collateral in violation of this Agreement.
- 7.2. <u>Creation of Liens</u>. Create or suffer to exist any Lien or transfer upon or against any of the Collateral in violation of this Agreement.
- 7.3. <u>Indebtedness</u>. Create, incur, assume or suffer to exist any Indebtedness, except (i) in the Ordinary Course of Business., and (ii) \$747,500,000.00 principal amount of 1.0% Convertible Senior Notes due 2026 and any restructurings thereof.
- 7.4. <u>Nature of Business</u>. Substantially change the nature of the business in which it is presently engaged, nor except as specifically permitted hereby purchase or invest, directly or indirectly, in any assets or property other than in the Ordinary Course of Business for assets or property which are useful in, necessary for and are to be used in its business as presently conducted.
- 7.5. <u>Fiscal Year and Accounting Changes</u>. Change its fiscal year end or make any change (a) in accounting treatment and reporting practices except as required by GAAP or (b) in tax reporting treatment except as required by law or with the prior consent of Lender, which will not be unreasonably withheld or delayed.
- 7.6. <u>Pledge of Credit</u>. Now or hereafter pledge Lender's credit on any purchases, commitments or contracts or for any purpose whatsoever or use any portion of any Advance in or for any business other than such Borrower's business operations as conducted on the Closing Date.
- 7.7. <u>Compliance with ERISA</u>. Cause, or permit any member of the Controlled Group to cause, a representation or warranty in Section 5.7(d) to cease to be true and correct.

### VIII. CONDITIONS PRECEDENT.

- 8.1. <u>Conditions to Initial Advances</u>. The agreement of Lender to make the initial Advance requested to be made on the Closing Date is subject to the satisfaction, or waiver by Lender, immediately prior to or concurrently with the making of such Advances, of the following conditions precedent:
- (a) <u>This Agreement</u>. Lender shall have received this Agreement duly executed and delivered by an authorized officer of Borrower;

- (b) <u>Note</u>. Lender shall have received the Note duly executed and delivered by an authorized officer of Borrower;
- (c) <u>Loan Documents</u>. Lender shall have received each of the executed Loan Documents, as applicable, including, without limitation, the Account Control Agreement in form and substance satisfactory to Lender;
- (d) <u>Borrowing Base</u>. Lender shall have received evidence from Custodian that the aggregate amount of Collateral in the Collateral Account is sufficient in value and amount, at or prior to closing, to support Advances in the amount requested by Borrower on the Closing Date;
- (e) Net Worth and Liquidity. Lender shall have received satisfactory evidence that Borrower meets the Net Worth and Minimum Liquidity covenants;
- (f) Filings, Registrations and Recordings. Each document (including any UCC financing statement) required by this Agreement, any related agreement or under law or reasonably requested by Lender to be filed, registered or recorded in order to create, in favor of Lender, a perfected security interest in or lien upon the Collateral shall have been properly filed, registered or recorded in each jurisdiction in which the filing, registration or recordation thereof is so required or requested, and Lender shall have received an acknowledgment copy, or other evidence satisfactory to it, of each such filing, registration or recordation and satisfactory evidence of the payment of any necessary fee, tax or expense relating thereto;
- Secretary's Certificates, Authorizing Resolutions and Good Standings of Borrower. Lender shall have received a certificate of the Secretary or Assistant Secretary (or other equivalent officer, partner or manager) of Borrower in form and substance satisfactory to Lender dated as of the Closing Date which shall certify (i) copies of resolutions in form and substance reasonably satisfactory to Lender, of the board of directors (or other equivalent governing body, member or partner) of such Borrower authorizing (x) the execution, delivery and performance of this Agreement, the Notes and each Loan Document to which such Borrower is a party (including authorization of the incurrence of indebtedness and borrowing of Revolving Advances), and (y) the granting by such Borrower of the security interests in and liens upon the Collateral to secure all of the Obligations of Borrower (and such certificate shall state that such resolutions have not been amended, modified, revoked or rescinded as of the date of such certificate), (ii) the incumbency and signature of the officers of such Borrower authorized to execute this Agreement and the Loan Documents, (iii) copies of the Organizational Documents of such Borrower as in effect on such date, complete with all amendments thereto, and (iv) the good standing (or equivalent status) of such Borrower in its jurisdiction of organization and each applicable jurisdiction where the conduct of such Borrower's business activities or the ownership of its properties necessitates qualification, as evidenced by good standing certificate(s) (or the equivalent thereof issued by any applicable jurisdiction) dated not more than thirty (30) days prior to the Closing Date, issued by the Secretary of State or other appropriate official of each such jurisdiction;
- (h) <u>Legal Opinion</u>. Lender shall have received in form and substance satisfactory to Lender, a legal opinion from Borrower's legal counsel as to good standing, due

authorization, enforceability as such other matters as customarily required for transaction in the nature of the Loan.

- (i) No Litigation. No litigation, investigation or proceeding before or by any arbitrator or Governmental Body shall be continuing or threatened against Borrower or against the officers or directors of Borrower (A) in connection with this Agreement, the Note, the Loan Documents or any of the transactions contemplated thereby and which, in the reasonable opinion of Lender, is deemed material or (B) which could, in the reasonable opinion of Lender, have a Material Adverse Effect; and no injunction, writ, restraining order or other order of any nature materially adverse to Borrower or the conduct of its business or inconsistent with the due consummation of the transactions contemplated by this Agreement shall have been issued by any Governmental Body;
- (j) <u>Fees.</u> Lender shall have received all fees payable to Lender on or prior to the Closing Date hereunder, including pursuant to Article III hereof;
- (k) <u>Insurance</u>. Subject to the provisions of Section 6.6(a), Lender shall have received in form and substance satisfactory to Lender, (i) evidence that adequate insurance required to be maintained under this Agreement is in full force and effect, (ii) insurance certificates issued by Borrower's insurance broker containing such information regarding Borrower's insurance policies as Lender shall request.
  - (1) Request for Advance. A properly completed Request for Advance; and
- (m) <u>Other</u>. All corporate and other proceedings, and all documents, instruments and other legal matters in connection with the transactions contemplated by this Agreement shall be satisfactory in form and substance to Lender and its counsel.
- 8.2. <u>Conditions to Each Advance</u>. The agreement of Lender to make any Advance requested to be made on any date (including the initial Advance), is subject to the satisfaction of the following conditions precedent as of the date such Advance is made:
- (a) Representations and Warranties. Each of the representations and warranties made by Borrower in or pursuant to this Agreement, the Loan Documents and any related agreements to which it is a party, and each of the representations and warranties contained in any certificate, document or financial or other statement furnished at any time under or in connection with this Agreement, the Loan Documents or any related agreement shall be true and correct in all respects on and as of such date as if made on and as of such date (except to the extent any such representation or warranty expressly relates only to any earlier and/or specified date);
  - (b) Request for Advance. A properly completed Request for Advance;
- (c) No Default. No Event of Default or Default shall have occurred and be continuing on such date, or would exist after giving effect to the Advances requested to be made, on such date; provided, however that Lender, in its sole discretion, may continue to make Advances notwithstanding the existence of an Event of Default or Default and that any Advances so made shall not be deemed a waiver of any such Event of Default or Default; and

(d) <u>Maximum Advances</u>. After giving effect to the requested Advance, the aggregate amount of Advances shall not exceed the Maximum Advance Amount.

Each request for an Advance by Borrower hereunder shall constitute a representation and warranty by Borrower as of the date of such Advance that the conditions contained in this Section shall have been satisfied.

## IX. INFORMATION AS TO BORROWER

Borrower shall, until satisfaction in full of the Obligations and the termination of this Agreement:

- 9.1. <u>Disclosure of Material Matters</u>. Immediately upon learning thereof, report to Lender all matters materially affecting the value, enforceability or collectability of any portion of the Collateral, or any Lien other than Lender's security interest, placed upon or asserted against Borrower or any Collateral.
- 9.2. <u>Financial Statements.</u> Borrower shall, in accordance with Section 13 or 15(d) of the Securities and Exchange Act of 1934, file with the SEC: (a) an annual financial report on Form 10-K within ninety (90) days after the end of each of its fiscal years, and (b) quarterly financial reports on Form 10-Q forty-five (45) days after the end of each of its first three fiscal quarters, each of the preceding dates a "Reporting Due Date." The Borrower's Reporting Due Date for any Form 10-K or Form 10-Q may be extended for only such time as permitted by the SEC if the Borrower shall file with the SEC an accurate and complete SEC Form NT 10-K or NT 10-Q, as applicable, prior to each respective Reporting Due Date and provide a copy of such Form NT to Lender immediately after filing.

### 9.3. <u>Intentionally Omitted.</u>

9.4. <u>Collateral Account Reports.</u> At or prior to 1:00 p.m. PT each day during an Availability Period, the Borrower shall make available a report to the Lender that sets forth the total amount of bitcoin subject to the Collateral Account, in each case as of 1:00 p.m. PT on the immediately preceding day. Notwithstanding the foregoing, Borrower will not be required to submit such reports for so long as Lender does not provide written notice to Borrower that the applicable Custodian is no longer providing such reports directly to Lender.

## 9.5. <u>Intentionally Omitted.</u>

- 9.6. <u>Additional Information</u>. Furnish Lender with such additional information as Lender shall reasonably request in order to enable Lender to determine whether the terms, covenants, provisions and conditions of this Agreement and the Notes have been complied with by Borrower including, without the necessity of any request by Lender, copies of all security audits and reviews.
- 9.7. <u>Notice of Suits, Adverse Events</u>. Furnish Lender with prompt written notice of (a) any lapse or other termination of any Consent issued to Borrower by any Governmental Body or any other Person that is material to the operation of Borrower's business, (b) any refusal by any

Governmental Body or any other Person to renew or extend any such Consent; and (c) copies of any periodic or special reports filed by Borrower with any Governmental Body or Person, if such reports indicate any material change in the business, operations, affairs or condition of Borrower, or if copies thereof are requested by Lender, and (d) copies of any material notices and other communications from any Governmental Body or Person which specifically relate to Borrower.

9.8. <u>Financial Disclosure</u>. Borrower hereby irrevocably authorizes and directs all accountants and auditors employed by such Borrower at any time during the Term to exhibit and deliver to Lender copies of any of such Borrower's financial statements, trial balances or other accounting records of any sort in the accountant's or auditor's possession, and to disclose to Lender any information such accountants may have concerning such Borrower's financial status and business operations. Borrower hereby authorizes all Governmental Bodies to furnish to Lender copies of reports or examinations relating to such Borrower, whether made by such Borrower or otherwise; however, Lender will attempt to obtain such information or materials directly from such Borrower prior to obtaining such information or materials from such accountants or Governmental Bodies.

### X. EVENTS OF DEFAULT.

The occurrence of any one or more of the following events shall constitute an "Event of Default":

- 10.1. Nonpayment. Failure by Borrower to pay (a) within ten (10) days of the date due, any scheduled installment of principal or interest on the Obligations, or (b) on or before the date due, any other fee, charge, amount or liability provided for herein, including, without limitation, any amounts owed under Section 2.4, or in any Loan Document, in each case whether at maturity, by reason of acceleration pursuant to the terms of this Agreement, by notice of intention to prepay or by required prepayment.
- 10.2. <u>Breach of Representation</u>. Any representation or warranty made or deemed made by Borrower in this Agreement, any Loan Document or any related agreement or in any certificate, document or financial or other statement furnished at any time in connection herewith or therewith shall prove to have been incorrect or misleading in any material respect on the date when made or deemed to have been made; provided that, in each case, such inaccuracies, to the extent capable of being corrected, are not corrected within ten (10) days following notice from Lender.
- 10.3. <u>Financial Information</u>. Failure by Borrower to (a) file financial reports in accordance with Section 9.2, or (b) provide Lender with financial information it may request within three (3) Business Days of such requests or (c) permit the inspection of its books or records in accordance with this Agreement;
- 10.4. <u>Judicial Actions</u>. Issuance of a notice of Lien, levy, assessment, injunction or attachment (a) against any Collateral or (b) against a material portion of Borrower's other property which is not stayed or lifted within thirty (30) days;
- 10.5. <u>Noncompliance</u>. Except as otherwise provided for in Sections 10.1, 10.2, 10.3 and 10.4, (a) failure or neglect of Borrower, or any Person to perform, keep or observe any term,

provision, condition, covenant herein contained, or contained in any Loan Document or any other agreement or arrangement, now or hereafter entered into between Borrower, or such Person, and Lender beyond any applicable cure and/or grace period;

- 10.6. <u>Judgments</u>. Any (a) judgment or judgments, writ(s), order(s) or decree(s) for the payment of money are rendered against Borrower for an aggregate amount in excess of \$500,000.00 and (b) (i) action shall be legally taken by any judgment creditor to levy upon assets or properties of Borrower to enforce any such judgment, (ii) such judgment shall remain undischarged for a period of thirty (30) consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, shall not be in effect, or (iii) any Liens arising by virtue of the rendition, entry or issuance of such judgment upon assets or properties of Borrower shall be senior to any Liens in favor of Lender on such assets or properties;
- 10.7. <u>Bankruptcy</u>. Borrower, any Subsidiary or Affiliate of Borrower shall (a) apply for, consent to or suffer the appointment of, or the taking of possession by, a receiver, custodian, trustee, liquidator or similar fiduciary of itself or of all or a substantial part of its property, (b) admit in writing its inability, or be generally unable, to pay its debts as they become due or cease operations of its present business, (c) make a general assignment for the benefit of creditors, (d) commence a voluntary case under any state or federal bankruptcy or receivership laws (as now or hereafter in effect), (e) be the subject of an Insolvency Event or otherwise be adjudicated a bankrupt or insolvent (including by entry of any order for relief in any involuntary bankruptcy or insolvency proceeding commenced against it), (f) file a petition seeking to take advantage of any other law providing for the relief of debtors, (g) acquiesce to, or fail to have dismissed, within thirty (30) days, any petition filed against it in any involuntary case under such bankruptcy laws, or (h) take any action for the purpose of effecting any of the foregoing;
- 10.8. <u>Material Adverse Effect</u>. The occurrence of any event or development which could reasonably be expected to have a Material Adverse Effect;
- 10.9. <u>Lien Priority</u>. Any Lien created hereunder or provided for hereby or under any related agreement for any reason ceases to be or is not a valid and perfected Lien having a first priority interest in Collateral;
- 10.10. Cross Default. Either (a) any specified "event of default" under any Indebtedness (other than the Obligations) of Borrower with a then-outstanding principal balance (or, in the case of any Indebtedness not so denominated, with a then-outstanding total obligation amount) of \$500,000.00 or more, or any other event or circumstance which would permit the holder of any such Indebtedness of Borrower to accelerate such Indebtedness (and/or the obligations of Borrower thereunder) prior to the scheduled maturity or termination thereof, shall occur (regardless of whether the holder of such Indebtedness shall actually accelerate, terminate or otherwise exercise any rights or remedies with respect to such Indebtedness), (b) a default of the obligations of Borrower under any other agreement to which it is a party shall occur which has or is reasonably likely to have a Material Adverse Effect, or (c) Lender declares a default or "event of default" under any other loan or extension of credit or other account agreement to Borrower or an Affiliate of Borrower;

### 10.11. Intentionally Omitted.

- 10.12. <u>Change of Control</u>. Any Change of Control shall occur without Borrower's prior written notice to Lender and Lender's prior written approval thereof.
- 10.13. <u>Invalidity</u>. Any material provision of this Agreement or any Loan Document shall, for any reason, cease to be valid and binding on Borrower, or Borrower shall so claim in writing to Lender or Borrower challenges the validity of or its liability under this Agreement or any Loan Document;
- 10.14. <u>Seizures</u>. Any portion of the Collateral shall be seized, subject to garnishment or taken by a Governmental Body;
- 10.15. <u>Anti-Money Laundering/International Trade Law Compliance</u>. Any representation or warranty contained in Section 13.16 is or becomes false or misleading at any time.

### XI. LENDER'S RIGHTS AND REMEDIES AFTER DEFAULT.

## 11.1. Rights and Remedies.

- During the continuance of: (i) an Event of Default pursuant to Section 10.7 (other than Section 10.7(g)), all Obligations shall be immediately due and payable accruing interest at the Default Rate, and this Agreement and the obligation of Lender to make Advances shall be deemed terminated, (ii) any of the other Events of Default and at any time thereafter, at the option of Lender, shall be immediately due and payable and Lender shall have the right to terminate this Agreement and to terminate the obligation of Lender to make Advances; and (iii) without limiting Section 8.2 hereof, any Default under Sections 10.7(g) hereof, the obligation of Lender to make Advances hereunder shall be suspended until such time as such involuntary petition shall be dismissed. Upon the occurrence of any Event of Default, Lender shall have the right to exercise any and all rights and remedies provided for herein, under the Loan Documents, under the Uniform Commercial Code and at law or equity generally, including the right to foreclose the security interests granted herein and to realize upon any Collateral by any available judicial or non-judicial procedure and/or to take possession of and sell any or all of the Collateral with or without judicial process. With or without having the Collateral at the time or place of sale, Lender may sell the Collateral, or any part thereof, at public or private sale, at any time or place, in one or more sales, at such price or prices, and upon such terms, either for cash, credit or future delivery, as Lender may elect. The cash proceeds realized from the sale of any Collateral shall be applied to the Obligations in the order determined by Lender in its reasonable discretion. Noncash proceeds will only be applied to the Obligations as they are converted into cash. If any deficiency shall arise, Borrower shall remain liable to Lender therefor. Notwithstanding anything to the contrary contained herein, Lender may sell, or direct any Custodian to sell, any bitcoin, Forked Assets or other digital assets in the Collateral Account without (i) any notice to Borrower, (ii) any publication of notice about a sale of such assets, it being acknowledged by Borrower that highly liquid and efficient markets exist for the sale of bitcoin, or (iii) any obligation to account to Borrower for any losses caused by Lender's decision to liquidate such Collateral.
- (b) To the extent that Applicable Law imposes duties on Lender to exercise remedies in a commercially reasonable manner, Borrower acknowledges and agrees that it is not commercially unreasonable for Lender, within customary and usual practices in the commercial

banking industry in the U.S.: (i) to fail to incur expenses reasonably deemed significant by Lender to safeguard the Collateral pending disposition; (ii) to fail to obtain third party consents for access to Collateral to be disposed of, or to obtain or, if not required by other law, to fail to obtain governmental or third party consents for the collection or disposition of Collateral to be collected or disposed of; (iii) to fail to remove Liens on or any adverse claims against Collateral; (iv) if required under the UCC, to advertise dispositions of Collateral through publications or media of general circulation, whether or not the Collateral is of a specialized nature; (v) to contact other Persons, whether or not in the same business as Borrower, for expressions of interest in acquiring all or any portion of such Collateral; (vi) to hire one or more market makers and/or brokers specializing in digital asset transactions to assist in the disposition of Collateral; (vii) to dispose of Collateral by utilizing internet sites that provide for the auction of assets of the types included in the Collateral or that have the reasonable capacity of doing so, or that match buyers and sellers of assets; (viii) to dispose of assets in wholesale rather than retail markets; (ix) to disclaim disposition warranties, such as title, possession or quiet enjoyment, (x) to purchase insurance or credit enhancements to insure Lender against risks of loss, collection or disposition of Collateral or to provide to Lender a guaranteed return from the collection or disposition of Collateral; or (xi) to the extent deemed appropriate by the Lender, to obtain the services of other brokers, investment bankers, consultants and other professionals to assist Lender in the collection or disposition of any of the Collateral. Borrower acknowledges that the purpose of this Section 11.1(b) is to provide non-exhaustive indications of what actions or omissions by Lender would not be commercially unreasonable in Lender's exercise of remedies against the Collateral and that other actions or omissions by Lender shall not be deemed commercially unreasonable solely on account of not being indicated in this Section 11.1(b). Upon the occurrence of any Event of Default, Lender shall have the right to exercise any and all rights and remedies provided for herein, under the Loan Documents, under the Uniform Commercial Code and at law or equity generally, including the right to foreclose the security interests granted herein and to realize upon any Collateral by any available judicial procedure and/or to take possession of and sell any or all of the Collateral with or without judicial process. Without limitation upon the foregoing, nothing contained in this Section 11.1(b) shall be construed to grant any rights to Borrower or to impose any duties on Lender that would not have been granted or imposed by this Agreement or by Applicable Law in the absence of this Section 11.1(b).

- 11.2. <u>Lender's Discretion</u>. Lender shall have the right in its commercially reasonable discretion to determine which rights, Liens, security interests or remedies Lender may at any time pursue, relinquish, subordinate, or modify, which procedures, timing and methodologies to employ, and what any other action to take with respect to any or all of the Collateral and in what order, thereto and such determination will not in any way modify or affect any of Lender's rights hereunder as against Borrower or each other.
- 11.3. <u>Setoff.</u> In addition to any other rights which Lender may have under Applicable Law, upon the occurrence of an Event of Default hereunder, Lender shall have a right, immediately and without notice of any kind, to apply Borrower's property held by Lender or any of its Affiliates to reduce the Obligations and to exercise any and all rights of setoff which may be available to Lender with respect to any deposits held by Lender.

- 11.4. <u>Liquidation without Notice</u>. Borrower acknowledges that the price of bitcoin is (a) volatile and thus may decline speedily in value, and (b) bitcoin is a type of asset customarily sold on a recognized market. Accordingly, Borrower acknowledges and agrees that it would not be necessary under Section 9-611(b) of the UCC to give notice of any proposed disposition of the Collateral.
- 11.5. <u>Rights and Remedies not Exclusive</u>. The enumeration of the foregoing rights and remedies is not intended to be exhaustive and the exercise of any rights or remedy shall not preclude the exercise of any other right or remedies provided for herein or otherwise provided by law, all of which shall be cumulative and not alternative.

### XII. WAIVERS AND JUDICIAL PROCEEDINGS.

- 12.1. <u>Waiver of Notice</u>. Borrower hereby waives notice of non-payment of any indebtedness, demand, presentment, protest and notice thereof with respect to any and all instruments, notice of acceptance hereof, notice of loans or advances made, credit extended, Collateral received or delivered, or any other action taken in reliance hereon, and all other demands and notices of any description, except such as are expressly provided for herein.
- 12.2. <u>Delay</u>. No delay or omission on Lender's 's part in exercising any right, remedy or option shall operate as a waiver of such or any other right, remedy or option or of any Default or Event of Default.

### XIII. EFFECTIVE DATE AND TERMINATION.

- 13.1. Term. This Agreement, which shall inure to the benefit of and shall be binding upon the respective successors and permitted assigns of Borrower and Lender shall become effective on the date hereof and shall continue in full force and effect until August 5, 2024 (the "Term") unless sooner terminated as herein provided. On the condition that the Extension Conditions are met, Borrower may request Lender to extend the term of the Loan for successive twelve (12) month periods, which Lender may grant or withhold in its sole and absolute discretion. For the avoidance of doubt, Lender may or may not grant such extension request for any such reason Lender determines, in its sole and absolute discretion, and Borrower shall not rely on Lender to grant any such approval of an extension. As used herein, the term "Extension Conditions" shall mean (i) no default or Event of Default has occurred, (ii) Borrower has provided Lender at least sixty (60) day prior written notice of its request to renew, (iii) Lender has reviewed and approved Borrower's creditworthiness and collateral requirements according to Lender's internal credit policies, (iv) Borrower recertifies that there has been no change in Beneficial Ownership since the inception of the Loan, or in the case of any changes, has provided Lender with such information about the then current Beneficial Ownership as Lender requests, and (v) the Borrower pays any required fees including but not limited to legal fees, commitment fees, and other loan related fees. Borrower may terminate this Agreement at any time upon ninety (90) days prior written notice to Lender upon payment in full of the Obligations.
- 13.2. <u>Termination</u>. The termination of the Agreement shall not affect Lender's rights, or any of the Obligations having their inception prior to the effective date of such termination or any Obligations which pursuant to the terms hereof continue to accrue after such date, and the

provisions hereof shall continue to be fully operative until all transactions entered into, rights or interests created and Obligations have been fully and indefeasibly paid, disposed of, concluded or liquidated. The security interests, Liens and rights granted to Lender hereunder and the financing statements filed hereunder shall continue in full force and effect, notwithstanding the termination of this Agreement or the fact that Borrower's account may from time to time be temporarily in a zero or credit position, until all of the Obligations of Borrower have been indefeasibly paid. Accordingly, Borrower waives any rights which it may have under the Uniform Commercial Code to demand the filing of termination statements with respect to the Collateral, and Lender shall not be required to send such termination statements to Borrower, or to file them with any filing office, unless and until this Agreement shall have been terminated in accordance with its terms and all Obligations have been indefeasibly paid in full in immediately available funds. All representations, warranties, covenants, waivers and agreements contained herein shall survive termination hereof until all Obligations are indefeasibly paid and performed in full.

13.3. Governing Law. This Agreement and each Loan Document (unless and except to the extent expressly provided otherwise in any such Loan Document), and all matters relating hereto or thereto or arising herefrom or therefrom (whether arising under contract law, tort law or otherwise) shall be governed by and construed in accordance with the laws of the State of New York applied to contracts to be performed wholly within the State of New York. Any judicial proceeding brought by or against Borrower with respect to any of the Obligations, this Agreement, the Loan Documents or any related agreement may be brought in any court of competent jurisdiction in the State of California or New York, United States of America, and, by execution and delivery of this Agreement, Borrower accepts for itself and in connection with its properties, generally and unconditionally, the non-exclusive jurisdiction of the aforesaid courts, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement. Borrower hereby waives personal service of any and all process upon it and consents that all such service of process may be made by certified or registered mail (return receipt requested) directed to Borrower at its address set forth in this Agreement and service so made shall be deemed completed five (5) days after the same shall have been so deposited in the mails of the United States of America, or, at Lender's option, by service upon Borrower which Borrower irrevocably appoints as such Borrower's agent for the purpose of accepting service within the State of New York or California, as applicable. Nothing herein shall affect the right to serve process in any manner permitted by law or shall limit the right of Lender to bring proceedings against Borrower in the courts of any other jurisdiction. Borrower waives any objection to jurisdiction and venue of any action instituted hereunder and shall not assert any defense based on lack of jurisdiction or venue or based upon forum non conveniens. Borrower waives the right to remove any judicial proceeding brought against such Borrower in any state court to any federal court. Any judicial proceeding by Borrower against Lender involving, directly or indirectly, any matter or claim in any way arising out of, related to or connected with this Agreement or any related agreement, shall be brought only in a federal or state court located in New York or California, at Lender's discretion.

### 13.4. Entire Understanding.

(a) THIS AGREEMENT AND THE DOCUMENTS EXECUTED CONCURRENTLY HEREWITH CONTAIN THE ENTIRE UNDERSTANDING BETWEEN

BORROWER AND LENDER AND SUPERSEDES ALL PRIOR AGREEMENTS AND UNDERSTANDINGS, IF ANY, RELATING TO THE SUBJECT MATTER OF THIS AGREEMENT ANY PROMISES, REPRESENTATIONS, WARRANTIES OR GUARANTEES NOT HEREIN CONTAINED AND HEREINAFTER MADE SHALL HAVE NO FORCE AND EFFECT UNLESS IN WRITING, SIGNED BY BORROWER'S AND LENDER'S RESPECTIVE OFFICERS. Neither this Agreement nor any portion or provisions hereof may be changed, modified, amended, waived, supplemented, discharged, cancelled or terminated orally or by any course of dealing, or in any manner other than by an agreement in writing, signed by the party to be charged. Notwithstanding the foregoing, Lender may modify this Agreement or any of the Loan Documents for the purposes of completing missing content or correcting erroneous content of an administrative nature, without the need for a written amendment, provided that the Lender shall send a copy of any such modification to Borrower (which copy may be provided by electronic mail). Borrower acknowledges that it has been advised by counsel in connection with the execution of this Agreement and Loan Documents and is not relying upon oral representations or statements inconsistent with the terms and provisions of this Agreement.

## 13.5. Successors and Assigns; Participations.

- (a) This Agreement shall be binding upon and inure to the benefit of Borrower and Lender, all future holders of the Obligations and their respective successors and assigns, except that Borrower may not assign or transfer any of its rights or obligations under this Agreement.
- (b) Borrower acknowledges that in the regular course of commercial banking business Lender may at any time and from time to time sell participating interests in the Advances to other Persons. Each Participant may exercise all rights of payment (including rights of set-off) with respect to the portion of such Advances held by it or other Obligations payable hereunder as fully as if such Participant were the direct holder thereof provided that (i) Borrower shall not be required to pay to any Participant more than the amount which it would have been required to pay to Lender which granted an interest in its Advances or other Obligations payable hereunder to such Participant had Lender retained such interest in the Advances hereunder or other Obligations payable hereunder unless the sale of the participation to such Participant is made with Borrower's prior written consent, and (ii) in no event shall Borrower be required to pay any such amount arising from the same circumstances and with respect to the same Advances or other Obligations payable hereunder to both Lender and such Participant. Borrower hereby grants to any Participant a continuing security interest in any deposits, moneys or other property actually or constructively held by such Participant as security for the Participant's interest in the Advances.
- 13.6. Notice. Any notice or request hereunder may be given to Borrower or to Lender at their respective addresses set forth below or at such other address as may hereafter be specified in a notice designated as a notice of change of address under this Section. Any notice, request, demand, direction or other communication (for purposes of this Section 13.6 only, a "Notice") to be given to or made upon any party hereto under any provision of this Agreement shall be given or made by telephone or in writing (which includes by means of electronic transmission (i.e., "email") or facsimile transmission or by setting forth such Notice on a website to which Borrower are directed (an "Internet Posting") if Notice of such Internet Posting (including the information necessary to access such site) has previously been delivered to the applicable parties hereto by

another means set forth in this Section 13.6) in accordance with this Section 13.6. Any such Notice must be delivered to the applicable parties hereto at the addresses and numbers set forth under their respective names in this Section 13.6 hereof or in accordance with any subsequent unrevoked Notice from any such party that is given in accordance with this Section 13.6. Any Notice shall be effective:

- (a) In the case of hand-delivery, when delivered;
- (b) If given by mail, four (4) days after such Notice is deposited with the United States Postal Service, with first-class postage prepaid, return receipt requested;
- (c) In the case of a telephonic Notice, when a party is contacted by telephone, if delivery of such telephonic Notice is confirmed no later than the next Business Day by hand delivery, a facsimile or electronic transmission, an Internet Posting or an overnight courier delivery of a confirmatory Notice (received at or before noon on such next Business Day);
- (d) In the case of a facsimile transmission, when sent to the applicable party's facsimile machine's telephone number, if the party sending such Notice receives confirmation of the delivery thereof from its own facsimile machine;
  - (e) In the case of electronic transmission, when actually received;
- (f) In the case of an Internet Posting, upon delivery of a Notice of such posting (including the information necessary to access such site) by another means set forth in this Section 13.6; and
- (g) If given by any other means (including by overnight courier), when actually received.

### (A) If to Lender at:

Silvergate Bank 4250 Executive Square, Suite 300 La Jolla, Ca 92037

Attention: Loan Servicing Telephone: (858) 362-6300 Facsimile: (858) 362-6323

Email: <u>loanservicing@silvergate.com</u>

# (B) If to Borrower:

Marathon Digital Holdings, Inc. 1180 North Town Center Drive, Suite 100 Las Vegas, NV 89144 Attention: Fred Thiel

35

Revolving Credit and Security Agreement Ref. No. 7465

Telephone: (725) 218-3415 Email: fred@marathondh.com

- 13.7. <u>Survival</u>. The obligations of Borrower under this Agreement shall survive termination of this Agreement and the Loan Documents and payment in full of the Obligations.
- 13.8. <u>Severability</u>. If any part of this Agreement is contrary to, prohibited by, or deemed invalid under Applicable Laws, such provision shall be inapplicable and deemed omitted to the extent so contrary, prohibited or invalid, but the remainder hereof shall not be invalidated thereby and shall be given effect so far as possible.
- 13.9. Expenses. Borrower shall pay all out-of-pocket expenses incurred by Lender and its Affiliates (including the reasonable fees, charges and disbursements of counsel for Lender), in connection with the preparation, negotiation, execution, delivery and administration of this Agreement and the Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated). Notwithstanding the forgoing, Borrower shall only pay out-of-pocket expenses in excess of \$10,000.00.
- 13.10. <u>Injunctive Relief.</u> Borrower recognizes that, in the event Borrower fails to perform, observe or discharge any of its obligations or liabilities under this Agreement, or threatens to fail to perform, observe or discharge such obligations or liabilities, any remedy at law may prove to be inadequate relief to Lender; therefor, Lender, if Lender so requests, shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving that actual damages are not an adequate remedy.

## 13.11 Limitations of Liability.

- (a) Neither Lender, nor any agent or attorney for Lender, shall be liable to Borrower (or any Affiliate of any such Person) for indirect, punitive, exemplary or consequential damages arising from any breach of contract, tort or other wrong relating to the establishment, administration or collection of the Obligations or as a result of any transaction contemplated under this Agreement or any Loan Document.
- (b) Notwithstanding anything to the contrary contained in this Agreement, Lender's liability will not exceed the greater of (i) the fair market value of the amount of the Collateral at the time in which the events giving rise to the liability occurred and (ii) the fair market value of the amount of Collateral at the time that Lender notifies Borrower in writing or Borrower otherwise has knowledge of the events giving rise to the liability. The fair market value of each digital asset held as Collateral will be determined by Lender according to the valuation policy used by the Custodian, which may differ from the way the Borrower values its digital asset holdings.
- 13.12. <u>Captions</u>. The captions at various places in this Agreement are intended for convenience only and do not constitute and shall not be interpreted as part of this Agreement.

- 13.13. <u>Counterparts; Facsimile Signatures</u>. This Agreement may be executed in any number of and by different parties hereto on separate counterparts, all of which, when so executed, shall be deemed an original, but all such counterparts shall constitute one and the same agreement. Any signature delivered by a party by facsimile or other form of electronic transmission (including email transmission of a PDF image or the use of a third-party platform, including DocuSign) shall be deemed to be an original signature hereto.
- 13.14. <u>Construction</u>. The parties acknowledge that each party and its counsel have reviewed this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendments, schedules or exhibits thereto.
- 13.15. <u>Publicity</u>. Borrower hereby authorizes Lender to make appropriate announcements of the financial arrangement entered into among Borrower and Lender, including announcements which are commonly known as tombstones, in such publications and to such selected parties as Lender shall in its reasonable discretion deem appropriate.

### 13.16. Anti-Terrorism Laws.

- (a) Borrower represents and warrants that (i) no Covered Entity is a Sanctioned Person and (ii) no Covered Entity, either in its own right or through any third party, (A) has any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Terrorism Law; (B) does business in or with, or derives any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law; or (C) engages in any dealings or transactions prohibited by any Anti-Terrorism Law.
- (b) Borrower covenants and agrees that (i) no Covered Entity will become a Sanctioned Person, (ii) no Covered Entity, either in its own right or through any third party, will (A) have any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Terrorism Law; (B) do business in or with, or derive any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law; (C) engage in any dealings or transactions prohibited by any Anti-Terrorism Law or (D) use the Advances to fund any operations in, finance any investments or activities in, or, make any payments to, a Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law, (iii) the funds used to repay the Obligations will not be derived from any unlawful activity, (iv) each Covered Entity shall comply with all Anti-Terrorism Laws and (v) the Borrower shall promptly notify the Lender in writing upon the occurrence of a Reportable Compliance Event.

## 13.17 Enforcement Waivers.

TO THE FULLEST EXTENT PERMITTED BY LAW, BORROWER HEREBY KNOWINGLY, INTENTIONALLY, AND VOLUNTARILY, WITH AND UPON THE ADVICE OF COMPETENT COUNSEL, AGREES NOT TO ELECT A TRIAL BY JURY OF ANY ISSUE TRIABLE OF RIGHT BY JURY, AND WAIVES ANY RIGHT TO TRIAL BY JURY FULLY TO THE EXTENT THAT ANY SUCH RIGHT SHALL NOW OR

HEREAFTER EXIST WITH REGARD TO THIS AGREEMENT, OR THE OTHER LOAN DOCUMENTS, OR ANY CLAIM, COUNTERCLAIM, OR OTHER ACTION ARISING IN CONNECTION THEREWITH. THIS WAIVER OF RIGHT TO TRIAL BY JURY IS INTENDED TO ENCOMPASS INDIVIDUALLY EACH INSTANCE AND EACH ISSUE AS TO WHICH THE RIGHT TO A TRIAL BY JURY WOULD OTHERWISE ACCRUE. LENDER IS HEREBY AUTHORIZED TO FILE A COPY OF THIS SECTION IN ANY PROCEEDING AS CONCLUSIVE EVIDENCE OF THIS WAIVER BY BORROWER.

Notwithstanding the foregoing to the contrary, in the event that the jury trial waiver contained herein shall be held or deemed to be unenforceable, Borrower hereby agrees that any controversy, dispute, or claim between the parties arising out of or relating to this Agreement shall be resolved by a reference proceeding in California in accordance with the provisions of California Code of Civil Procedure § 638. The referee shall be a retired California state court judge selected by mutual written agreement of the parties. If the parties are unable to agree upon a referee within ten (10) calendar days after one party serves a written notice of its intent to commence a judicial reference proceeding on the other party, then the referee will be selected by the court in accordance with California Code of Civil Procedure § 640(b). The referee shall be appointed to sit as a temporary judge, with all of the powers of a temporary judge, as authorized by law, and upon selection should take and subscribe to the oath of office as provided for in Rule 244 of the California Rules of Court (or any subsequently enacted Rule). The referee shall determine the manner in which the reference proceeding is conducted including the time and place of all hearings, the order of presentation of evidence, and all other questions that arise with respect to the course of the reference proceeding. The referee shall render a written statement of decision and shall conduct the proceedings in accordance with the California Code of Civil Procedure, the California Rules of Court, and the California Evidence Code, except as otherwise specifically agreed by the parties and approved by the referee. The referee's statement of decision shall set forth findings of fact and conclusions of law. The referee's decision shall be entered as a judgment in the court in accordance with the provisions of California Code of Civil Procedure §§ 644 and 645, and shall be appealable in accordance with California law.

Nothing in this Agreement shall be deemed to apply to or limit Lender's right to: (i) exercise self-help remedies such as (but not limited to) setoff, (ii) foreclose judicially or nonjudicially against any real or personal property collateral, or to exercise judicial or nonjudicial power of sale rights; (iii) obtain from a court provisional or ancillary remedies (including, without limitation, injunctive relief, a writ of possession, prejudgment attachment, a protective order, or the appointment of a receiver); or (iv) pursue its rights against any Person in a third-party proceeding in any action brought against Lender (including, without limitation, actions in bankruptcy court). Neither the exercise of any self-help remedies nor the institution or maintenance of an action for foreclosure or provisional or ancillary remedies, or the opposition to any such provisional remedies, shall constitute a waiver of the right of any party, including, without limitation, the claimant in any such action, to require submission to judicial reference the merits of the dispute occasioning resort to such remedies. No provision in the Loan Documents regarding submission to jurisdiction or venue in any court is intended to or shall be construed to be in derogation of the foregoing general judicial reference.

The foregoing judicial reference procedure constitutes a full and complete waiver of the right to a trial by jury that the parties may otherwise have and this waiver and judicial reference procedure is a material consideration to each party hereto.

If the enabling legislation which provides for appointment of a referee is repealed (and no successor statute is enacted), any dispute between the parties that would otherwise be determined by the reference procedure herein described will be resolved and determined by arbitration. The arbitration will be conducted by a retired California state court judge, in accordance with the California Arbitration Act, California Code of Civil Procedure §§ 1280 through 1294.2 as amended from time to time.

[Remainder of page intentionally left blank]

BORROWER:	
	MARATHON DIGITAL HOLDINGS, INC., A NEVADA CORPORATION
<u>LENDER</u> :	By:
	SILVERGATE BANK
	By: Name: Title:

Each of the parties has signed this Agreement as of the day and year first above written.

Signature Page to Revolving Credit and Security Agreement Ref. No. 7465

# Exhibit A

# Form of Request for Advance

	1,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	[ ], 20[ ]
To: SILVERGATE BANK		[ ], 20[_]
To Whom It May Concern:		
We refer to the Revolving Credit and Security Agreement dated as of July 25, 2022 (the "Agreement") between Silvergate Bank and Marathon Digital Holdings, Inc., a Nevada corporation. Terms defined in the Agreement have the same meanings in this Request for Advance (the "Request").		
Pursuant to $\underline{\text{Section 2.2}}$ of the Agreement, the Borrower hereby requests an Advance from the Lender in the amount of $\underline{\hspace{1cm}}$ on $\underline{\hspace{1cm}}$ , $20\underline{\hspace{1cm}}$ .		
Borrower hereby requests funds from the Advance should be credited to Borrower as follows:		
	Deposited into Borrower's account #	with Lender; or
	Wired to Borrower's account as follows:	
	Bank Name:  Bank ABA:  Credit Account Name:  Credit Account Number:  Reference:	
The Borrower hereby represents and certifies to the Lender that as of the date of this Request, no Event of Default exists and each of the conditions to the requested Advance set forth in the Agreement, including Section 8.2, has been satisfied (or, with respect to Sections 8.2(b) and 8.2(c) will be satisfied) or otherwise waived by the Lender.		
		Very truly yours,
		Marathon Digital Holdings, Inc., a Nevada corporation
		Ву:
		Name: Title:

Revolving Credit and Security Agreement Ref. No. 7465

\$100,000,000.00 July 28, 2022

FOR VALUE RECEIVED, Marathon Digital Holdings, Inc., a Nevada corporation (the "Borrower"), promises to pay to the order of SILVERGATE BANK, California-chartered commercial bank (the "Lender"), on the Maturity Date as provided in that certain Revolving Credit and Security Agreement dated as of the date hereof (as the same may be amended, supplemented or restated from time to time, the "Loan Agreement"), by and among Borrower and Lender, in lawful money of the United States of America and in immediately available funds, the principal sum of ONE HUNDRED MILLION AND NO/100THS DOLLARS (\$100,000,000.00) or, if less, the aggregate unpaid principal amount of all advances made by the Lender to the Borrower under the Loan Agreement (collectively, the "Advances"), together with interest from the date hereof until this Note is fully paid on the principal amount hereunder remaining unpaid from time to time, computed in the manner, and at the rates specified in the Loan Agreement. The principal hereof and interest accruing thereon shall be due and payable as provided in the Loan Agreement.

This Note evidences the Advances. This Note is a Loan Document under the Loan Agreement and is entitled to the benefits and security, and is subject to the terms and conditions, of the Loan Agreement, including, without limitation, acceleration upon the terms provided therein and in the other Loan Documents. All capitalized terms used herein which are defined in the Loan Agreement and not otherwise defined herein shall have the meanings given in the Loan Agreement.

This Note is subject to voluntary and mandatory prepayment, in full or in part, in accordance with, and subject to the terms of, the Loan Agreement. All payments of principal and interest under this Note shall be made in lawful money of the United States of America in immediately available funds at the office of the Lender or at such other place as may be designated by the Lender to the Borrower in writing. Borrower waives any rights pursuant to California Civil Code Sections 1479 and 2822 (and any amendments or successors thereto), to designate how payments will be applied, and acknowledges and agrees that Lender shall have the right in its sole discretion to determine the order and method of the application of payments on this Note or any other Loan Document

Upon the occurrence of an Event of Default, the outstanding principal balance hereunder, together with any accrued but unpaid interest and together with all of the other Obligations, may be accelerated and become immediately due and payable at the option of the Lender and without demand or notice of every kind (which are hereby expressly waived by the Borrower).

The Borrower agrees to pay all costs of collection, including attorneys' fees, all as provided in the Loan Agreement, if this Note is not paid when due, whether or not legal proceedings are commenced.

Presentment or other demand for payment, notice of dishonor and protest are expressly waived.

THIS NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (BUT OTHERWISE WITHOUT REGARD TO CONFLICT OF LAWS PROVISIONS) OF THE STATE OF NEW YORK.

The Borrower irrevocably and unconditionally agrees that it will not commence any action, litigation, or proceeding of any kind or description, whether in law or equity, whether in contract or in tort or otherwise, against the Lender in any way relating to this Note or any other Loan Document or the transactions relating hereto or thereto, in any forum other than the U.S. Federal or California state courts sitting in San Diego County, California or in New York County, New York, and the Borrower irrevocably and unconditionally submits to the jurisdiction of such courts and agrees that all claims in respect of any such action, litigation or proceeding may be heard and determined in such California State or New York State, or, to the extent permitted by law, in such Federal court, pursuant to the Loan Agreement. The Borrower agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Nothing in this Note or any other Loan

NYDIG Ref. No. 7465 Document shall affect any right that the Lender may otherwise have to bring any action or proceeding against the Borrower or its properties in the courts of any other jurisdiction.

The Borrower irrevocably and unconditionally waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any suit, action or proceeding arising out of or relating to this Note or any other Loan Document in any court referred to in the preceding paragraph. The Borrower irrevocably waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court.

The Borrower irrevocably consents to service of process in the manner provided for notices in the Loan Agreement. Nothing in this Note or any other Loan Document will affect the right of the Borrower or the Lender to serve process in any other manner permitted by law.

TO THE FULLEST EXTENT PERMITTED BY LAW, BORROWER HEREBY KNOWINGLY, INTENTIONALLY, AND VOLUNTARILY, WITH AND UPON THE ADVICE OF COMPETENT COUNSEL, AGREES NOT TO ELECT A TRIAL BY JURY OF ANY ISSUE TRIABLE OF RIGHT BY JURY, AND WAIVES ANY RIGHT TO TRIAL BY JURY FULLY TO THE EXTENT THAT ANY SUCH RIGHT SHALL NOW OR HEREAFTER EXIST WITH REGARD TO THIS NOTE, THE SECURITY INSTRUMENT, OR THE OTHER LOAN DOCUMENTS, OR ANY CLAIM, COUNTERCLAIM, OR OTHER ACTION ARISING IN CONNECTION THEREWITH. THIS WAIVER OF RIGHT TO TRIAL BY JURY IS INTENDED TO ENCOMPASS INDIVIDUALLY EACH INSTANCE AND EACH ISSUE AS TO WHICH THE RIGHT TO A TRIAL BY JURY WOULD OTHERWISE ACCRUE. LENDER IS HEREBY AUTHORIZED TO FILE A COPY OF THIS SECTION IN ANY PROCEEDING AS CONCLUSIVE EVIDENCE OF THIS WAIVER BY BORROWER.

Notwithstanding the foregoing to the contrary, in the event that the jury trial waiver contained herein shall be held or deemed to be unenforceable, Borrower hereby agrees that any controversy, dispute, or claim between the parties arising out of or relating to this Note shall be resolved by a reference proceeding in California in accordance with the provisions of California Code of Civil Procedure § 638. The referee shall be a retired California state court judge selected by mutual written agreement of the parties. If the parties are unable to agree upon a referee within ten (10) calendar days after one party serves a written notice of its intent to commence a judicial reference proceeding on the other party, then the referee will be selected by the court in accordance with California Code of Civil Procedure § 640(b). The referee shall be appointed to sit as a temporary judge, with all of the powers of a temporary judge, as authorized by law, and upon selection should take and subscribe to the oath of office as provided for in Rule 244 of the California Rules of Court (or any subsequently enacted Rule). The referee shall determine the manner in which the reference proceeding is conducted including the time and place of all hearings, the order of presentation of evidence, and all other questions that arise with respect to the course of the reference proceeding. The referee shall render a written statement of decision and shall conduct the proceedings in accordance with the California Code of Civil Procedure, the California Rules of Court, and the California Evidence Code, except as otherwise specifically agreed by the parties and approved by the referee. The referee's statement of decision shall set forth findings of fact and conclusions of law. The referee's decision shall be entered as a judgment in the court in accordance with the provisions of California Code of Civil Procedure §§ 644 and 645, and shall be appealable in accordance with California law.

Nothing in this Note shall be deemed to apply to or limit Lender's right to: (i) exercise self-help remedies such as (but not limited to) setoff; (ii) foreclose judicially or nonjudicially against any real or personal property collateral, or to exercise judicial or nonjudicial power of sale rights; (iii) obtain from a court provisional or ancillary remedies (including, without limitation, injunctive relief, a writ of possession, prejudgment attachment, a protective order, or the appointment of a receiver); or (iv) pursue its rights against any Person in a third-party proceeding in any action brought against Lender (including, without limitation, actions in bankruptcy court). Neither the exercise of any self-help remedies nor the institution or maintenance of an action for foreclosure or provisional or ancillary remedies, or the opposition to any such provisional remedies, shall constitute a waiver of the right of any party, including, without limitation, the claimant in any such action, to require submission to judicial reference the merits of the dispute

NYDIG Ref. No. 7465 occasioning resort to such remedies. No provision in the Loan Documents regarding submission to jurisdiction or venue in any court is intended to or shall be construed to be in derogation of the foregoing general judicial reference.

The foregoing judicial reference procedure constitutes a full and complete waiver of the right to a trial by jury that the parties may otherwise have and this waiver and judicial reference procedure is a material consideration to each party hereto.

If the enabling legislation which provides for appointment of a referee is repealed (and no successor statute is enacted), any dispute between the parties that would otherwise be determined by the reference procedure herein described will be resolved and determined by arbitration. The arbitration will be conducted by a retired California state court judge, in accordance with the California Arbitration Act, California Code of Civil Procedure §§ 1280 through 1294.2 as amended from time to time.

Delivery of an executed counterpart of a signature page to this Note by facsimile or in electronic (i.e., "pdf" or "tif") format shall be effective as delivery of a manually executed counterpart of this Note.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer as of the day and year and at the place set forth above.

MARATHON DIGITAL HOLDINGS, INC., A NEVADA CORPORATION

By:\_\_\_\_\_\_\_Name: Fred Thiel

Its: Chief Executive Officer

(Signature page to Revolving Note)

NYDIG Ref. No. 7465

# TERM CREDIT AND SECURITY AGREEMENT

AMONG

SILVERGATE BANK

(AS LENDER)

AND

MARATHON DIGITAL HOLDINGS, INC.

(AS BORROWER)

July 28, 2022

## TERM CREDIT AND SECURITY AGREEMENT

This Term Credit and Security Agreement is dated as of July 28, 2022 among Marathon Digital Holdings, Inc., a Nevada corporation (the "<u>Borrower</u>") and SILVERGATE BANK, a California-chartered commercial bank ("<u>Lender</u>").

IN CONSIDERATION of the mutual covenants and undertakings herein contained, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Borrower and Lender hereby agree as follows:

## DEFINITIONS.

- 1.1. <u>Accounting Terms.</u> As used in this Agreement, the Loan Documents or any certificate, report or other document made or delivered pursuant to this Agreement, accounting terms not defined in Section 1.2 or elsewhere in this Agreement and accounting terms partly defined in Section 1.2 to the extent not defined shall have the respective meanings given to them under GAAP.
- 1.2. <u>General Terms</u>. For purposes of this Agreement the following terms shall have the following meanings:
- " $\underline{Account}$ " shall mean, collectively, any deposit accounts maintained by Borrower with Lender.
- "Account Control Agreement" shall mean that certain Account Control Agreement, dated on or about the date hereof, between Borrower, Lender and Custodian, as the same may be amended, modified, extended, restated, replaced, or supplemented from time to time, including, without limitation, with a successor Custodian.
- "Affiliate" of any Person shall mean (a) any Person which, directly or indirectly, is in control of, is controlled by, or is under common control with such Person, or (b) any Person who is a director, manager, member, managing member, general partner or officer (i) of such Person, (ii) of any Subsidiary of such Person or (iii) of any Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the power, direct or indirect, (x) to vote ten percent (10%) or more of the Equity Interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for any such Person, or (y) to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.
- "Agreement" shall mean this Term Credit and Security Agreement, as the same may be amended, amended and restated, replaced and restated, extended, supplemented and/or otherwise modified from time to time.
- "Anti-Terrorism Laws" shall mean any Laws relating to terrorism, trade sanctions programs and embargoes, import/export licensing, money laundering or bribery, and any regulation, order, or directive promulgated, issued or enforced pursuant to such Laws, all as amended, supplemented or replaced from time to time.

1

"Applicable Law" shall mean all Laws applicable to the Person, conduct, transaction, covenant, Loan Document or contract in question, all provisions of all applicable state, federal and foreign constitutions, statutes, rules, regulations, treaties, directives and orders of any Governmental Body, and all orders, judgments and decrees of all courts and arbitrators.

"Applicable Margin" shall mean one and three-quarter percent (1.75%).

"Approved Electronic Communication" shall mean each notice, demand, communication, information, document and other material transmitted, posted or otherwise made or communicated by e-mail, E-Fax, or any other equivalent electronic service agreed to by Lender, whether owned, operated or hosted by Lender, that any party is obligated to, or otherwise chooses to, provide to Lender pursuant to this Agreement or any Loan Document, including any financial statement, financial and other report, notice, request, certificate and other information material; provided that Approved Electronic Communications shall not include any notice, demand, communication, information, document or other material that Lender specifically instructs a Person to deliver in physical form.

"Base Rate" is the rate of interest per annum from time to time published in the money rates section of The Wall Street Journal or any successor publication thereto as the "prime rate" then in effect; provided that if such rate of interest, as set forth from time to time in the money rates section of The Wall Street Journal, becomes unavailable for any reason as determined by Lender, the "Prime Rate" shall mean the rate of interest per annum announced by Lender as its prime rate in effect at its principal office in the State of California (such Lender announced Prime Rate not being intended to be the lowest rate of interest charged by Lender in connection with extensions of credit to debtors).

"Beneficial Owner" shall mean, for Borrower, each of the following: (a) each individual, if any, who, directly or indirectly, owns 25% or more of such Borrower's Equity Interests; and (b) a single individual with significant responsibility to control, manage, or direct such Borrower.

"Bitcoin" means a digital commodity (or "digital asset") based on the decentralized, open source protocol of the peer-to-peer Bitcoin computer network. By common convention, Bitcoin with a capital "B" typically refers to the Bitcoin Network as a whole, whereas bitcoin with a lowercase "b" refers to the digital asset native to the Bitcoin Network. For purposes of Article IV of this Agreement, the term bitcoin shall be deemed to include Forked Assets, which will be considered proceeds of the original bitcoin for which they share a common public address.

"Bitcoin Network" means the peer-to-peer Bitcoin computer network;

"Blockchain" means the public transaction ledger of the Bitcoin Network;

"Borrower" shall have the meaning set forth in the preamble to this Agreement and shall extend to all permitted successors and assigns of such Persons.

"Business Day" shall mean any day other than Saturday or Sunday or a legal holiday on which commercial banks are authorized or required by Law to be closed for business in California.

"<u>Certificate of Beneficial Ownership</u>" shall mean, for Borrower, a certificate in form and substance acceptable to Lender (as amended or modified by Lender from time to time in its reasonable discretion), certifying, among other things, the Beneficial Owner of Borrower.

"Change in Law" shall mean the occurrence, after the Closing Date, of any of the following: (a) the adoption or taking effect of any Applicable Law; (b) any change in any Applicable Law or in the administration, implementation, interpretation or application thereof by any Governmental Body; or (c) the making or issuance of any request, rule, guideline or directive (whether or not having the force of Law) by any Governmental Body; provided that notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, regulations, guidelines, interpretations or directives thereunder or issued in connection therewith (whether or not having the force of Applicable Law) and (y) all requests, rules, regulations, guidelines, interpretations or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities (whether or not having the force of Law), in each case pursuant to Basel III, shall in each case be deemed to be a Change in Law regardless of the date enacted, adopted, issued, promulgated or implemented.

"Change of Control" shall mean: (a) the occurrence of any event (whether in one or more transactions) which results in a transfer of control of Borrower to a Person; (b) any merger, consolidation or sale of substantially all of the property or assets of Borrower; or (c) removal of Fred Thiel as an officer and director of the Borrower. For purposes of this definition, "control of Borrower" shall mean the power, direct or indirect, in a single transaction, not involving a public offering (x) to vote more than fifty percent (50%) of the Equity Interests having ordinary voting power for the election of directors (or the individuals performing similar functions) of Borrower or (y) to direct or cause all or substantially all of the direction of the management and policies of Borrower by contract or otherwise.

"<u>Closing Date</u>" shall mean the date on which the conditions precedent set forth in Section 8.1 shall be satisfied or waived, in accordance with the terms hereof or such other date as may be agreed to in writing by the parties hereto.

"Code" shall mean the Internal Revenue Code of 1986, as the same may be amended or supplemented from time to time, and any successor statute of similar import, and the rules and regulations thereunder, as from time to time in effect.

"Collateral" shall mean and include all of Borrower's rights, title and interest, in and to any assets, including bitcoin, deposited, or required to be deposited, in the Collateral Account, together with all security entitlements (as such term is defined in Article 8 of the Uniform Commercial Code as adopted in the State of New York), in and to any assets, including bitcoin, held in the Collateral Account, together with all substitutions, replacements and proceeds arising out of any of the foregoing.

"Collateral Account" means the account in the name of the Borrower established to hold bitcoin collateral pledged by Borrower to the Lender, as security for the Indebtedness owed by Borrower to Lender, and made subject to the Account Control Agreement.

"Collateral Value" means, as of any date of determination, the value in Dollars of bitcoin held in the Collateral Account, in each case as reasonably determined by Lender as of 1:00 p.m. Pacific Time ("PT") on the Business Day of valuation ("Valuation Time") under the Valuation Method. In the absence of fraud or manifest error, any valuations prepared by Custodian and approved by Lender shall be deemed reasonable for purposes of this paragraph. In determining the Collateral Value, Lender may exclude from the calculation of Collateral Value any bitcoin or other assets that Lender reasonably determines creates an unreasonable risk relating to OFAC, BSA/AML compliance or similar matters based on Lender's blockchain analytics analysis and other diligence. Unless Lender is required by Applicable Law or other governmental or regulatory directive to immediately exclude any assets from the determination of the Collateral Value, Lender shall provide Borrower prior written notice before any such adjustment is made effective.

"Consents" shall mean all filings and all licenses, permits, consents, approvals, authorizations, qualifications and orders of Governmental Bodies and other third parties, domestic or foreign, necessary to carry on Borrower's business or necessary (including to avoid a conflict or breach under any agreement, instrument, other document, license, permit or other authorization) for the execution, delivery or performance of this Agreement, or the Loan Documents, including any Consents required under all applicable federal, state or other Applicable Law.

"Controlled Group" shall mean, at any time, Borrower and all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control and all other entities which, together with Borrower, are treated as a single employer under Section 414 of the Code or 29 U.S.C. 1301(b)(1).

"Covered Entity" shall mean (a) Borrower, each of Borrower's Subsidiaries, all Guarantors and all pledgors of Collateral and (b) each Person that, directly or indirectly, is in control of a Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the direct or indirect (x) ownership of, or power to vote, 25% or more of the issued and outstanding Equity Interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for such Person, or (y) power to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.

"<u>Custodian</u>" shall mean NYDIG Trust Company LLC, or such other successor custodian that Lender may approve from time to time, which approval may be granted or withheld in Lender's reasonable discretion.

"<u>Default</u>" shall mean an event, circumstance or condition which, with the giving of notice or passage of time or both, would constitute an Event of Default.

"Dollar" and the sign "\$" shall mean lawful money of the United States of America.

"Effective Date" means the date indicated in a document or agreement to be the date on which such document or agreement becomes effective, or, if there is no such indication, the date of execution of such document or agreement.

"Eligible Contract Participant" has the meaning ascribed to such term under Section 1(a)(12) of the Commodity Exchange Act, as amended and which may further be amended from time to time, including as amended by the Commodity Futures Modernization Act of 2000.

"Equity Interests" means shares of capital stock, partnership interests, membership interests in a limited liability company, beneficial interests in a trust or other equity ownership interests in a Person, and any warrants, options or other rights entitling the holder thereof to purchase or acquire any such equity interest.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as the same may be amended or supplemented from time to time and the rules and regulations promulgated thereunder.

"Event of Default" shall have the meaning set forth in Article X hereof.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

"Forked Assets" shall mean digital assets that result from any Hard Fork occurring after the Effective Date.

"GAAP" shall mean generally accepted accounting principles in the United States of America in effect from time to time.

"Governmental Body" shall mean any nation or government, any state or other political subdivision thereof or any entity, authority, agency, division or department exercising the executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to a government (including any supra-national bodies such as the European Union or the European Central Bank) and any group or body charged with setting financial accounting or regulatory capital rules or standards (including, without limitation, the Financial Accounting Standards Board, the Bank for International Settlements or the Basel Committee on Banking Supervision or any successor or similar authority to any of the foregoing).

"<u>Hard Fork</u>" means a substantial software modification to the Blockchain which results in two or more competing and incompatible Blockchain implementations, one running the premodification software program and the other running the modified version (i.e., a second "Bitcoin network").

"Indebtedness" shall mean, as to any Person at any time, any and all indebtedness, obligations or liabilities for borrowed money (whether matured or unmatured, liquidated or unliquidated, direct or indirect, absolute or contingent, or joint or several) of such Person. For the avoidance of doubt, the issuance of commercial letters of credit and the financing of mining equipment shall both be considered obligations in the Ordinary Course of Business.

"Indemnified Taxes" shall mean all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Body, including any interest, additions to tax or penalties applicable thereto.

"Insolvency Event" shall mean, with respect to any Person, including without limitation any Lender, such Person or such Person's direct or indirect parent company (a) becomes the subject of a bankruptcy or insolvency proceeding (including any proceeding under Title 11 of the United States Code), or regulatory restrictions, or (b) has had a receiver, conservator, trustee, administrator, custodian, assignee for the benefit of creditors or similar Person charged with the reorganization or liquidation of its business appointed for it or has called a meeting of its creditors, or (c) admits in writing its inability, or be generally unable, to pay its debts as they become due or cease operations of its present business.

"Interest Payment Date" shall mean the first day of each month, or if such day is not a Business Day, the next succeeding Business Day, commencing on the first day of the first full calendar month after the initial Closing Date; provided that the final Interest Payment Date shall be the Maturity Date.

"Interest Period" shall mean, with respect to any Interest Payment Date, the period of time from and including the first day of the month of the immediately preceding Interest Payment Date through the last day of such month; provided that the Interest Period for the initial Interest Payment Date shall be the period from and including the initial Closing Date to, and including, the last day of the month of the Closing Date, and the final Interest Period shall end on the Maturity Date.

"Interest Rate" means a per annum rate equal to the greater of (i) the Base Rate *plus* the Applicable Margin or (ii) five and three quarter percent (5.75%).

"Law(s)" shall mean any law(s) (including common law and equitable principles), constitution, statute, treaty, regulation, rule, ordinance, opinion, issued guidance, code, release, ruling, order, executive order, injunction, writ, decree, bond, judgment, authorization or approval, lien or award of or any settlement arrangement, by agreement, consent or otherwise, with any Governmental Body, foreign or domestic.

"Lien" shall mean any mortgage, deed of trust, pledge, hypothecation, assignment, security interest, lien (whether statutory or otherwise), charge, claim or encumbrance, or preference, priority or other security agreement or preferential arrangement held or asserted in respect of any Collateral of any kind or nature whatsoever including any conditional sale or other title retention agreement, any lease having substantially the same economic effect as any of the foregoing, and the filing of, or agreement to give, any financing statement under the Uniform Commercial Code or comparable law of any jurisdiction.

"<u>Loan Documents</u>" shall mean, collectively, this Agreement, the Account Control Agreement, the Note and any and all other documents, instruments and agreements made by Borrower in favor of Lender in connection with the Term Loan, as the same may be hereafter amended, modified or supplemented from time to time.

"Loan Party" shall mean, at each relevant time of determination, (a) Borrower, and (b) any other Person that is now or hereafter becomes a party to this Agreement as a "Borrower" or as a "Guarantor"; and "Loan Parties" shall mean collectively all such Persons.

"LTV Ratio" means, as of any date of determination, the quotient expressed as a percentage of (x) the outstanding principal balance of the Term Loan *divided by* (y) the Collateral Value.

"Material Adverse Effect" shall mean (a) a material adverse effect upon, the operations, business, assets or financial condition of Borrower, including, without limitation, any material disruption of the Bitcoin Network and/or the marketplaces where bitcoin is exchanged for U.S. Dollars; (b) a material impairment of the ability of Borrower to perform its obligations under any Loan Document to which it is a party; or (c) a material adverse effect upon the legality, validity and/or enforceability of this Agreement or any of the Loan Documents; in each case, when taken as a whole as determined by Lender in its commercially reasonable discretion.

"Maturity Date" means the earlier to occur of (a) the last day of the Term (as defined in Section 13.1 below), and (b) the date on which the Term Loan becomes due in accordance with Article XI of this Agreement.

"Maximum Rate" means, on any day, the highest rate of interest (if any) permitted by applicable law on such day.

"Note" shall have the meaning given to that term in Section 2.1, as amended, restated, extended, supplemented or otherwise modified from time to time.

"Obligations" shall mean and include any and all loans (including without limitation, the Term Loan), advances, debts, liabilities, obligations, covenants and duties owing by Borrower or any Subsidiary of Borrower under this Agreement or any Loan Document (and any amendments, extensions, renewals or increases thereto), to Lender (or to any other direct or indirect subsidiary or affiliate of Lender) of any kind or nature, present or future (including any interest or other amounts accruing thereon, any fees accruing under or in connection therewith, any costs and expenses of any Person payable by Borrower and any indemnification obligations payable by Borrower arising or payable after maturity, or after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding relating to Borrower, whether or not a claim for post-filing or post-petition interest, fees or other amounts is allowable or allowed in such proceeding), whether direct or indirect (including those acquired by assignment or participation), absolute or contingent, joint or several, due or to become due, now existing or hereafter arising, contractual or tortious, liquidated or unliquidated, regardless of how such indebtedness or liabilities arise including all costs and expenses of Lender incurred in the documentation, negotiation, modification, enforcement, collection or otherwise in connection with any of the foregoing, including but not limited to reasonable attorneys' fees and expenses and all material obligations of Borrower to Lender to perform acts or refrain from taking any action.

"Ordinary Course of Business" shall mean, with respect to Borrower, the ordinary course of such Borrower's business as conducted on the Closing Date and reasonable extensions thereof.

"Organizational Documents" shall mean, with respect to any Person, any charter, articles or certificate of incorporation, certificate of organization, registration or formation, certificate of partnership or limited partnership, bylaws, operating agreement, limited liability company agreement, or partnership agreement of such Person and any and all other applicable documents relating to such Person's formation, organization or entity governance matters (including any

shareholders' or equity holders' agreement or voting trust agreement) and specifically includes, without limitation, any certificates of designation for preferred stock or other forms of preferred equity.

"Other Taxes" shall mean all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies arising from any payment made hereunder or under any Loan Document or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any Loan Document.

"Participant" shall mean each Person who shall be granted the right by any Lender to participate the Term Loan and who shall have entered into a participation agreement in form and substance satisfactory to such Lender.

"<u>Payment Office</u>" shall mean such office of Lender or account instructions, which Lender designates to receive payments from Borrower of amounts due under this Agreement.

"PBGC" shall mean the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA or any successor.

"Person" shall mean any individual, sole proprietorship, partnership, corporation, business trust, joint stock company, trust, unincorporated organization, association, limited liability company, limited liability partnership, institution, public benefit corporation, joint venture, entity or Governmental Body (whether federal, state, county, city, municipal or otherwise, including any instrumentality, division, agency, body or department thereof).

"<u>Plan</u>" shall mean any employee benefit plan within the meaning of Section 3(3) of ERISA maintained by Borrower or any member of the Controlled Group or to which Borrower or any member of the Controlled Group is required to contribute.

"Reportable Compliance Event" shall mean that any Covered Entity becomes a Sanctioned Person, or is charged by indictment, criminal complaint or similar charging instrument, arraigned, or custodially detained in connection with any Anti-Terrorism Law or any predicate crime to any Anti-Terrorism Law, or has knowledge of facts or circumstances to the effect that it is reasonably likely that any aspect of its operations is in actual or probable violation of any Anti-Terrorism Law.

"Sanctioned Country" shall mean a country subject to a sanctions program maintained under any Anti-Terrorism Law.

"Sanctioned Person" shall mean any individual person, group, regime, entity or thing listed or otherwise recognized as a specially designated, prohibited, sanctioned or debarred person, group, regime, entity or thing, or subject to any limitations or prohibitions (including but not limited to the blocking of property or rejection of transactions), under any Anti-Terrorism Law.

"<u>Servicing and Custody Agreements</u>" shall mean one or more agreements entered into by Lender and the Custodian relating to the Collateral.

"Securities Act" shall mean the Securities Act of 1933, as amended.

"Subsidiary" of any Person shall mean a corporation or other entity of whose Equity Interests having ordinary voting power (other than Equity Interests having such power only by reason of the happening of a contingency) to elect a majority of the directors of such corporation, or other Persons performing similar functions for such entity, are owned, directly or indirectly, by such Person.

"Term Commitment" shall mean the obligation of Lender to make the Term Loan in the amount up to \$100,000,000.00. With \$50,000,000.00 to be made as of the Closing Date (the "Initial Draw"), and \$50,000,000.00 to be made, at Borrower's request, on or before April 25, 2023 (the "Delayed Draw"), and subject to satisfaction of the conditions set forth in Sections 2.1 and 8.2.

"Term Loan" shall mean the extension of credit made pursuant to this Agreement.

" $\underline{\text{Uniform Commercial Code}}$ " or "UCC" shall have the meaning set forth in Section 1.3 hereof.

"<u>USA PATRIOT Act</u>" shall mean the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

"Valuation Method" shall mean the price of bitcoin at the specified Valuation Time using the XBX index ("Index") published on TradeBlock website available at https://tradeblock.com/markets/index/xbx ("Valuation Source"). If at any time the Index or Valuation Source becomes unavailable Lender may designate an alternate Index or Valuation Source as determined by Lender in its reasonable discretion. Lender agrees to provide Borrower with prompt notice of any change in the Index or Valuation Source.

- 1.3. <u>Uniform Commercial Code Terms</u>. All terms used herein and defined in the Uniform Commercial Code as adopted in the State of New York from time to time (the "<u>Uniform Commercial Code</u>") shall have the meaning given therein unless otherwise defined herein. To the extent the definition of any category or type of collateral is expanded by any amendment, modification or revision to the Uniform Commercial Code, such expanded definition will apply automatically as of the date of such amendment, modification or revision.
- 1.4. <u>Certain Matters of Construction</u>. The terms "herein", "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular section, paragraph or subdivision. All references herein to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, this Agreement. Any pronoun used shall be deemed to cover all genders. Wherever appropriate in the context, terms used herein in the singular also include the plural and vice versa. All references to statutes and related regulations shall include any amendments of same and any successor statutes and regulations. Unless otherwise provided, all references to any instruments or agreements to which Lender is a party, including references to any of the Loan Documents, shall include any and all modifications, supplements or amendments thereto, any and all restatements or replacements thereof and any and all extensions or renewals thereof. Except as otherwise expressly provided for herein, all references herein to the time of day shall mean the

time in San Diego, California. Whenever the words "including" or "include" shall be used, such words shall be understood to mean "including, without limitation" or "include, without limitation" A Default or an Event of Default shall be deemed to exist at all times during the period commencing on the date that such Default or Event of Default occurs to the date on which such Default or Event of Default is waived in writing pursuant to this Agreement or, in the case of a Default, is cured within any period of cure expressly provided for in this Agreement; and an Event of Default shall "continue" or be "continuing" until such Event of Default has been waived in writing pursuant to this Agreement. Any Lien referred to in this Agreement or any of the Term Loan Documents as having been created in favor of Lender, any agreement entered into by Lender pursuant to this Agreement or any of the Loan Documents, any payment made by or to or funds received by Lender pursuant to or as contemplated by this Agreement or any of the Loan Documents, or any act taken or omitted to be taken by Lender, shall, unless otherwise expressly provided, be created, entered into, made or received, or taken or omitted, for the benefit or account of Lender. All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an exception to, or otherwise within the limitations of, another covenant shall not avoid the occurrence of a default if such action is taken or condition exists. In addition, all representations and warranties hereunder shall be given independent effect so that if a particular representation or warranty proves to be incorrect or is breached, the fact that another representation or warranty concerning the same or similar subject matter is correct or is not breached will not affect the incorrectness of a breach of a representation or warranty hereunder.

## II. TERM LOAN, PAYMENTS.

<u>Term Loan</u>. Subject to the terms and conditions hereof, Lender agrees to make the term loan (the "Term Loan") to Borrower in an amount not to exceed the amount of the Term Commitment. The Term Loan shall be evidenced by, and repayable in accordance with the terms of the Term Credit Note payable to Lender in an amount equal to the Term Commitment held by Lender (the "Note") and this Agreement. After repayment, the Term Loan may not be reborrowed. The Initial Draw in the amount of \$50,000,000.00 shall be funded by Lender to Borrower on the Closing Date. Borrower shall have the option, but not the obligation, to request a single disbursement of the Delayed Draw in the amount of \$50,000,000.00 on or before April 25, 2023 (the "Delayed Draw Deadline") by providing a written request to Lender and Custodian by no later than two (2) Business Days prior to the proposed disbursement date of the Delayed Draw in the form of Exhibit A hereto (a "Request for Delayed Draw"). Notwithstanding the foregoing, Lender and Custodian may, in their discretion, accept Borrower's Request for Delayed Draw with less than two (2) Business Days prior written notice. Lender shall be entitled to rely on any Request for the Delayed Draw signed by the individual signing this Agreement on behalf of Borrower or otherwise designated by Borrower, in writing, as authorized to make request for disbursement of the Delayed Draw. Upon the satisfaction or waiver of each of the conditions precedent set forth in Section 8.2, the Lender shall disburse such Delayed Draw to Borrower's deposit account with Lender. If Borrower requests that the Delayed Draw be deposited into another account of Borrower, Borrower shall deliver complete wire instructions for that account when it submits its Request for Delayed Draw, and Borrower shall be responsible for any wire transfer fees imposed by Lender in accordance with its fee schedule governing Borrower's bank transactions. Any wire requests and/or instructions submitted after 10:00 a.m. PT on the date of disbursement of the

Delayed Draw may not be deposited until the next Business Day. Furthermore, if Borrower does not exercise its option to request the Delayed Draw on or before the Delayed Draw Deadline, Lender shall have no further obligation to advance the Delayed Draw.

2.2. <u>Prepayments.</u> The Term Loan may be prepaid in whole or in part upon thirty (30) days prior written notice to the Lender. In the event of any prepayment of the Term Loan, whether by voluntary prepayment, acceleration or otherwise, the Borrower shall pay to Lender a prepayment charge equal to (A) in the case of a prepayment made on or prior to the first twelve (12) month anniversary of the Closing Date, one-half percent (0.50)% of the principal balance being prepaid, (B) in the case of a prepayment made after the first twelve (12) month anniversary of the Closing Date, one-quarter percent (0.25%) of the principal balance being prepaid, and (C) in the case of a prepayment made after the eighteenth (18) month anniversary of the Closing Date, one-quarter percent (0.25%) of the principal balance being prepaid, and (C) in the case of a prepayment made after the eighteenth (18) month anniversary of the Closing Date, zero percent (0%) of the principal balance being prepaid.

## 2.3. Manner and Repayment of Term Loan.

- (a) The Term Loan shall be due and payable in full on the last day of the Term subject to earlier prepayment as herein provided. Notwithstanding the foregoing, Term Loan shall be subject to earlier repayment upon (x) acceleration upon the occurrence of an Event of Default under this Agreement or (y) termination of this Agreement.
- (b) All payments of principal, interest and other amounts payable hereunder, or under any of the Loan Documents shall be made to Lender at the Payment Office not later than 3:00 p.m. PT on the due date therefor in Dollars in federal funds or other funds immediately available to Lender. Lender shall have the right to effectuate payment of any and all Obligations due and owing hereunder by charging Borrower's Account.
- (c) Except as expressly provided herein, all payments (including prepayments) to be made by Borrower on account of principal, interest, fees and other amounts payable hereunder shall be made without deduction, setoff or counterclaim and shall be made to Lender, in each case on or prior to 3:00 p.m.PT, in Dollars and in immediately available funds.
- 2.4. <u>Collateral Shortfall</u>. If, on any day, the LTV Ratio is equal to or greater than seventy-five percent (75%) (a "Collateral Shortfall"), the Borrower shall have one (1) Business Day to cause the LTV Ratio to be less than or equal to sixty-five percent (65%). If the LTV Ratio is not less than or equal to sixty-five percent (65%) by 1 p.m. PST on the first (1st) Business Day after the occurrence of a Collateral Shortfall, an Event of Default shall be deemed to exist, in which case Lender shall be entitled to exercise any and all rights and remedies under Article XI and/or applicable law.
- 2.5. <u>Use of Proceeds</u>. Borrower shall apply the proceeds of the Term Loan to (i) provide financing for capital expenditures, refinance of existing debt and general corporate expenses related to the Borrower's bitcoin mining business (collectively, the "<u>Transactions</u>"), and (ii) pay fees and expenses relating to the Transactions.

2.6 Return of Excess Collateral. If on any day, the LTV Ratio is less than sixty-five percent (65%) as a result of excess Collateral in the Collateral Account (the "Excess Collateral"). Borrower may, but not more frequently than once per Business Day and limited to no more than two (2) requests per work week (i.e. Monday through Friday), request a return of some or all of the Excess Collateral. Requests for a return of Excess Collateral must be in writing (which may be delivered by email) and be for a minimum amount of at least \$1,000,000.00 (or equivalent amount of BTC). Borrower shall deliver written requests to the Custodian. Lender shall use commercially reasonable efforts to authorize the Custodian to process such requests within two (2) Business Days after Lender's receipt of the request. In no event shall Borrower be entitled to a return of any Collateral, if, after giving effect to the return of that Collateral, the principal amount outstanding of the Term Loan (together with any accrued but unpaid interest and any other amounts owed to Lender) would result in sixty-five percent (65%) LTV Ratio or greater.

## III. INTEREST AND FEES.

- 3.1. Interest. Subject to the last sentence of this paragraph, the Obligations shall bear interest at the Interest Rate. Interest on the Term Loan shall be payable in arrears on each Interest Payment Date; provided that all accrued and unpaid interest shall be due and payable at the end of the Term Loan. Upon and after the occurrence of an Event of Default, and during the continuation thereof, at the option of Lender, or in the case of any Event of Default under Article X, immediately and automatically upon the occurrence of any such Event of Default without the requirement of any affirmative action by any party, the Obligations shall bear interest at the lesser of: (a) the Interest Rate, plus six percent (6%), or (b) the Maximum Rate (the "Default Rate").
- 3.2. <u>Origination Fee</u>. Borrower shall pay to Lender an origination fee of \$150,000.00 upon the execution of this Agreement.
- 3.3. <u>Contingent Draw Fee</u>. Borrower shall pay to Lender a contingent draw fee in the amount of \$250,000.00 (the, "Contingent Draw Fee") upon the execution of this Agreement. This Contingent Draw Fee will be refunded by Lender to Borrower if Borrower borrows the Delayed Draw (as outlined in Section 2.1 above) by no later than November 25, 2022.
- 3.4. <u>Computation of Interest and Fees</u>. Interest and fees hereunder shall be computed on the basis of a year of 360 days and for the actual number of days elapsed. If any payment to be made hereunder becomes due and payable on a day other than a Business Day, the due date thereof shall be extended to the next succeeding Business Day and interest thereon shall be payable at the applicable Interest Rate for the outstanding principal balance of the Term Loan during such extension.
- 3.5. <u>Maximum Charges</u>. In no event whatsoever shall interest and other charges charged hereunder exceed the highest rate permissible under Applicable Law. In the event interest and other charges as computed hereunder would otherwise exceed the highest rate permitted under Applicable Law: (a) the interest rates hereunder will be reduced to the maximum rate permitted under Applicable Law; (b) such excess amount shall be first applied to any unpaid principal balance owed by Borrower; and (c) if the then remaining excess amount is greater than the previously unpaid principal balance, Lender shall promptly refund such excess amount to Borrower and the provisions hereof shall be deemed amended to provide for such permissible rate.

Notwithstanding anything to the contrary contained in this Agreement or in any Loan Document, all agreements which either now are or which shall become agreements among Loan Parties and Lender are hereby limited so that in no contingency or event whatsoever shall the total liability for payments in the nature of interest, additional interest and other charges exceed the applicable limits imposed by any applicable usury laws. If any payments in the nature of interest, additional interest and other charges made under this Agreement or any Loan Document are held to be in excess of the limits imposed by any applicable usury laws, it is agreed that any such amount held to be in excess shall be considered payment of principal hereunder, and the indebtedness evidenced hereby shall be reduced by such amount so that the total liability for payments in the nature of interest, additional interest and other charges shall not exceed the applicable limits imposed by any applicable usury laws, in compliance with the desires of Loan Parties and Lender. The foregoing provisions shall never be superseded or waived and shall control every other provision of this Agreement or any Loan Document and all agreements among Borrower and Lender, or their respective successors and assigns. If the applicable state or federal law is amended in the future to allow a greater rate of interest to be charged under this Agreement than is presently allowed by applicable state or federal law, then the limitation of interest hereunder shall be increased to the maximum rate of interest allowed by applicable state or federal law as amended, which increase shall be effective hereunder on the effective date of such amendment, and all interest charges owing to Lender by reason thereof shall be payable in accordance with Section 3.1 of this Agreement.

- 3.6. <u>Increased Costs.</u> In the event that any Applicable Law or any Change in Law or compliance by Lender with any request or directive (whether or not having the force of law) from any central bank or other financial, monetary or other authority, shall:
- (a) subject Lender to any tax of any kind whatsoever with respect to this Agreement, or change the basis of taxation of payments to Lender in respect;
- (b) impose, modify or deem applicable any reserve, special deposit, assessment, special deposit, compulsory loan, insurance charge or similar requirement against assets held by, or deposits in or for the account of, advances or loans by, or other credit extended by, any office of Lender, including pursuant to Regulation D of the Board of Governors of the Federal Reserve System; or
- (c) impose on Lender any other condition, loss or expense (other than Taxes) affecting this Agreement or any Loan Document or the Term Loan made by Lender, and the result of any of the foregoing is to increase the cost to Lender of making, converting to, continuing, renewing or maintaining the Term Loan hereunder by an amount that Lender deems to be material or to reduce the amount of any payment (whether of principal, interest or otherwise) in respect of the Term Loan by an amount Lender deems to be material, then, in any case Borrower shall promptly pay Lender, upon its demand, such additional amount as will compensate Lender for such additional cost or such reduction, as the case may be, provided that the foregoing shall not apply to increased costs which are reflected in the Base Rate, as the case may be. Lender shall certify the amount of such additional cost or reduced amount to Borrower, and such certification shall be conclusive absent manifest error. For purposes of this Section 3.6, the Dodd-Frank Wall Street Reform and Consumer Protection Act and any and all rules, regulations, orders, requests, guidelines, and directives adopted, promulgated or implemented in connection therewith are

deemed to have been introduced and adopted after the date of this Agreement. Any such surcharge shall be limited to the direct increase in taxes or other fees caused thereby, and no convenience fee or other surcharges may be added by Lender.

3.7. Application of Payments. So long as no Event of Default then exists, all payments from Borrower received by Lender shall first be applied to accrued but unpaid interest; second, to the principal balance outstanding; and third, to any late charges and costs of collection. Upon the occurrence of an Event of Default, Lender may apply payments in such order and amounts as Lender elects, in its reasonable discretion.

## 3.8. Taxes

- (a) Any and all payments by or on account of any Obligations hereunder or under any Loan Document shall be made free and clear of and without reduction or withholding for any Indemnified Taxes or Other Taxes; provided that if Borrower shall be required by Applicable Law to deduct any Indemnified Taxes (including any Other Taxes) from such payments, then (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section) Lender receives an amount equal to the sum it would have received had no such deductions been made, (ii) Borrower shall make such deductions and (iii) Borrower shall timely pay the full amount deducted to the relevant Governmental Body in accordance with Applicable Law.
- (b) Without limiting the provisions of Section 3.8(a) above, Borrower shall timely pay any Other Taxes to the relevant Governmental Body in accordance with Applicable Law.
- (c) Borrower shall indemnify Lender, within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section) paid by Lender and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Body. A certificate as to the amount of such payment or liability delivered to Borrower by Lender shall be conclusive absent manifest error.
- (d) As soon as practicable after any payment of Indemnified Taxes or Other Taxes by Borrower to a Governmental Body, Borrower shall deliver to Lender the original or a certified copy of a receipt issued by such Governmental Body evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to Lender.
- (e) If Lender determines, in its reasonable discretion, that it has received a refund of any Indemnified Taxes or Other Taxes as to which it has been indemnified by Borrower or with respect to which Borrower have paid additional amounts pursuant to this Section, it shall pay to Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by Borrower under this Section with respect to the Indemnified Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses of the Lender and without interest (other than any interest paid by the relevant Governmental Body with respect

to such refund); provided that Borrower, upon the request of Lender, agrees to repay the amount paid over to Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Body) to Lender in the event Lender is required to repay such refund to such Governmental Body. This Section shall not be construed to require Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to Borrower or any other Person.

## IV. COLLATERAL: GENERAL TERMS.

- 4.1. <u>Security Interest in the Collateral</u>. To secure the prompt payment and performance to Lender of the Obligations, Borrower hereby assigns, pledges and grants to Lender, a continuing security interest in and to and Lien on all of the Collateral, whether now owned or existing or hereafter created, acquired or arising and wheresoever located.
- Perfection of Security Interest. Borrower shall take all action that may be necessary or commercially reasonably desirable, or that Lender may commercially reasonably request, so as at all times to maintain the validity, perfection, enforceability and priority of Lender's security interest in and Lien on the Collateral or to enable Lender to protect, exercise or enforce its rights hereunder and in the Collateral, including, but not limited to, (a) immediately discharging all Liens on the Collateral other than Lender's security interest, (b) entering into custodial arrangements satisfactory to Lender, and (c) executing and delivering financing statements, control agreements, instruments of pledge, notices and assignments, in each case in form and substance satisfactory to Lender, relating to the creation, validity, perfection, maintenance or continuation of Lender's security interest and Lien under the UCC or other Applicable Law. By its signature hereto, Borrower hereby authorizes Lender to file against such Borrower, one or more financing, continuation or amendment statements pursuant to the UCC in form and substance satisfactory to Lender, on the condition that the description of the collateral in any such statement only is limited to only the Collateral. Subject to any expense limitation set forth in Section 13.9 all charges, expenses and fees Lender may incur in doing any of the foregoing, and any local taxes relating thereto, shall be charged to Borrower's Account, and added to the Obligations, or, at Lender's option, shall be paid by Borrower to Lender immediately upon demand.
- 4.3. Preservation of Collateral. Following the occurrence of a Default or Event of Default, in addition to the rights and remedies set forth in Section 11.1 hereof, Lender: (a) may at any time take such steps as Lender deems necessary to protect Lender's interest in and to preserve the Collateral, including the hiring of cyber and security consultants or implementing other security protection measures as Lender may deem appropriate; (b) may purchase or lease cold storage vaults to preserve the Collateral and to otherwise do all acts necessary to protect Lender's interests in the Collateral; and (c) may acquire one or more hardware wallets where Lender may store the private keys associated with all or part of the Collateral. Borrower shall cooperate fully with all of Lender's efforts to preserve the Collateral and will take such actions to preserve the Collateral as Lender may direct. Borrower shall reimburse Lender all of Lender's expenses of preserving the Collateral, including any expenses relating to the bonding or insuring of a custodian, which reimbursement obligation shall be part of the Obligations. Nothing contained in this Section 4.3 is intended to impose any burden or obligation on Lender to preserve or hold any portion of the Collateral, and Lender may immediately, with or without notice, liquidate some or all of the Collateral following an Event of Default.

- 4.4. <u>Ownership and Location of Collateral</u>. With respect to the Collateral, at the time the Collateral becomes subject to Lender's security interest, Borrower shall be the sole owner of and fully authorized and able to sell, transfer, pledge and/or grant a first priority security interest in each and every item of its respective Collateral to Lender; and the Collateral shall be free and clear of all Liens whatsoever.
- 4.5. <u>Defense of Lender's Interests</u>. Until (a) payment and performance in full of all of the Obligations and (b) termination of this Agreement, Lender's interests in the Collateral shall continue in full force and effect. During such period Borrower shall not, without Lender's prior written consent, pledge, sell, assign, transfer, create or suffer to exist a Lien upon or encumber or allow or suffer to be encumbered in any way, any part of the Collateral. Borrower shall defend Lender's interests in the Collateral against any and all Persons whatsoever. At any time, following demand by Lender for payment of all Obligations during the continuance of an Event of Default, Lender shall have the right to take control of the Collateral in whatever manner stored, including hardware wallets, software wallets, whether encrypted or unencrypted, and all medium upon which the private keys associated with the Collateral are retrievable. In addition, with respect to all Collateral, Lender shall be entitled to all of the rights and remedies set forth herein and further provided by the Uniform Commercial Code or other Applicable Law.
- 4.6. <u>Financing Statements</u>. Except with respect to the financing statements filed by Lender, no financing statement covering any of the Collateral or any proceeds thereof is or will be on file in any public office.

## 4.7 Custodian

- (a) Lender shall have no liability to Borrower or other person for Custodian's failure to comply with any of its contractual obligations with respect to the Collateral or any other acts or omissions of Custodian, including, without limitation, any losses or damages arising out of the theft or misappropriation of the Collateral or claims that Custodian failed to properly safeguard the Collateral. Any diligence performed by Lender with respect to the qualifications and suitability of Custodian is solely for Lender's benefit. Lender makes no representations or statements about the Custodian's experience or qualifications to perform its obligations under the Collateral Covenants. Borrower is engaged in commercial activities involving virtual currency and digital assets, has significant experience buying, selling and storing virtual currency and other digital assets, including bitcoin, and has done its own evaluation and diligence concerning Custodian. By entering into this Agreement and accepting the Term Loan, Borrower represents and warrants to Lender that it has performed independent diligence on the Custodian and determined Custodian to be qualified to custody the Collateral. Borrower is under no obligation to enter into the transactions evidenced by this Agreement and does so only after making the determination in the immediately preceding sentence.
- (b) In the event Borrower suffers any loss or damages as a result of Custodian's acts or omissions, including without limitation, any theft or misappropriation of the Collateral, Borrower shall look solely to Custodian for any such loss or damages. Borrower hereby waives and releases Lender from any and all claims, liabilities, damages, losses or otherwise caused by Custodian or Custodian's employees, agents or contractors, including, without limitation, any claims or causes of action based on the theory that Borrower sustained damages or losses as a result of Lender

conditioning the making of the Term Loan on the requirement that a Lender-approved custodian control the Collateral during the term of the Term Loan.

- (c) Without limiting the generality of the provisions in subparagraph (a) above, Borrower also acknowledges that it shall have no right to offset any amounts owed under this Agreement or any Loan Documents by any claims against Custodian, including, without limitation, claims arising out of the theft or misappropriation of any of the Collateral, it being understood that Borrower's obligations hereunder are absolute and irrespective of any such losses or claims.
- (d) Borrower acknowledges that Lender may have entered into separate agreements with Custodian and/or one or more Affiliates of Custodian pursuant to which Lender compensates Custodian or Affiliates of Custodian, including, without limitation, for performing certain loan servicing functions with respect to the Collateral and/or the Term Loan. The obligations of the Custodian or the Affiliates of Custodian under those agreements to service certain aspects of the Term Loan and provide trade execution services are for the sole benefit of Lender and Borrower is not an intended third-party beneficiary under any such agreements.
- (e) Lender shall not be liable to Borrower for any delays in the return of any Collateral to Borrower resulting from any delays caused by the Custodian or any other cause outside the reasonable control of Lender, including, without limitation, any disruption of the bitcoin network or governmental directive preventing the lawful transfer of the Collateral to Borrower.
- (f) Custodian shall constitute a "securities intermediary" as that term is defined in Article 8 of the Uniform Commercial Code of the State of New York ("Article 8"), or an agent thereof. Borrower and Lender acknowledge and agree that (i) the Collateral Account is considered a "securities account" under Article 8, and (iii) the Collateral in the Collateral Account shall be considered "financial assets" under Article 8. The treatment of Collateral as a financial asset under Article 8 does not determine the characterization or treatment of those assets under any other law or rule
- Lender may from time to time require Borrower to establish the Collateral Account with a substitute Custodian (each, a "Replacement Custodian"). Upon Borrower's receipt of Lender's written request to establish a Collateral Account with a substitute Custodian, Borrower shall, within thirty (30) Business Days after receiving Lender's request, (i) cause such replacement Collateral Account to be established, (ii) transfer the Collateral to the replacement Collateral Account, and (iii) enter into, and cause the substitute Custodian to enter into, an account control agreement in form and substance acceptable to Lender. Borrower's failure to perform the foregoing obligations within that thirty (30) Business Days period shall entitle Lender to immediately (x) cause the liquidation of some or all of the Collateral and the application of the proceeds to repay all outstanding Obligations, and/or (y) direct the Custodian to transfer some or all of the Collateral to another account under the exclusive control of the Lender, which other account may be maintained with another custodian or subject to self-custody (the actions in clauses (x) and (y), collectively, "Protective Measures". Notwithstanding anything to the contrary contained herein, in the event Lender determines a material adverse change has occurred with respect to the then current Custodian, Lender may immediately take such Protective Measures as Lender deems necessary or prudent, in its reasonable discretion, without any prior notice to, or consent from, Borrower. In the event Lender exercises any right to liquidate Collateral under this

Section 4.7(g), Borrower shall remain liable for any remaining outstanding principal balance of the Term Loan after application of the proceeds from the liquidated Collateral. Subject to Borrower's timely compliance with the foregoing provisions of this Section 4.7(g), in the event Lender previously caused the Collateral to be transferred to another custodian or to self-custody the Collateral, Borrower may thereafter request that Lender custody the Collateral with a substitute Custodian mutually acceptable to both Lender and Borrower. If, after thirty (30) days from the date Borrower provides such written request to Lender, Borrower and Lender are unable to agree upon a substitute Custodian, Borrower may, without any prepayment fees, repay the outstanding principal balance of the Term Loan, together with all accrued and unpaid interest and all other amounts due under the Loan Documents. Borrower acknowledges that the provisions of this Section 4.7 are not intended to condition the effectiveness of Lender's designation of a substitute Custodian on Borrower's consent or approval.

## V. REPRESENTATIONS AND WARRANTIES.

Borrower represents and warrants as follows:

- Authority. Borrower has full power, authority and legal right to enter into this Agreement and the Loan Documents to which it is a party and to perform all its respective Obligations hereunder and thereunder. This Agreement and the Loan Documents to which it is a party have been duly executed and delivered by Borrower, and this Agreement and the Loan Documents to which it is a party constitute the legal, valid and binding obligation of such Borrower enforceable in accordance with their terms, except as such enforceability may be limited by any applicable bankruptcy, insolvency, moratorium or similar laws affecting creditors' rights generally. The execution, delivery and performance of this Agreement and of the Loan Documents to which it is a party (a) are within such Borrower's corporate or company powers, as applicable, have been duly authorized by all necessary corporate or company action, as applicable, are not in contravention of law or the terms of such Borrower's Organizational Documents or to the conduct of such Borrower's business or of any material contract or undertaking to which such Borrower is a party or by which such Borrower is bound, (b) will not conflict with or violate any law or regulation, or any judgment, order or decree of any Governmental Body, (c) will not require the Consent of any Governmental Body, any party to a material contract or any other Person, and (d) will not conflict with, nor result in any breach in any of the provisions of or constitute a default under or result in the creation of any Lien upon any asset of such Borrower (other than the liens created hereunder in favor of Lender) under the provisions of any agreement, instrument, or other document to which such Borrower is a party or by which it or its property is a party or by which it may be bound.
- 5.2. <u>Formation and Qualification</u>. Borrower is duly incorporated or formed, as applicable, and in good standing under the laws of the state in which it is organized and is qualified to do business and is in good standing in the states where it conducts business which constitute all states in which qualification and good standing are necessary for Borrower to conduct its business and own its property and where the failure to so qualify could reasonably be expected to have a Material Adverse Effect on such Borrower. Borrower has delivered to Lender true and complete copies of its Organizational Documents and will promptly notify Lender of any amendment or changes thereto.

- 5.3. <u>Survival of Representations and Warranties</u>. All representations and warranties of such Borrower contained in this Agreement and the Loan Documents to which it is a party shall be true at the time of such Borrower's execution of this Agreement and the Loan Documents to which it is a party, and shall survive the execution, delivery and acceptance thereof by the parties thereto and the closing of the transactions described therein or related thereto.
- 5.4. <u>Tax Returns</u>. Borrower has filed all federal, state and local tax returns, to the extent required, and other reports each is required by law to file and has paid all taxes, assessments, fees and other governmental charges that are due and payable. The provision for taxes on the books of Borrower is adequate for all years not closed by applicable statutes, and for its current fiscal year, and no Borrower has any knowledge of any deficiency or additional assessment in connection therewith not provided for on its books.
- 5.5. <u>Financial Statements</u>. The December 31, 2020, December 31, 2021 and year-to-date March 31, 2022 balance sheets of Borrower (the "<u>Balance Sheet</u>") furnished to Lender are accurate, complete and correct in all material respects and fairly reflects the financial condition of Borrower as of the Closing Date, and has been prepared in accordance with GAAP, consistently applied. The Balance Sheet has been certified as accurate, complete and correct in all material respects by the Chief Financial Officer of Borrower. All financial statements referred to in this section 5.5, including the related schedules and notes thereto, have been prepared in accordance with GAAP, except as may be disclosed in such financial statements.
- 5.6. Entity Names. Except as disclosed to Lender, in writing, during the application process, Borrower has not been known by any other company or corporate name, as applicable, in the past five (5) years, nor has Borrower been the surviving corporation or company, as applicable, of a merger or consolidation or acquired all or substantially all of the assets of any Person during the preceding five (5) years.

## 5.7. Solvency; No Litigation, Violation, Indebtedness or Default; ERISA Compliance.

- (a) (i) Borrower is, and after giving effect to the Term Loan, Borrower will be solvent, able to pay its debts as they mature, and has, and after giving effect to the Term Loan, will have capital sufficient to carry on its business and all businesses in which it is about to engage, (ii) as of the Closing Date, the fair present saleable value of its assets, calculated on a going concern basis, is in excess of the amount of its liabilities, and (iii) subsequent to the Closing Date, the fair saleable value of its assets (calculated on a going concern basis) will be in excess of the amount of its liabilities.
- (b) Except as disclosed by Borrower to Lender in writing prior to the Closing Date, no Borrower has any pending or threatened litigation, arbitration, actions or proceedings. No Borrower has any outstanding Indebtedness other than the Obligations, except for (i) Indebtedness reflected on the Borrower's financial statements delivered to Lender, and (ii) Indebtedness incurred in the ordinary course of Borrower's business that is not secured by any lien, security interest other charge in or against the Collateral.
- (c) Borrower is not in violation of any applicable statute, law, rule, regulation or ordinance in any respect which could reasonably be expected to have a Material Adverse Effect,

nor is Borrower in violation of any order of any court, Governmental Body or arbitration board or tribunal.

- (d) The Borrower and each member of the Controlled Group have fulfilled their obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and have not incurred any liability to the PBGC or a Plan under Title IV of ERISA in connection with the termination of a Plan, and each Plan maintained by the Borrower is in compliance with the presently applicable provisions of ERISA and the Code.
- 5.8. <u>Licenses and Permits</u>. Borrower (a) is in compliance with and (b) has procured and is now in possession of, all material licenses or permits required by any applicable federal, state or local law, rule or regulation for the operation of its business in each jurisdiction wherein it is now conducting or proposes to conduct business and where the failure to procure such licenses or permits could reasonably be expected to have a Material Adverse Effect.
- 5.9. <u>Default of Indebtedness</u>. No Borrower is in default in the payment of the principal of or interest on any Indebtedness or under any instrument or agreement under or subject to which any Indebtedness has been issued and no event has occurred under the provisions of any such instrument or agreement which with or without the lapse of time or the giving of notice, or both, constitutes or would constitute an event of default thereunder.
- 5.10. No Default. No Borrower is in default in the payment or performance of any of its contractual obligations and no Default or Event of Default has occurred.
- 5.11. <u>Disclosure</u>. No representation or warranty made by Borrower in this Agreement or in any financial statement, report, certificate or any other document furnished in connection herewith contains any untrue statement of a material fact or omits to state any material fact necessary to make the statements herein or therein not misleading. There is no fact known to Borrower or which reasonably should be known to such Borrower which such Borrower has not disclosed to Lender in writing with respect to the transactions contemplated by or this Agreement which could reasonably be expected to have a Material Adverse Effect.
- 5.12. <u>Business and Property of Borrower</u>. Upon and after the Closing Date, Borrower does not propose to engage in any business other than the business it currently operates as of the Effective Date, and activities incidental thereto or otherwise necessary to conduct the foregoing. On the Closing Date, Borrower will own all the property and possess all of the rights and Consents necessary for the conduct of the business of such Borrower.
- 5.13. <u>Certificate of Beneficial Ownership</u>. The Certificate of Beneficial Ownership executed and delivered to Lender for Borrower on or prior to the Closing Date, as updated from time to time in accordance with this Agreement, is accurate, complete and correct as of the date hereof and as of the date any such update is delivered. The Borrower acknowledges and agrees that the Certificate of Beneficial Ownership is one of the Loan Documents.
- 5.14 <u>Eligible Contract Participant</u>. Borrower qualifies as an Eligible Contract Participant.

## 5.15 <u>Intentionally omitted.</u>

## VI. AFFIRMATIVE COVENANTS

Borrower shall, until payment in full of the Obligations and termination of this Agreement:

- 6.1. Compliance with Laws. Comply with all Applicable Laws with respect to the Collateral or any part thereof or to the operation of such Borrower's business the non-compliance with which could reasonably be expected to have a Material Adverse Effect (except to the extent any separate provision of this Agreement shall expressly require compliance with any particular Applicable Law(s) pursuant to another standard). Borrower may, however, contest or dispute any Applicable Laws in any reasonable manner, provided that any related Lien is inchoate or stayed and sufficient reserves are established to the reasonable satisfaction of Lender to protect Lender's Lien on or security interest in the Collateral.
- 6.2. Conduct of Business and Maintenance of Existence and Assets. (a) Conduct continuously and operate actively its business according to good business practices; (b) keep in full force and effect its existence and comply in all material respects with the laws and regulations governing the conduct of its business where the failure to do so could reasonably be expected to have a Material Adverse Effect; and (c) make all such reports and pay all such franchise and other taxes and license fees and do all such other acts and things as may be lawfully required to maintain its rights, licenses, leases, powers and franchises under the laws of the United States or any political subdivision thereof where the failure to do so could reasonably be expected to have a Material Adverse Effect.
- 6.3. <u>Books and Records</u>. Keep proper books of record and account in which full, true and correct entries will be made of all dealings or transactions of or in relation to its business and affairs (including without limitation accruals for taxes), all in accordance with, or as required by, GAAP consistently applied in the opinion of such independent public accountant as shall then be regularly engaged by Borrower.
- 6.4. Payment of Taxes. Pay, when due, all taxes, assessments and other charges lawfully levied or assessed upon such Borrower or any of the Collateral, including personal property taxes, assessments and charges and all franchise, income, employment, social security benefits, withholding, and sales taxes. If any tax by any Governmental Body is or may be imposed on or as a result of any transaction between Borrower and Lender which Lender may be required to withhold or pay or if any taxes, assessments, or other charges remain unpaid after the date fixed for their payment, or if any claim shall be made which, in Lender's 's opinion, may possibly create a valid Lien on the Collateral, Lender may without notice to Borrower pay the taxes, assessments or other charges and Borrower hereby indemnifies and holds Lender harmless in respect thereof. The amount of any payment by Lender under this Section 6.4 shall be charged to Borrower' Account as an Advance and added to the Obligations and, until Borrower shall furnish Lender with an indemnity therefor (or supply Lender with evidence satisfactory to Lender that due provision for the payment thereof has been made), Lender may hold without interest any balance standing to Borrower' credit and Lender shall retain its security interest in and Lien on any and all Collateral held by Lender.

## 6.5. Financial Covenants.

- (a) Adjusted Net Worth. Borrower shall maintain at all times an Adjusted Net Worth in an amount not less than \$350,000,000.00. Lender shall determine compliance quarterly based on Borrower's consolidated financial statements filed with the United States Securities and Exchange Commission (SEC). As used herein, the term "Adjusted Net Worth" means for Borrower the excess of its total assets over its total liabilities, with consideration for bitcoin at fair market value based on the date of such financial statement.
- (b) <u>Minimum Liquidity</u>. Borrower shall maintain minimum unrestricted and unencumbered cash of at least \$25,000,000.00, which shall be verified quarterly based on consolidated financial statements filed with the SEC for the preceding quarter.

## 6.6. <u>Insurance</u>.

- (a) (i) Borrower shall maintain, at a minimum, commercial general liability insurance in an amount satisfactory to Lender, and such other insurance coverages, as is customary in the case of companies engaged in businesses similar to Borrower, and having deductibles consistent with customary practice. Borrower shall require the Custodian to maintain coverage that insures against the theft, or the accidental or intentional loss or destruction of, the private keys associated with the Collateral, subject to certain exclusions from coverage that are generally applicable to all digital assets in the custody of the Custodian (the "Required Digital Assets Coverage"). During the time in which the Collateral remains in the custody of the Custodian and the Custodian maintains the Required Digital Asset Coverage, Borrower is not required to maintain separate coverage for losses covered by the Required Digital Assets Coverage.
- (b) With respect to any Required Digital Assets Coverage maintained by a Custodian, Borrower acknowledges that such coverages may not be sufficient to cover the entire loss in the event some or all of the Collateral is lost or stolen. Lender's acceptance of such insurance should not be considered as a representation, opinion or indication that the coverages provided by the Custodian are suitable for Borrower. In some cases, the amount of proceeds available on account of a loss may only be sufficient to repay the Obligations and may not be sufficient to cover some or all of Borrower's equity in the Collateral. As such, Borrower should evaluate with its risk management personnel what insurance coverage it should maintain independent of those maintained by the Custodian.
- 6.7. <u>Payment of Indebtedness</u>. Pay, discharge or otherwise satisfy at or before maturity (subject, where applicable, to specified grace periods) all its Indebtedness, except when the failure to do so would not reasonably be expected to have a Material Adverse Effect.
- 6.8. <u>Standards of Financial Statements</u>. Cause all financial statements referred to in Section 9.2 as to which GAAP is applicable to be complete and correct in all material respects (subject, in the case of interim financial statements, to normal year-end audit adjustments) and to be prepared in reasonable detail and in accordance with GAAP applied consistently throughout the periods reflected therein (except as disclosed therein and agreed to by such reporting accountants or officer, as applicable).

- 6.9. <u>Execution of Supplemental Instruments</u>. Execute and deliver to Lender from time to time, upon demand, such supplemental agreements, statements, assignments and transfers, or instructions or documents relating to the Collateral, and such other instruments as Lender may request, in order that the full intent of this Agreement may be carried into effect.
- 6.10. Certificate of Beneficial Ownership and Other Additional Information. Provide to Lender: (i) confirmation of the accuracy of the information set forth in the most recent Certificate of Beneficial Ownership provided to the Lender; (ii) a new Certificate of Beneficial Ownership, in form and substance satisfactory to Lender when the individual(s) to be identified as a Beneficial Owner have changed; and (iii) such other information and documentation as may reasonably be requested by Lender from time to time for purposes of compliance by Lender with applicable laws (including without limitation the USA PATRIOT ACT and other "know your customer" and antimoney laundering rules and regulations), and any policy or procedure implemented by Lender or such Lender to comply therewith.
- 6.11 Operational Security. Borrower shall follow and maintain operational security best practices throughout the Term of the Term Loan, including, without limitation, ensuring that all communications initiated by Borrower to Lender and/or Custodian involving private or public keys or other sensitive information are accomplished through secure means, such as encrypted email communication. Borrower shall employ adequate security measures to safeguard any passwords, personal identification numbers and other credentials that can be used to access information about the Term Loan, the Collateral and any deposits and/or withdrawals of digital assets involving the Term Loan.

## VII. NEGATIVE COVENANTS.

Borrower shall not, until satisfaction in full of the Obligations and termination of this Agreement:

- 7.1. <u>Sale of Collateral</u>. Sell, lease, transfer or otherwise dispose of any Collateral in violation of this Agreement.
- 7.2. <u>Creation of Liens</u>. Create or suffer to exist any Lien or transfer upon or against any of the Collateral in violation of this Agreement.
- 7.3. <u>Indebtedness</u>. Create, incur, assume or suffer to exist any Indebtedness, except (i) in the Ordinary Course of Business and (ii) \$747,500,000.00 principal amount of 1.0% Convertible Senior Notes due 2026 and any restructurings thereof.
- 7.4. <u>Nature of Business</u>. Substantially change the nature of the business in which it is presently engaged, nor except as specifically permitted hereby purchase or invest, directly or indirectly, in any assets or property other than in the Ordinary Course of Business for assets or property which are useful in, necessary for and are to be used in its business as presently conducted.
- 7.5. <u>Fiscal Year and Accounting Changes</u>. Change its fiscal year end or make any change (a) in accounting treatment and reporting practices except as required by GAAP or (b) in

tax reporting treatment except as required by law or with the prior consent of Lender, which will not be unreasonably withheld or delayed.

- 7.6. <u>Pledge of Credit</u>. Now or hereafter pledge Lender's credit on any purchases, commitments or contracts or for any purpose whatsoever or use any portion of any Advance in or for any business other than such Borrower's business operations as conducted on the Closing Date.
- 7.7. <u>Compliance with ERISA</u>. Cause, or permit any member of the Controlled Group to cause, a representation or warranty in Section 5.7(d) to cease to be true and correct.

## VIII. CONDITIONS PRECEDENT.

- 8.1 The agreement of Lender to make the Initial Draw on the Closing Date is subject to the satisfaction, or waiver by Lender, immediately prior to or concurrently with the making of the Initial Draw, of the following conditions precedent:
- (a) <u>This Agreement</u>. Lender shall have received this Agreement duly executed and delivered by an authorized officer of Borrower;
- (b)  $\underline{\text{Note}}$ . Lender shall have received the Note duly executed and delivered by an authorized officer of Borrower;
- (c) <u>Loan Documents</u>. Lender shall have received each of the executed Loan Documents, as applicable, including, without limitation, and the Account Control Agreement in form and substance satisfactory to Lender;
- (d) <u>Collateral Value</u>. Lender shall have received evidence from Custodian that the aggregate amount of Collateral in the Collateral Account is sufficient in value and amount, at or prior to closing, to support the amount of the Term Loan on the Closing Date;
- (e) Net Worth and Liquidity. Lender shall have received satisfactory evidence that Borrower meets the Net Worth and Minimum Liquidity covenants;
- (f) Filings, Registrations and Recordings. Each document (including any UCC financing statement) required by this Agreement, any related agreement or under law or reasonably requested by Lender to be filed, registered or recorded in order to create, in favor of Lender, a perfected security interest in or lien upon the Collateral shall have been properly filed, registered or recorded in each jurisdiction in which the filing, registration or recordation thereof is so required or requested, and Lender shall have received an acknowledgment copy, or other evidence satisfactory to it, of each such filing, registration or recordation and satisfactory evidence of the payment of any necessary fee, tax or expense relating thereto;
- (g) <u>Secretary's Certificates, Authorizing Resolutions and Good Standings of Borrower</u>. Lender shall have received a certificate of the Secretary or Assistant Secretary (or other equivalent officer, partner or manager) of Borrower in form and substance satisfactory to Lender dated as of the Closing Date which shall certify (i) copies of resolutions in form and substance reasonably satisfactory to Lender, of the board of directors (or other equivalent governing body, member or partner) of such Borrower authorizing (x) the execution, delivery and performance of

this Agreement, the Notes and each Loan Document to which such Borrower is a party (including authorization of the incurrence of indebtedness and borrowing of the Term Loan), and (y) the granting by such Borrower of the security interests in and liens upon the Collateral to secure all of the Obligations of Borrower (and such certificate shall state that such resolutions have not been amended, modified, revoked or rescinded as of the date of such certificate), (ii) the incumbency and signature of the officers of such Borrower authorized to execute this Agreement and the Loan Documents, (iii) copies of the Organizational Documents of such Borrower as in effect on such date, complete with all amendments thereto, and (iv) the good standing (or equivalent status) of such Borrower in its jurisdiction of organization and each applicable jurisdiction where the conduct of such Borrower's business activities or the ownership of its properties necessitates qualification, as evidenced by good standing certificate(s) (or the equivalent thereof issued by any applicable jurisdiction) dated not more than thirty (30) days prior to the Closing Date, issued by the Secretary of State or other appropriate official of each such jurisdiction;

- (h) <u>Legal Opinion</u>. Lender shall have received in form and substance satisfactory to Lender, a legal opinion from Borrower's legal counsel as to good standing, due authorization, enforceability as such other matters as customarily required for transaction in the nature of the Term Loan;
- (i) No Litigation. No litigation, investigation or proceeding before or by any arbitrator or Governmental Body shall be continuing or threatened against Borrower or against the officers or directors of Borrower (A) in connection with this Agreement, the Note, the Term Loan Documents or any of the transactions contemplated thereby and which, in the reasonable opinion of Lender, is deemed material or (B) which could, in the reasonable opinion of Lender, have a Material Adverse Effect; and no injunction, writ, restraining order or other order of any nature materially adverse to Borrower or the conduct of its business or inconsistent with the due consummation of the transactions contemplated by this Agreement shall have been issued by any Governmental Body:
- (j) <u>Fees.</u> Lender shall have received all fees payable to Lender on or prior to the Closing Date hereunder, including pursuant to Article III hereof;
- (k) <u>Insurance</u>. Subject to the provisions of Section 6.6(a), Lender shall have received in form and substance satisfactory to Lender, (i) evidence that adequate insurance required to be maintained under this Agreement is in full force and effect, (ii) insurance certificates issued by Borrower's insurance broker containing such information regarding Borrower's insurance policies as Lender shall request;
- (l) <u>Other.</u> All corporate and other proceedings, and all documents, instruments and other legal matters in connection with the transactions contemplated by this Agreement shall be satisfactory in form and substance to Lender and its counsel; and
- (m) Representations and Warranties. Each of the representations and warranties made by Borrower in or pursuant to this Agreement, the Loan Documents and any related agreements to which it is a party, and each of the representations and warranties contained in any certificate, document or financial or other statement furnished at any time under or in connection with this Agreement, the Loan Documents or any related agreement shall be true and correct in all

respects on and as of such date as if made on and as of such date (except to the extent any such representation or warranty expressly relates only to any earlier and/or specified date).

- 8.2. <u>Conditions to Delayed Draw.</u> In addition to the conditions precedent set forth in Section 8.1, the agreement of Lender to make the Delayed Draw is subject to the satisfaction of the following conditions precedent as of the date such Delayed Draw is made:
- (a) <u>Request for Delayed Draw.</u> A properly completed Request for Delayed Draw received on or before the Delayed Draw Deadline;
- (b) No Default. No Event of Default or Default shall have occurred and be continuing on such date, or would exist after giving effect to the disbursement requested to be made, on such date; provided, however that Lender, in its sole discretion, may make the Delayed Draw notwithstanding the existence of an Event of Default or Default and that the disbursement of the Delayed Draw made shall not be deemed a waiver of any such Event of Default or Default;
- (c) <u>Maximum Disbursements</u>. After giving effect to the requested disbursement, the aggregate amount of disbursements shall not exceed the amount of the Term Commitment; and
- (d) <u>Collateral Value</u>. Lender shall have received evidence from Custodian that the aggregate amount of Collateral in the Collateral Account is sufficient in value and amount, at or prior to the disbursement of the Delayed Draw, to support the outstanding principal balance of the Term Loan on the day of disbursement.

Borrower's request for the Delayed Draw hereunder shall constitute a representation and warranty by Borrower as of the date of such disbursement of the Delayed Draw that the conditions contained in this Section shall have been satisfied.

## IX. INFORMATION AS TO BORROWER.

Borrower shall, until satisfaction in full of the Obligations and the termination of this Agreement:

- 9.1. <u>Disclosure of Material Matters</u>. Immediately upon learning thereof, report to Lender all matters materially affecting the value, enforceability or collectability of any portion of the Collateral, or any Lien other than Lender's security interest, placed upon or asserted against Borrower or any Collateral.
- 9.2. <u>Financial Statements.</u> Borrower shall, in accordance with Section 13 or 15(d) of the Securities and Exchange Act of 1934, file with the SEC: (a) an annual financial report on Form 10-K within ninety (90) days after the end of each of its fiscal years, and (b) quarterly financial reports on Form 10-Q forty-five (45) days after the end of each of its first three fiscal quarters, each of the preceding dates a "Reporting Due Date." The Borrower's Reporting Due Date for any Form 10-K or Form 10-Q may be extended for only such time as permitted by the SEC if the Borrower shall file with the SEC an accurate and complete SEC Form NT 10-K or NT 10-Q, as

applicable, prior to each respective Reporting Due Date and provide a copy of such Form NT to Lender immediately after filing.

## 9.3. <u>Intentionally Omitted.</u>

9.4. <u>Collateral Account Reports.</u> At or prior to 1:00 p.m. PT each day during an Availability Period, the Borrower shall make available a report to the Lender that sets forth the total amount of bitcoin subject to the Collateral Account, in each case as of 1:00 p.m. PT on the immediately preceding day. Notwithstanding the foregoing, Borrower will not be required to submit such reports for so long as Lender does not provide written notice to Borrower that the applicable Custodian is no longer providing such reports directly to Lender.

## 9.5. <u>Intentionally Omitted.</u>

- 9.6. <u>Additional Information</u>. Furnish Lender with such additional information as Lender shall reasonably request in order to enable Lender to determine whether the terms, covenants, provisions and conditions of this Agreement and the Notes have been complied with by Borrower including, without the necessity of any request by Lender, copies of all security audits and reviews.
- 9.7. Notice of Suits, Adverse Events. Furnish Lender with prompt written notice of (a) any lapse or other termination of any Consent issued to Borrower by any Governmental Body or any other Person that is material to the operation of Borrower's business, (b) any refusal by any Governmental Body or any other Person to renew or extend any such Consent; and (c) copies of any periodic or special reports filed by Borrower with any Governmental Body or Person, if such reports indicate any material change in the business, operations, affairs or condition of Borrower, or if copies thereof are requested by Lender, and (d) copies of any material notices and other communications from any Governmental Body or Person which specifically relate to Borrower.
- 9.8. <u>Financial Disclosure</u>. Borrower hereby irrevocably authorizes and directs all accountants and auditors employed by such Borrower at any time during the Term to exhibit and deliver to Lender copies of any of such Borrower's financial statements, trial balances or other accounting records of any sort in the accountant's or auditor's possession, and to disclose to Lender any information such accountants may have concerning such Borrower's financial status and business operations. Borrower hereby authorizes all Governmental Bodies to furnish to Lender copies of reports or examinations relating to such Borrower, whether made by such Borrower or otherwise; however, Lender will attempt to obtain such information or materials directly from such Borrower prior to obtaining such information or materials from such accountants or Governmental Bodies.

## X. EVENTS OF DEFAULT.

The occurrence of any one or more of the following events shall constitute an "Event of Default":

10.1. Nonpayment. Failure by Borrower to pay (a) within ten (10) days of the date due, any scheduled installment of principal or interest on the Obligations, or (b) on or before the date

due, any other fee, charge, amount or liability provided for herein, including, without limitation, any amounts owed under Section 2.4, or in any Loan Document, in each case whether at maturity, by reason of acceleration pursuant to the terms of this Agreement, by notice of intention to prepay or by required prepayment.

- 10.2. <u>Breach of Representation</u>. Any representation or warranty made or deemed made by Borrower in this Agreement, any Loan Document or any related agreement or in any certificate, document or financial or other statement furnished at any time in connection herewith or therewith shall prove to have been incorrect or misleading in any material respect on the date when made or deemed to have been made; provided that, in each case, such inaccuracies, to the extent capable of being corrected, are not corrected within ten (10) days following notice from Lender.
- 10.3. <u>Financial Information</u>. Failure by Borrower to (a) file financial reports in accordance with Section 9.2, or (b) provide Lender with financial information it may request within three (3) Business Days of such requests or (c) permit the inspection of its books or records in accordance with this Agreement;
- 10.4. <u>Judicial Actions</u>. Issuance of a notice of Lien, levy, assessment, injunction or attachment (a) against any Collateral or (b) against a material portion of Borrower's other property which is not stayed or lifted within thirty (30) days;
- 10.5. <u>Noncompliance</u>. Except as otherwise provided for in Sections 10.1, 10.2, 10.3 and 10.4, (a) failure or neglect of Borrower, or any Person to perform, keep or observe any term, provision, condition, covenant herein contained, or contained in any Loan Document or any other agreement or arrangement, now or hereafter entered into between Borrower, or such Person, and Lender beyond any applicable cure and/or grace period;
- 10.6. <u>Judgments</u>. Any (a) judgment or judgments, writ(s), order(s) or decree(s) for the payment of money are rendered against Borrower for an aggregate amount in excess of \$500,000.00 and (b) (i) action shall be legally taken by any judgment creditor to levy upon assets or properties of Borrower to enforce any such judgment, (ii) such judgment shall remain undischarged for a period of thirty (30) consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, shall not be in effect, or (iii) any Liens arising by virtue of the rendition, entry or issuance of such judgment upon assets or properties of Borrower shall be senior to any Liens in favor of Lender on such assets or properties;
- 10.7. <u>Bankruptcy</u>. Borrower, any Subsidiary or Affiliate of Borrower shall (a) apply for, consent to or suffer the appointment of, or the taking of possession by, a receiver, custodian, trustee, liquidator or similar fiduciary of itself or of all or a substantial part of its property, (b) admit in writing its inability, or be generally unable, to pay its debts as they become due or cease operations of its present business, (c) make a general assignment for the benefit of creditors, (d) commence a voluntary case under any state or federal bankruptcy or receivership laws (as now or hereafter in effect), (e) be the subject of an Insolvency Event or otherwise be adjudicated a bankrupt or insolvent (including by entry of any order for relief in any involuntary bankruptcy or insolvency proceeding commenced against it), (f) file a petition seeking to take advantage of any other law providing for the relief of debtors, (g) acquiesce to, or fail to have dismissed, within

thirty (30) days, any petition filed against it in any involuntary case under such bankruptcy laws, or (h) take any action for the purpose of effecting any of the foregoing;

- 10.8. <u>Material Adverse Effect</u>. The occurrence of any event or development which could reasonably be expected to have a Material Adverse Effect;
- 10.9. <u>Lien Priority</u>. Any Lien created hereunder or provided for hereby or under any related agreement for any reason ceases to be or is not a valid and perfected Lien having a first priority interest in the Collateral;
- 10.10. Cross Default. Either (a) any specified "event of default" under any Indebtedness (other than the Obligations) of Borrower with a then-outstanding principal balance (or, in the case of any Indebtedness not so denominated, with a then-outstanding total obligation amount) of \$500,000.00 or more, or any other event or circumstance which would permit the holder of any such Indebtedness of Borrower to accelerate such Indebtedness (and/or the obligations of Borrower thereunder) prior to the scheduled maturity or termination thereof, shall occur (regardless of whether the holder of such Indebtedness shall actually accelerate, terminate or otherwise exercise any rights or remedies with respect to such Indebtedness), (b) a default of the obligations of Borrower under any other agreement to which it is a party shall occur which has or is reasonably likely to have a Material Adverse Effect, or (c) Lender declares a default or "event of default" under any other loan or extension of credit or other account agreement to Borrower or an Affiliate of Borrower;

## 10.11. Intentionally Omitted.

- 10.12. <u>Change of Control</u>. Any Change of Control shall occur without Borrower's prior written notice to Lender and Lender's prior written approval thereof;
- 10.13. <u>Invalidity</u>. Any material provision of this Agreement or any Loan Document shall, for any reason, cease to be valid and binding on Borrower, or Borrower shall so claim in writing to Lender or Borrower challenges the validity of or its liability under this Agreement or any Loan Document:
- 10.14. <u>Seizures</u>. Any portion of the Collateral shall be seized, subject to garnishment or taken by a Governmental Body; and
- 10.15. <u>Anti-Money Laundering/International Trade Law Compliance</u>. Any representation or warranty contained in Section 13.16 is or becomes false or misleading at any time.

## XI. LENDER'S RIGHTS AND REMEDIES AFTER DEFAULT.

## 11.1. Rights and Remedies.

(a) During the continuance of: (i) an Event of Default pursuant to Section 10.7 (other than Section 10.7(g)), all Obligations shall be immediately due and payable accruing interest at the Default Rate, and (ii) any of the other Events of Default and at any time thereafter, at the option of Lender, shall be immediately due and payable. Upon the occurrence of any Event of Default, Lender shall have the right to exercise any and all rights and remedies provided for herein,

under the Loan Documents, under the Uniform Commercial Code and at law or equity generally, including the right to foreclose the security interests granted herein and to realize upon any Collateral by any available judicial or non-judicial procedure and/or to take possession of and sell any or all of the Collateral with or without judicial process. With or without having the Collateral at the time or place of sale, Lender may sell the Collateral, or any part thereof, at public or private sale, at any time or place, in one or more sales, at such price or prices, and upon such terms, either for cash, credit or future delivery, as Lender may elect. The cash proceeds realized from the sale of any Collateral shall be applied to the Obligations in the order determined by Lender in its reasonable discretion. Noncash proceeds will only be applied to the Obligations as they are converted into cash. If any deficiency shall arise, Borrower shall remain liable to Lender therefor. Notwithstanding anything to the contrary contained herein, Lender may sell, or direct any Custodian to sell, any bitcoin, Forked Assets or other digital assets in the Collateral Account without (i) any notice to Borrower, (ii) any publication of notice about a sale of such assets, it being acknowledged by Borrower that highly liquid and efficient markets exist for the sale of bitcoin, or (iii) any obligation to account to Borrower for any losses caused by Lender's decision to liquidate such Collateral.

To the extent that Applicable Law imposes duties on Lender to exercise remedies in a commercially reasonable manner, Borrower acknowledges and agrees that it is not commercially unreasonable for Lender, within customary and usual practices in the commercial banking industry in the U.S.: (i) to fail to incur expenses reasonably deemed significant by Lender to safeguard the Collateral pending disposition; (ii) to fail to obtain third party consents for access to Collateral to be disposed of, or to obtain or, if not required by other law, to fail to obtain governmental or third party consents for the collection or disposition of Collateral to be collected or disposed of; (iii) to fail to remove Liens on or any adverse claims against Collateral; (iv) if required under the UCC, to advertise dispositions of Collateral through publications or media of general circulation, whether or not the Collateral is of a specialized nature; (v) to contact other Persons, whether or not in the same business as Borrower, for expressions of interest in acquiring all or any portion of such Collateral; (vi) to hire one or more market makers and/or brokers specializing in digital asset transactions to assist in the disposition of Collateral; (vii) to dispose of Collateral by utilizing internet sites that provide for the auction of assets of the types included in the Collateral or that have the reasonable capacity of doing so, or that match buyers and sellers of assets; (viii) to dispose of assets in wholesale rather than retail markets; (ix) to disclaim disposition warranties, such as title, possession or quiet enjoyment, (x) to purchase insurance or credit enhancements to insure Lender against risks of loss, collection or disposition of Collateral or to provide to Lender a guaranteed return from the collection or disposition of Collateral; or (xi) to the extent deemed appropriate by the Lender, to obtain the services of other brokers, investment bankers, consultants and other professionals to assist Lender in the collection or disposition of any of the Collateral. Borrower acknowledges that the purpose of this Section 11.1(b) is to provide non-exhaustive indications of what actions or omissions by Lender would not be commercially unreasonable in Lender's exercise of remedies against the Collateral and that other actions or omissions by Lender shall not be deemed commercially unreasonable solely on account of not being indicated in this Section 11.1(b). Upon the occurrence of any Event of Default, Lender shall have the right to exercise any and all rights and remedies provided for herein, under the Loan Documents, under the Uniform Commercial Code and at law or equity generally, including the right to foreclose the security interests granted herein and to realize upon any Collateral by any

available judicial procedure and/or to take possession of and sell any or all of the Collateral with or without judicial process. Without limitation upon the foregoing, nothing contained in this Section 11.1(b) shall be construed to grant any rights to Borrower or to impose any duties on Lender that would not have been granted or imposed by this Agreement or by Applicable Law in the absence of this Section 11.1(b).

- 11.2. <u>Lender's Discretion</u>. Lender shall have the right in its commercially reasonable discretion to determine which rights, Liens, security interests or remedies Lender may at any time pursue, relinquish, subordinate, or modify, which procedures, timing and methodologies to employ, and what any other action to take with respect to any or all of the Collateral and in what order, thereto and such determination will not in any way modify or affect any of Lender's rights hereunder as against Borrower or each other.
- 11.3. Setoff. In addition to any other rights which Lender may have under Applicable Law, upon the occurrence of an Event of Default hereunder, Lender shall have a right, immediately and without notice of any kind, to apply Borrower's property held by Lender or any of its Affiliates to reduce the Obligations and to exercise any and all rights of setoff which may be available to Lender with respect to any deposits held by Lender.
- 11.4. <u>Liquidation without Notice</u>. Borrower acknowledges that the price of bitcoin is (a) volatile and thus may decline speedily in value, and (b) bitcoin is a type of asset customarily sold on a recognized market. Accordingly, Borrower acknowledges and agrees that it would not be necessary under Section 9-611(b) of the UCC to give notice of any proposed disposition of the Collateral.
- 11.5. <u>Rights and Remedies not Exclusive</u>. The enumeration of the foregoing rights and remedies is not intended to be exhaustive and the exercise of any rights or remedy shall not preclude the exercise of any other right or remedies provided for herein or otherwise provided by law, all of which shall be cumulative and not alternative.

#### XII. WAIVERS AND JUDICIAL PROCEEDINGS.

- 12.1. <u>Waiver of Notice</u>. Borrower hereby waives notice of non-payment of any indebtedness, demand, presentment, protest and notice thereof with respect to any and all instruments, notice of acceptance hereof, notice of loans or advances made, credit extended, Collateral received or delivered, or any other action taken in reliance hereon, and all other demands and notices of any description, except such as are expressly provided for herein.
- 12.2. <u>Delay</u>. No delay or omission on Lender's 's part in exercising any right, remedy or option shall operate as a waiver of such or any other right, remedy or option or of any Default or Event of Default.

## XIII. EFFECTIVE DATE AND TERMINATION.

13.1. <u>Term.</u> This Agreement, which shall inure to the benefit of and shall be binding upon the respective successors and permitted assigns of Borrower and Lender shall become

effective on the date hereof and shall continue in full force and effect until August 5, 2024 (the "Term") unless sooner terminated as herein provided.

- 13.2. Termination. The termination of the Agreement shall not affect Lender's rights, or any of the Obligations having their inception prior to the effective date of such termination or any Obligations which pursuant to the terms hereof continue to accrue after such date, and the provisions hereof shall continue to be fully operative until all transactions entered into, rights or interests created and Obligations have been fully and indefeasibly paid, disposed of, concluded or liquidated. The security interests, Liens and rights granted to Lender hereunder and the financing statements filed hereunder shall continue in full force and effect, notwithstanding the termination of this Agreement or the fact that Borrower's account may from time to time be temporarily in a zero or credit position, until all of the Obligations of Borrower have been indefeasibly paid. Accordingly, Borrower waives any rights which it may have under the Uniform Commercial Code to demand the filing of termination statements with respect to the Collateral, and Lender shall not be required to send such termination statements to Borrower, or to file them with any filing office, unless and until this Agreement shall have been terminated in accordance with its terms and all Obligations have been indefeasibly paid in full in immediately available funds. representations, warranties, covenants, waivers and agreements contained herein shall survive termination hereof until all Obligations are indefeasibly paid and performed in full.
- 13.3. Governing Law. This Agreement and each Loan Document (unless and except to the extent expressly provided otherwise in any such Loan Document), and all matters relating hereto or thereto or arising herefrom or therefrom (whether arising under contract law, tort law or otherwise) shall be governed by and construed in accordance with the laws of the State of New York applied to contracts to be performed wholly within the State of New York. Any judicial proceeding brought by or against Borrower with respect to any of the Obligations, this Agreement, the Loan Documents or any related agreement may be brought in any court of competent jurisdiction in the State of California or New York, United States of America, and, by execution and delivery of this Agreement, Borrower accepts for itself and in connection with its properties, generally and unconditionally, the non-exclusive jurisdiction of the aforesaid courts, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement. Borrower hereby waives personal service of any and all process upon it and consents that all such service of process may be made by certified or registered mail (return receipt requested) directed to Borrower at its address set forth in this Agreement and service so made shall be deemed completed five (5) days after the same shall have been so deposited in the mails of the United States of America, or, at Lender's option, by service upon Borrower which Borrower irrevocably appoints as such Borrower's agent for the purpose of accepting service within the State of New York or California, as applicable. Nothing herein shall affect the right to serve process in any manner permitted by law or shall limit the right of Lender to bring proceedings against Borrower in the courts of any other jurisdiction. Borrower waives any objection to jurisdiction and venue of any action instituted hereunder and shall not assert any defense based on lack of jurisdiction or venue or based upon forum non conveniens. Borrower waives the right to remove any judicial proceeding brought against such Borrower in any state court to any federal court. Any judicial proceeding by Borrower against Lender involving, directly or indirectly, any matter or claim in any way arising out of, related to or connected with this Agreement or any related

agreement, shall be brought only in a federal or state court located in New York or California, at Lender's discretion

### 13.4. Entire Understanding.

THIS AGREEMENT AND THE DOCUMENTS EXECUTED CONCURRENTLY HEREWITH CONTAIN THE ENTIRE UNDERSTANDING BETWEEN BORROWER AND LENDER AND SUPERSEDES ALL PRIOR AGREEMENTS AND UNDERSTANDINGS, IF ANY, RELATING TO THE SUBJECT MATTER OF THIS ANY PROMISES, REPRESENTATIONS, WARRANTIES OR GUARANTEES NOT HEREIN CONTAINED AND HEREINAFTER MADE SHALL HAVE NO FORCE AND EFFECT UNLESS IN WRITING, SIGNED BY BORROWER'S AND LENDER'S RESPECTIVE OFFICERS. Neither this Agreement nor any portion or provisions hereof may be changed, modified, amended, waived, supplemented, discharged, cancelled or terminated orally or by any course of dealing, or in any manner other than by an agreement in writing, signed by the party to be charged. Notwithstanding the foregoing, Lender may modify this Agreement or any of the Loan Documents for the purposes of completing missing content or correcting erroneous content of an administrative nature, without the need for a written amendment, provided that the Lender shall send a copy of any such modification to Borrower (which copy may be provided by electronic mail). Borrower acknowledges that it has been advised by counsel in connection with the execution of this Agreement and Loan Documents and is not relying upon oral representations or statements inconsistent with the terms and provisions of this Agreement.

### Successors and Assigns; Participations.

- (a) This Agreement shall be binding upon and inure to the benefit of Borrower and Lender, all future holders of the Obligations and their respective successors and assigns, except that Borrower may not assign or transfer any of its rights or obligations under this Agreement.
- (b) Borrower acknowledges that in the regular course of commercial banking business Lender may at any time and from time to time sell participating interests in the Obligations to other Persons. Each Participant may exercise all rights of payment (including rights of set-off) with respect to the portion of such Obligations held by it or other Obligations payable hereunder as fully as if such Participant were the direct holder thereof provided that (i) Borrower shall not be required to pay to any Participant more than the amount which it would have been required to pay to Lender which granted an interest in the Obligations payable hereunder to such Participant had Lender retained such interest in the Obligations payable hereunder unless the sale of the participation to such Participant is made with Borrower's prior written consent, and (ii) in no event shall Borrower be required to pay any such amount arising from the same circumstances and with respect to the other Obligations payable hereunder to both Lender and such Participant. Borrower hereby grants to any Participant a continuing security interest in any deposits, moneys or other property actually or constructively held by such Participant as security for the Participant's interest in the Obligations.
- 13.6. <u>Notice</u>. Any notice or request hereunder may be given to Borrower or to Lender at their respective addresses set forth below or at such other address as may hereafter be specified in

a notice designated as a notice of change of address under this Section. Any notice, request, demand, direction or other communication (for purposes of this Section 13.6 only, a "Notice") to be given to or made upon any party hereto under any provision of this Agreement shall be given or made by telephone or in writing (which includes by means of electronic transmission (i.e., "e-mail") or facsimile transmission or by setting forth such Notice on a website to which Borrower are directed (an "Internet Posting") if Notice of such Internet Posting (including the information necessary to access such site) has previously been delivered to the applicable parties hereto by another means set forth in this Section 13.6) in accordance with this Section 13.6. Any such Notice must be delivered to the applicable parties hereto at the addresses and numbers set forth under their respective names in this Section 13.6 hereof or in accordance with any subsequent unrevoked Notice from any such party that is given in accordance with this Section 13.6. Any Notice shall be effective:

- (a) In the case of hand-delivery, when delivered;
- (b) If given by mail, four (4) days after such Notice is deposited with the United States Postal Service, with first-class postage prepaid, return receipt requested;
- (c) In the case of a telephonic Notice, when a party is contacted by telephone, if delivery of such telephonic Notice is confirmed no later than the next Business Day by hand delivery, a facsimile or electronic transmission, an Internet Posting or an overnight courier delivery of a confirmatory Notice (received at or before noon on such next Business Day):
- (d) In the case of a facsimile transmission, when sent to the applicable party's facsimile machine's telephone number, if the party sending such Notice receives confirmation of the delivery thereof from its own facsimile machine;
  - (e) In the case of electronic transmission, when actually received;
- (f) In the case of an Internet Posting, upon delivery of a Notice of such posting (including the information necessary to access such site) by another means set forth in this Section 13.6; and
- (g) If given by any other means (including by overnight courier), when actually received.
  - (A) If to Lender at:

Silvergate Bank 4250 Executive Square, Suite 300 La Jolla, Ca 92037

Attention: Loan Servicing Telephone: (858) 362-6300 Facsimile: (858) 362-6323

Email: loanservicing@silvergate.com

(B) If to Borrower:

Marathon Digital Holdings, Inc. 1180 North Town Center Drive, Suite 100 Las Vegas, NV 89144 Attention: Fred Thiel Telephone: (725) 218-3415 Email: fred@marathondh.com

- 13.7. <u>Survival</u>. The obligations of Borrower under this Agreement shall survive termination of this Agreement and the Loan Documents and payment in full of the Obligations.
- 13.8. <u>Severability</u>. If any part of this Agreement is contrary to, prohibited by, or deemed invalid under Applicable Laws, such provision shall be inapplicable and deemed omitted to the extent so contrary, prohibited or invalid, but the remainder hereof shall not be invalidated thereby and shall be given effect so far as possible.
- 13.9. Expenses. Borrower shall pay all out-of-pocket expenses incurred by Lender and its Affiliates (including the reasonable fees, charges and disbursements of counsel for Lender), in connection with the preparation, negotiation, execution, delivery and administration of this Agreement and the Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated). Notwithstanding the forgoing, Borrower shall only pay out-of-pocket expenses in excess of \$10,000.00.
- 13.10. <u>Injunctive Relief.</u> Borrower recognizes that, in the event Borrower fails to perform, observe or discharge any of its obligations or liabilities under this Agreement, or threatens to fail to perform, observe or discharge such obligations or liabilities, any remedy at law may prove to be inadequate relief to Lender; therefor, Lender, if Lender so requests, shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving that actual damages are not an adequate remedy.

### 13.11 Limitations of Liability.

- (a) Neither Lender, nor any agent or attorney for Lender, shall be liable to Borrower (or any Affiliate of any such Person) for indirect, punitive, exemplary or consequential damages arising from any breach of contract, tort or other wrong relating to the establishment, administration or collection of the Obligations or as a result of any transaction contemplated under this Agreement or any Loan Document.
- (b) Notwithstanding anything to the contrary contained in this Agreement, Lender's liability will not exceed the greater of (i) the fair market value of the amount of the Collateral at the time in which the events giving rise to the liability occurred and (ii) the fair market value of the amount of Collateral at the time that Lender notifies Borrower in writing or Borrower otherwise has knowledge of the events giving rise to the liability. The fair market value of each digital asset held as Collateral will be determined by Lender according to the valuation policy used by the Custodian, which may differ from the way the Borrower values its digital asset holdings.

- 13.12. <u>Captions</u>. The captions at various places in this Agreement are intended for convenience only and do not constitute and shall not be interpreted as part of this Agreement.
- 13.13. <u>Counterparts; Facsimile Signatures</u>. This Agreement may be executed in any number of and by different parties hereto on separate counterparts, all of which, when so executed, shall be deemed an original, but all such counterparts shall constitute one and the same agreement. Any signature delivered by a party by facsimile or other form of electronic transmission (including email transmission of a PDF image or the use of a third-party platform, including DocuSign) shall be deemed to be an original signature hereto.
- 13.14. <u>Construction</u>. The parties acknowledge that each party and its counsel have reviewed this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendments, schedules or exhibits thereto.
- 13.15. <u>Publicity</u>. Borrower hereby authorizes Lender to make appropriate announcements of the financial arrangement entered into among Borrower and Lender, including announcements which are commonly known as tombstones, in such publications and to such selected parties as Lender shall in its reasonable discretion deem appropriate.

### 13.16. Anti-Terrorism Laws.

- (a) Borrower represents and warrants that (i) no Covered Entity is a Sanctioned Person and (ii) no Covered Entity, either in its own right or through any third party, (A) has any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Terrorism Law; (B) does business in or with, or derives any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law; or (C) engages in any dealings or transactions prohibited by any Anti-Terrorism Law.
- (b) Borrower covenants and agrees that (i) no Covered Entity will become a Sanctioned Person, (ii) no Covered Entity, either in its own right or through any third party, will (A) have any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Terrorism Law; (B) do business in or with, or derive any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law; (C) engage in any dealings or transactions prohibited by any Anti-Terrorism Law or (D) use the proceeds of the Term Loan to fund any operations in, finance any investments or activities in, or, make any payments to, a Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law, (iii) the funds used to repay the Obligations will not be derived from any unlawful activity, (iv) each Covered Entity shall comply with all Anti-Terrorism Laws and (v) the Borrower shall promptly notify the Lender in writing upon the occurrence of a Reportable Compliance Event.

### 13.17 Enforcement Waivers.

TO THE FULLEST EXTENT PERMITTED BY LAW, BORROWER HEREBY KNOWINGLY, INTENTIONALLY, AND VOLUNTARILY, WITH AND UPON THE

ADVICE OF COMPETENT COUNSEL, AGREES NOT TO ELECT A TRIAL BY JURY OF ANY ISSUE TRIABLE OF RIGHT BY JURY, AND WAIVES ANY RIGHT TO TRIAL BY JURY FULLY TO THE EXTENT THAT ANY SUCH RIGHT SHALL NOW OR HEREAFTER EXIST WITH REGARD TO THIS AGREEMENT, OR THE OTHER LOAN DOCUMENTS, OR ANY CLAIM, COUNTERCLAIM, OR OTHER ACTION ARISING IN CONNECTION THEREWITH. THIS WAIVER OF RIGHT TO TRIAL BY JURY IS INTENDED TO ENCOMPASS INDIVIDUALLY EACH INSTANCE AND EACH ISSUE AS TO WHICH THE RIGHT TO A TRIAL BY JURY WOULD OTHERWISE ACCRUE. LENDER IS HEREBY AUTHORIZED TO FILE A COPY OF THIS SECTION IN ANY PROCEEDING AS CONCLUSIVE EVIDENCE OF THIS WAIVER BY BORROWER.

Notwithstanding the foregoing to the contrary, in the event that the jury trial waiver contained herein shall be held or deemed to be unenforceable, Borrower hereby agrees that any controversy, dispute, or claim between the parties arising out of or relating to this Agreement shall be resolved by a reference proceeding in California in accordance with the provisions of California Code of Civil Procedure § 638. The referee shall be a retired California state court judge selected by mutual written agreement of the parties. If the parties are unable to agree upon a referee within ten (10) calendar days after one party serves a written notice of its intent to commence a judicial reference proceeding on the other party, then the referee will be selected by the court in accordance with California Code of Civil Procedure § 640(b). The referee shall be appointed to sit as a temporary judge, with all of the powers of a temporary judge, as authorized by law, and upon selection should take and subscribe to the oath of office as provided for in Rule 244 of the California Rules of Court (or any subsequently enacted Rule). The referee shall determine the manner in which the reference proceeding is conducted including the time and place of all hearings, the order of presentation of evidence, and all other questions that arise with respect to the course of the reference proceeding. The referee shall render a written statement of decision and shall conduct the proceedings in accordance with the California Code of Civil Procedure, the California Rules of Court, and the California Evidence Code, except as otherwise specifically agreed by the parties and approved by the referee. The referee's statement of decision shall set forth findings of fact and conclusions of law. The referee's decision shall be entered as a judgment in the court in accordance with the provisions of California Code of Civil Procedure §§ 644 and 645, and shall be appealable in accordance with California law.

Nothing in this Agreement shall be deemed to apply to or limit Lender's right to: (i) exercise self-help remedies such as (but not limited to) setoff; (ii) foreclose judicially or nonjudicially against any real or personal property collateral, or to exercise judicial or nonjudicial power of sale rights; (iii) obtain from a court provisional or ancillary remedies (including, without limitation, injunctive relief, a writ of possession, prejudgment attachment, a protective order, or the appointment of a receiver); or (iv) pursue its rights against any Person in a third-party proceeding in any action brought against Lender (including, without limitation, actions in bankruptcy court). Neither the exercise of any self-help remedies nor the institution or maintenance of an action for foreclosure or provisional or ancillary remedies, or the opposition to any such provisional remedies, shall constitute a waiver of the right of any party, including, without limitation, the claimant in any such action, to require submission to judicial reference the merits of the dispute occasioning resort to such remedies. No provision in the Loan Documents regarding submission to jurisdiction or venue

in any court is intended to or shall be construed to be in derogation of the foregoing general judicial reference.

The foregoing judicial reference procedure constitutes a full and complete waiver of the right to a trial by jury that the parties may otherwise have and this waiver and judicial reference procedure is a material consideration to each party hereto.

If the enabling legislation which provides for appointment of a referee is repealed (and no successor statute is enacted), any dispute between the parties that would otherwise be determined by the reference procedure herein described will be resolved and determined by arbitration. The arbitration will be conducted by a retired California state court judge, in accordance with the California Arbitration Act, California Code of Civil Procedure §§ 1280 through 1294.2 as amended from time to time.

[Remainder of page intentionally left blank]

BORROWER:	
	MARATHON DIGITAL HOLDINGS, INC., A NEVADA CORPORATION
LENDER:	By: Name: Fred Thiel Title: Chief Executive Officer
	SILVERGATE BANK
	By: Name: Title:

Each of the parties has signed this Agreement as of the day and year first above written.

Signature Page to Term Credit and Security Agreement Ref. No. 7470

# Exhibit A

# Form of Request for Delayed Draw

To: SILVERGATE BANK		]	], 20[]	
To Whom It May Concern:				
Silvergate Ba	e Term Credit and Security Agreement dated as of July 2 ink and Marathon Digital Holdings, Inc., a Nevada we the same meanings in this Request for Delayed Draw	corporation. Terms def		
	ection 2.1 of the Agreement, the Borrower hereby requ 0,000,000.00 on, 20	ests a Draw from the L	ender in the	
Borro	wer hereby requests funds of the Delayed Draw should be	be credited to Borrower	as follows:	
	Deposited into Borrower's account #	with Lender; or		
Wired to Borrower's account as follows:				
	Bank Name:  Bank ABA;  Credit Account Name:  Credit Account Number:  Reference:			
The Borrower hereby represents and certifies to the Lender that as of the date of this Draw, no Event of Default exists and each of the conditions to the requested Draw set forth in the Agreement, including <u>Section 8.2</u> , has been satisfied (or, with respect to <u>Sections 8.2(a)</u> and <u>8.2(b)</u> will be satisfied) or otherwise waived by the Lender.				
Very truly yours,				
	Mar	rathon Digital Holdings evada corporation	, Inc.,	
	1	Name: Fitle:		

Term Credit and Security Agreement Ref. No. 7470

# TERM CREDIT AND SECURITY AGREEMENT

AMONG

SILVERGATE BANK

(AS LENDER)

AND

MARATHON DIGITAL HOLDINGS, INC.

(AS BORROWER)

July 28, 2022

### TERM CREDIT AND SECURITY AGREEMENT

This Term Credit and Security Agreement is dated as of July 28, 2022 among Marathon Digital Holdings, Inc., a Nevada corporation (the "<u>Borrower</u>") and SILVERGATE BANK, a California-chartered commercial bank ("<u>Lender</u>").

IN CONSIDERATION of the mutual covenants and undertakings herein contained, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Borrower and Lender hereby agree as follows:

### DEFINITIONS.

- 1.1. <u>Accounting Terms.</u> As used in this Agreement, the Loan Documents or any certificate, report or other document made or delivered pursuant to this Agreement, accounting terms not defined in Section 1.2 or elsewhere in this Agreement and accounting terms partly defined in Section 1.2 to the extent not defined shall have the respective meanings given to them under GAAP.
- 1.2. <u>General Terms</u>. For purposes of this Agreement the following terms shall have the following meanings:
- " $\underline{Account}$ " shall mean, collectively, any deposit accounts maintained by Borrower with Lender.
- "Account Control Agreement" shall mean that certain Account Control Agreement, dated on or about the date hereof, between Borrower, Lender and Custodian, as the same may be amended, modified, extended, restated, replaced, or supplemented from time to time, including, without limitation, with a successor Custodian.
- "Affiliate" of any Person shall mean (a) any Person which, directly or indirectly, is in control of, is controlled by, or is under common control with such Person, or (b) any Person who is a director, manager, member, managing member, general partner or officer (i) of such Person, (ii) of any Subsidiary of such Person or (iii) of any Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the power, direct or indirect, (x) to vote ten percent (10%) or more of the Equity Interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for any such Person, or (y) to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.
- "Agreement" shall mean this Term Credit and Security Agreement, as the same may be amended, amended and restated, replaced and restated, extended, supplemented and/or otherwise modified from time to time.
- "Anti-Terrorism Laws" shall mean any Laws relating to terrorism, trade sanctions programs and embargoes, import/export licensing, money laundering or bribery, and any regulation, order, or directive promulgated, issued or enforced pursuant to such Laws, all as amended, supplemented or replaced from time to time.

1

"Applicable Law" shall mean all Laws applicable to the Person, conduct, transaction, covenant, Loan Document or contract in question, all provisions of all applicable state, federal and foreign constitutions, statutes, rules, regulations, treaties, directives and orders of any Governmental Body, and all orders, judgments and decrees of all courts and arbitrators.

"Applicable Margin" shall mean one and three-quarter percent (1.75%).

"Approved Electronic Communication" shall mean each notice, demand, communication, information, document and other material transmitted, posted or otherwise made or communicated by e-mail, E-Fax, or any other equivalent electronic service agreed to by Lender, whether owned, operated or hosted by Lender, that any party is obligated to, or otherwise chooses to, provide to Lender pursuant to this Agreement or any Loan Document, including any financial statement, financial and other report, notice, request, certificate and other information material; provided that Approved Electronic Communications shall not include any notice, demand, communication, information, document or other material that Lender specifically instructs a Person to deliver in physical form.

"Base Rate" is the rate of interest per annum from time to time published in the money rates section of The Wall Street Journal or any successor publication thereto as the "prime rate" then in effect; provided that if such rate of interest, as set forth from time to time in the money rates section of The Wall Street Journal, becomes unavailable for any reason as determined by Lender, the "Prime Rate" shall mean the rate of interest per annum announced by Lender as its prime rate in effect at its principal office in the State of California (such Lender announced Prime Rate not being intended to be the lowest rate of interest charged by Lender in connection with extensions of credit to debtors).

"Beneficial Owner" shall mean, for Borrower, each of the following: (a) each individual, if any, who, directly or indirectly, owns 25% or more of such Borrower's Equity Interests; and (b) a single individual with significant responsibility to control, manage, or direct such Borrower.

"Bitcoin" means a digital commodity (or "digital asset") based on the decentralized, open source protocol of the peer-to-peer Bitcoin computer network. By common convention, Bitcoin with a capital "B" typically refers to the Bitcoin Network as a whole, whereas bitcoin with a lowercase "b" refers to the digital asset native to the Bitcoin Network. For purposes of Article IV of this Agreement, the term bitcoin shall be deemed to include Forked Assets, which will be considered proceeds of the original bitcoin for which they share a common public address.

"Bitcoin Network" means the peer-to-peer Bitcoin computer network;

"Blockchain" means the public transaction ledger of the Bitcoin Network;

"Borrower" shall have the meaning set forth in the preamble to this Agreement and shall extend to all permitted successors and assigns of such Persons.

"Business Day" shall mean any day other than Saturday or Sunday or a legal holiday on which commercial banks are authorized or required by Law to be closed for business in California.

"<u>Certificate of Beneficial Ownership</u>" shall mean, for Borrower, a certificate in form and substance acceptable to Lender (as amended or modified by Lender from time to time in its reasonable discretion), certifying, among other things, the Beneficial Owner of Borrower.

"Change in Law" shall mean the occurrence, after the Closing Date, of any of the following: (a) the adoption or taking effect of any Applicable Law; (b) any change in any Applicable Law or in the administration, implementation, interpretation or application thereof by any Governmental Body; or (c) the making or issuance of any request, rule, guideline or directive (whether or not having the force of Law) by any Governmental Body; provided that notwithstanding anything herein to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, regulations, guidelines, interpretations or directives thereunder or issued in connection therewith (whether or not having the force of Applicable Law) and (y) all requests, rules, regulations, guidelines, interpretations or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities (whether or not having the force of Law), in each case pursuant to Basel III, shall in each case be deemed to be a Change in Law regardless of the date enacted, adopted, issued, promulgated or implemented.

"Change of Control" shall mean: (a) the occurrence of any event (whether in one or more transactions) which results in a transfer of control of Borrower to a Person; (b) any merger, consolidation or sale of substantially all of the property or assets of Borrower; or (c) removal of Fred Thiel as an officer and director of the Borrower. For purposes of this definition, "control of Borrower" shall mean the power, direct or indirect, in a single transaction, not involving a public offering (x) to vote more than fifty percent (50%) of the Equity Interests having ordinary voting power for the election of directors (or the individuals performing similar functions) of Borrower or (y) to direct or cause all or substantially all of the direction of the management and policies of Borrower by contract or otherwise.

"<u>Closing Date</u>" shall mean the date on which the conditions precedent set forth in Section 8.1 shall be satisfied or waived, in accordance with the terms hereof or such other date as may be agreed to in writing by the parties hereto.

"Code" shall mean the Internal Revenue Code of 1986, as the same may be amended or supplemented from time to time, and any successor statute of similar import, and the rules and regulations thereunder, as from time to time in effect.

"Collateral" shall mean and include all of Borrower's rights, title and interest, in and to any assets, including bitcoin, deposited, or required to be deposited, in the Collateral Account, together with all security entitlements (as such term is defined in Article 8 of the Uniform Commercial Code as adopted in the State of New York), in and to any assets, including bitcoin, held in the Collateral Account, together with all substitutions, replacements and proceeds arising out of any of the foregoing.

"Collateral Account" means the account in the name of the Borrower established to hold bitcoin collateral pledged by Borrower to the Lender, as security for the Indebtedness owed by Borrower to Lender, and made subject to the Account Control Agreement.

"Collateral Value" means, as of any date of determination, the value in Dollars of bitcoin held in the Collateral Account, in each case as reasonably determined by Lender as of 1:00 p.m. Pacific Time ("PT") on the Business Day of valuation ("Valuation Time") under the Valuation Method. In the absence of fraud or manifest error, any valuations prepared by Custodian and approved by Lender shall be deemed reasonable for purposes of this paragraph. In determining the Collateral Value, Lender may exclude from the calculation of Collateral Value any bitcoin or other assets that Lender reasonably determines creates an unreasonable risk relating to OFAC, BSA/AML compliance or similar matters based on Lender's blockchain analytics analysis and other diligence. Unless Lender is required by Applicable Law or other governmental or regulatory directive to immediately exclude any assets from the determination of the Collateral Value, Lender shall provide Borrower prior written notice before any such adjustment is made effective.

"Consents" shall mean all filings and all licenses, permits, consents, approvals, authorizations, qualifications and orders of Governmental Bodies and other third parties, domestic or foreign, necessary to carry on Borrower's business or necessary (including to avoid a conflict or breach under any agreement, instrument, other document, license, permit or other authorization) for the execution, delivery or performance of this Agreement, or the Loan Documents, including any Consents required under all applicable federal, state or other Applicable Law.

"Controlled Group" shall mean, at any time, Borrower and all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control and all other entities which, together with Borrower, are treated as a single employer under Section 414 of the Code or 29 U.S.C. 1301(b)(1).

"Covered Entity" shall mean (a) Borrower, each of Borrower's Subsidiaries, all Guarantors and all pledgors of Collateral and (b) each Person that, directly or indirectly, is in control of a Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the direct or indirect (x) ownership of, or power to vote, 25% or more of the issued and outstanding Equity Interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for such Person, or (y) power to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.

"<u>Custodian</u>" shall mean NYDIG Trust Company LLC, or such other successor custodian that Lender may approve from time to time, which approval may be granted or withheld in Lender's reasonable discretion.

"<u>Default</u>" shall mean an event, circumstance or condition which, with the giving of notice or passage of time or both, would constitute an Event of Default.

"Dollar" and the sign "\$" shall mean lawful money of the United States of America.

"Effective Date" means the date indicated in a document or agreement to be the date on which such document or agreement becomes effective, or, if there is no such indication, the date of execution of such document or agreement.

"<u>Eligible Contract Participant</u>" has the meaning ascribed to such term under Section 1(a)(12) of the Commodity Exchange Act, as amended and which may further be amended from time to time, including as amended by the Commodity Futures Modernization Act of 2000.

"Equity Interests" means shares of capital stock, partnership interests, membership interests in a limited liability company, beneficial interests in a trust or other equity ownership interests in a Person, and any warrants, options or other rights entitling the holder thereof to purchase or acquire any such equity interest.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as the same may be amended or supplemented from time to time and the rules and regulations promulgated thereunder.

"Event of Default" shall have the meaning set forth in Article X hereof.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

"Forked Assets" shall mean digital assets that result from any Hard Fork occurring after the Effective Date.

"GAAP" shall mean generally accepted accounting principles in the United States of America in effect from time to time.

"Governmental Body" shall mean any nation or government, any state or other political subdivision thereof or any entity, authority, agency, division or department exercising the executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to a government (including any supra-national bodies such as the European Union or the European Central Bank) and any group or body charged with setting financial accounting or regulatory capital rules or standards (including, without limitation, the Financial Accounting Standards Board, the Bank for International Settlements or the Basel Committee on Banking Supervision or any successor or similar authority to any of the foregoing).

"<u>Hard Fork</u>" means a substantial software modification to the Blockchain which results in two or more competing and incompatible Blockchain implementations, one running the premodification software program and the other running the modified version (i.e., a second "Bitcoin network").

"Indebtedness" shall mean, as to any Person at any time, any and all indebtedness, obligations or liabilities for borrowed money (whether matured or unmatured, liquidated or unliquidated, direct or indirect, absolute or contingent, or joint or several) of such Person. For the avoidance of doubt, the issuance of commercial letters of credit and the financing of mining equipment shall both be considered obligations in the Ordinary Course of Business.

"Indemnified Taxes" shall mean all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Body, including any interest, additions to tax or penalties applicable thereto.

"Insolvency Event" shall mean, with respect to any Person, including without limitation any Lender, such Person or such Person's direct or indirect parent company (a) becomes the subject of a bankruptcy or insolvency proceeding (including any proceeding under Title 11 of the United States Code), or regulatory restrictions, or (b) has had a receiver, conservator, trustee, administrator, custodian, assignee for the benefit of creditors or similar Person charged with the reorganization or liquidation of its business appointed for it or has called a meeting of its creditors, or (c) admits in writing its inability, or be generally unable, to pay its debts as they become due or cease operations of its present business.

"Interest Payment Date" shall mean the first day of each month, or if such day is not a Business Day, the next succeeding Business Day, commencing on the first day of the first full calendar month after the initial Closing Date; provided that the final Interest Payment Date shall be the Maturity Date.

"Interest Period" shall mean, with respect to any Interest Payment Date, the period of time from and including the first day of the month of the immediately preceding Interest Payment Date through the last day of such month; provided that the Interest Period for the initial Interest Payment Date shall be the period from and including the initial Closing Date to, and including, the last day of the month of the Closing Date, and the final Interest Period shall end on the Maturity Date.

"Interest Rate" means a per annum rate equal to the greater of (i) the Base Rate *plus* the Applicable Margin or (ii) five and three quarter percent (5.75%).

"Law(s)" shall mean any law(s) (including common law and equitable principles), constitution, statute, treaty, regulation, rule, ordinance, opinion, issued guidance, code, release, ruling, order, executive order, injunction, writ, decree, bond, judgment, authorization or approval, lien or award of or any settlement arrangement, by agreement, consent or otherwise, with any Governmental Body, foreign or domestic.

"Lien" shall mean any mortgage, deed of trust, pledge, hypothecation, assignment, security interest, lien (whether statutory or otherwise), charge, claim or encumbrance, or preference, priority or other security agreement or preferential arrangement held or asserted in respect of any Collateral of any kind or nature whatsoever including any conditional sale or other title retention agreement, any lease having substantially the same economic effect as any of the foregoing, and the filing of, or agreement to give, any financing statement under the Uniform Commercial Code or comparable law of any jurisdiction.

"<u>Loan Documents</u>" shall mean, collectively, this Agreement, the Account Control Agreement, the Note and any and all other documents, instruments and agreements made by Borrower in favor of Lender in connection with the Term Loan, as the same may be hereafter amended, modified or supplemented from time to time.

"Loan Party" shall mean, at each relevant time of determination, (a) Borrower, and (b) any other Person that is now or hereafter becomes a party to this Agreement as a "Borrower" or as a "Guarantor"; and "Loan Parties" shall mean collectively all such Persons.

"LTV Ratio" means, as of any date of determination, the quotient expressed as a percentage of (x) the outstanding principal balance of the Term Loan divided by (y) the Collateral Value.

"Material Adverse Effect" shall mean (a) a material adverse effect upon, the operations, business, assets or financial condition of Borrower, including, without limitation, any material disruption of the Bitcoin Network and/or the marketplaces where bitcoin is exchanged for U.S. Dollars; (b) a material impairment of the ability of Borrower to perform its obligations under any Loan Document to which it is a party; or (c) a material adverse effect upon the legality, validity and/or enforceability of this Agreement or any of the Loan Documents; in each case, when taken as a whole as determined by Lender in its commercially reasonable discretion.

"<u>Maturity Date</u>" means the earlier to occur of (a) the last day of the Term (as defined in Section 13.1 below), and (b) the date on which the Term Loan becomes due in accordance with Article XI of this Agreement.

" $\underline{\text{Maximum Rate}}$ " means, on any day, the highest rate of interest (if any) permitted by applicable law on such day.

"Note" shall have the meaning given to that term in Section 2.1, as amended, restated, extended, supplemented or otherwise modified from time to time.

"Obligations" shall mean and include any and all loans (including without limitation, the Term Loan), advances, debts, liabilities, obligations, covenants and duties owing by Borrower or any Subsidiary of Borrower under this Agreement or any Loan Document (and any amendments, extensions, renewals or increases thereto), to Lender (or to any other direct or indirect subsidiary or affiliate of Lender) of any kind or nature, present or future (including any interest or other amounts accruing thereon, any fees accruing under or in connection therewith, any costs and expenses of any Person payable by Borrower and any indemnification obligations payable by Borrower arising or payable after maturity, or after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding relating to Borrower, whether or not a claim for post-filing or post-petition interest, fees or other amounts is allowable or allowed in such proceeding), whether direct or indirect (including those acquired by assignment or participation), absolute or contingent, joint or several, due or to become due, now existing or hereafter arising, contractual or tortious, liquidated or unliquidated, regardless of how such indebtedness or liabilities arise including all costs and expenses of Lender incurred in the documentation, negotiation, modification, enforcement, collection or otherwise in connection with any of the foregoing, including but not limited to reasonable attorneys' fees and expenses and all material obligations of Borrower to Lender to perform acts or refrain from taking any action.

"Ordinary Course of Business" shall mean, with respect to Borrower, the ordinary course of such Borrower's business as conducted on the Closing Date and reasonable extensions thereof.

"Organizational Documents" shall mean, with respect to any Person, any charter, articles or certificate of incorporation, certificate of organization, registration or formation, certificate of partnership or limited partnership, bylaws, operating agreement, limited liability company agreement, or partnership agreement of such Person and any and all other applicable documents relating to such Person's formation, organization or entity governance matters (including any

shareholders' or equity holders' agreement or voting trust agreement) and specifically includes, without limitation, any certificates of designation for preferred stock or other forms of preferred equity.

"Other Taxes" shall mean all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies arising from any payment made hereunder or under any Loan Document or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any Loan Document.

"<u>Participant</u>" shall mean each Person who shall be granted the right by any Lender to participate the Term Loan and who shall have entered into a participation agreement in form and substance satisfactory to such Lender.

"<u>Payment Office</u>" shall mean such office of Lender or account instructions, which Lender designates to receive payments from Borrower of amounts due under this Agreement.

"PBGC" shall mean the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA or any successor.

"Person" shall mean any individual, sole proprietorship, partnership, corporation, business trust, joint stock company, trust, unincorporated organization, association, limited liability company, limited liability partnership, institution, public benefit corporation, joint venture, entity or Governmental Body (whether federal, state, county, city, municipal or otherwise, including any instrumentality, division, agency, body or department thereof).

"<u>Plan</u>" shall mean any employee benefit plan within the meaning of Section 3(3) of ERISA maintained by Borrower or any member of the Controlled Group or to which Borrower or any member of the Controlled Group is required to contribute.

"Reportable Compliance Event" shall mean that any Covered Entity becomes a Sanctioned Person, or is charged by indictment, criminal complaint or similar charging instrument, arraigned, or custodially detained in connection with any Anti-Terrorism Law or any predicate crime to any Anti-Terrorism Law, or has knowledge of facts or circumstances to the effect that it is reasonably likely that any aspect of its operations is in actual or probable violation of any Anti-Terrorism Law.

"Sanctioned Country" shall mean a country subject to a sanctions program maintained under any Anti-Terrorism Law.

"Sanctioned Person" shall mean any individual person, group, regime, entity or thing listed or otherwise recognized as a specially designated, prohibited, sanctioned or debarred person, group, regime, entity or thing, or subject to any limitations or prohibitions (including but not limited to the blocking of property or rejection of transactions), under any Anti-Terrorism Law.

"Servicing and Custody Agreements" shall mean one or more agreements entered into by Lender and the Custodian relating to the Collateral.

"Securities Act" shall mean the Securities Act of 1933, as amended.

"Subsidiary" of any Person shall mean a corporation or other entity of whose Equity Interests having ordinary voting power (other than Equity Interests having such power only by reason of the happening of a contingency) to elect a majority of the directors of such corporation, or other Persons performing similar functions for such entity, are owned, directly or indirectly, by such Person.

"Term Commitment" shall mean the obligation of Lender to make the Term Loan in the amount up to \$100,000,000.00. With \$50,000,000.00 to be made as of the Closing Date (the "Initial Draw"), and \$50,000,000.00 to be made, at Borrower's request, on or before April 25, 2023 (the "Delayed Draw"), and subject to satisfaction of the conditions set forth in Sections 2.1 and 8.2.

"Term Loan" shall mean the extension of credit made pursuant to this Agreement.

" $\underline{\text{Uniform Commercial Code}}$ " or "UCC" shall have the meaning set forth in Section 1.3 hereof.

"<u>USA PATRIOT Act</u>" shall mean the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

"Valuation Method" shall mean the price of bitcoin at the specified Valuation Time using the XBX index ("Index") published on TradeBlock website available at https://tradeblock.com/markets/index/xbx ("Valuation Source"). If at any time the Index or Valuation Source becomes unavailable Lender may designate an alternate Index or Valuation Source as determined by Lender in its reasonable discretion. Lender agrees to provide Borrower with prompt notice of any change in the Index or Valuation Source.

- 1.3. <u>Uniform Commercial Code Terms</u>. All terms used herein and defined in the Uniform Commercial Code as adopted in the State of New York from time to time (the "<u>Uniform Commercial Code</u>") shall have the meaning given therein unless otherwise defined herein. To the extent the definition of any category or type of collateral is expanded by any amendment, modification or revision to the Uniform Commercial Code, such expanded definition will apply automatically as of the date of such amendment, modification or revision.
- 1.4. <u>Certain Matters of Construction</u>. The terms "herein", "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular section, paragraph or subdivision. All references herein to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, this Agreement. Any pronoun used shall be deemed to cover all genders. Wherever appropriate in the context, terms used herein in the singular also include the plural and vice versa. All references to statutes and related regulations shall include any amendments of same and any successor statutes and regulations. Unless otherwise provided, all references to any instruments or agreements to which Lender is a party, including references to any of the Loan Documents, shall include any and all modifications, supplements or amendments thereto, any and all restatements or replacements thereof and any and all extensions or renewals thereof. Except as otherwise expressly provided for herein, all references herein to the time of day shall mean the

time in San Diego, California. Whenever the words "including" or "include" shall be used, such words shall be understood to mean "including, without limitation" or "include, without limitation". A Default or an Event of Default shall be deemed to exist at all times during the period commencing on the date that such Default or Event of Default occurs to the date on which such Default or Event of Default is waived in writing pursuant to this Agreement or, in the case of a Default, is cured within any period of cure expressly provided for in this Agreement; and an Event of Default shall "continue" or be "continuing" until such Event of Default has been waived in writing pursuant to this Agreement. Any Lien referred to in this Agreement or any of the Term Loan Documents as having been created in favor of Lender, any agreement entered into by Lender pursuant to this Agreement or any of the Loan Documents, any payment made by or to or funds received by Lender pursuant to or as contemplated by this Agreement or any of the Loan Documents, or any act taken or omitted to be taken by Lender, shall, unless otherwise expressly provided, be created, entered into, made or received, or taken or omitted, for the benefit or account of Lender. All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an exception to, or otherwise within the limitations of, another covenant shall not avoid the occurrence of a default if such action is taken or condition exists. In addition, all representations and warranties hereunder shall be given independent effect so that if a particular representation or warranty proves to be incorrect or is breached, the fact that another representation or warranty concerning the same or similar subject matter is correct or is not breached will not affect the incorrectness of a breach of a representation or warranty hereunder.

### II. TERM LOAN, PAYMENTS.

Term Loan. Subject to the terms and conditions hereof, Lender agrees to make the term loan (the "Term Loan") to Borrower in an amount not to exceed the amount of the Term Commitment. The Term Loan shall be evidenced by, and repayable in accordance with the terms of the Term Credit Note payable to Lender in an amount equal to the Term Commitment held by Lender (the "Note") and this Agreement. After repayment, the Term Loan may not be reborrowed. The Initial Draw in the amount of \$50,000,000.00 shall be funded by Lender to Borrower on the Closing Date. Borrower shall have the option, but not the obligation, to request a single disbursement of the Delayed Draw in the amount of \$50,000,000.00 on or before April 25, 2023 (the "Delayed Draw Deadline") by providing a written request to Lender and Custodian by no later than two (2) Business Days prior to the proposed disbursement date of the Delayed Draw in the form of Exhibit A hereto (a "Request for Delayed Draw"). Notwithstanding the foregoing, Lender and Custodian may, in their discretion, accept Borrower's Request for Delayed Draw with less than two (2) Business Days prior written notice. Lender shall be entitled to rely on any Request for the Delayed Draw signed by the individual signing this Agreement on behalf of Borrower or otherwise designated by Borrower, in writing, as authorized to make request for disbursement of the Delayed Draw. Upon the satisfaction or waiver of each of the conditions precedent set forth in Section 8.2, the Lender shall disburse such Delayed Draw to Borrower's deposit account with Lender. If Borrower requests that the Delayed Draw be deposited into another account of Borrower, Borrower shall deliver complete wire instructions for that account when it submits its Request for Delayed Draw, and Borrower shall be responsible for any wire transfer fees imposed by Lender in accordance with its fee schedule governing Borrower's bank transactions. Any wire requests and/or instructions submitted after 10:00 a.m. PT on the date of disbursement of the

Delayed Draw may not be deposited until the next Business Day. Furthermore, if Borrower does not exercise its option to request the Delayed Draw on or before the Delayed Draw Deadline, Lender shall have no further obligation to advance the Delayed Draw.

2.2. <u>Prepayments.</u> The Term Loan may be prepaid in whole or in part upon thirty (30) days prior written notice to the Lender. In the event of any prepayment of the Term Loan, whether by voluntary prepayment, acceleration or otherwise, the Borrower shall pay to Lender a prepayment charge equal to (A) in the case of a prepayment made on or prior to the first twelve (12) month anniversary of the Closing Date, one-half percent (0.50)% of the principal balance being prepaid, (B) in the case of a prepayment made after the first twelve (12) month anniversary of the Closing Date, one-quarter percent (0.25%) of the principal balance being prepaid, and (C) in the case of a prepayment made after the eighteenth (18) month anniversary of the Closing Date, one-quarter percent (0.25%) of the principal balance being prepaid, and (C) in the case of a prepayment made after the eighteenth (18) month anniversary of the Closing Date, zero percent (0%) of the principal balance being prepaid.

### 2.3. Manner and Repayment of Term Loan.

- (a) The Term Loan shall be due and payable in full on the last day of the Term subject to earlier prepayment as herein provided. Notwithstanding the foregoing, Term Loan shall be subject to earlier repayment upon (x) acceleration upon the occurrence of an Event of Default under this Agreement or (y) termination of this Agreement.
- (b) All payments of principal, interest and other amounts payable hereunder, or under any of the Loan Documents shall be made to Lender at the Payment Office not later than 3:00 p.m. PT on the due date therefor in Dollars in federal funds or other funds immediately available to Lender. Lender shall have the right to effectuate payment of any and all Obligations due and owing hereunder by charging Borrower's Account.
- (c) Except as expressly provided herein, all payments (including prepayments) to be made by Borrower on account of principal, interest, fees and other amounts payable hereunder shall be made without deduction, setoff or counterclaim and shall be made to Lender, in each case on or prior to 3:00 p.m.PT, in Dollars and in immediately available funds.
- 2.4. <u>Collateral Shortfall</u>. If, on any day, the LTV Ratio is equal to or greater than seventy-five percent (75%) (a "Collateral Shortfall"), the Borrower shall have one (1) Business Day to cause the LTV Ratio to be less than or equal to sixty-five percent (65%). If the LTV Ratio is not less than or equal to sixty-five percent (65%) by 1 p.m. PST on the first (1st) Business Day after the occurrence of a Collateral Shortfall, an Event of Default shall be deemed to exist, in which case Lender shall be entitled to exercise any and all rights and remedies under Article XI and/or applicable law.
- 2.5. <u>Use of Proceeds</u>. Borrower shall apply the proceeds of the Term Loan to (i) provide financing for capital expenditures, refinance of existing debt and general corporate expenses related to the Borrower's bitcoin mining business (collectively, the "<u>Transactions</u>"), and (ii) pay fees and expenses relating to the Transactions.

2.6 Return of Excess Collateral. If on any day, the LTV Ratio is less than sixty-five percent (65%) as a result of excess Collateral in the Collateral Account (the "Excess Collateral"). Borrower may, but not more frequently than once per Business Day and limited to no more than two (2) requests per work week (i.e. Monday through Friday), request a return of some or all of the Excess Collateral. Requests for a return of Excess Collateral must be in writing (which may be delivered by email) and be for a minimum amount of at least \$1,000,000.00 (or equivalent amount of BTC). Borrower shall deliver written requests to the Custodian. Lender shall use commercially reasonable efforts to authorize the Custodian to process such requests within two (2) Business Days after Lender's receipt of the request. In no event shall Borrower be entitled to a return of any Collateral, if, after giving effect to the return of that Collateral, the principal amount outstanding of the Term Loan (together with any accrued but unpaid interest and any other amounts owed to Lender) would result in sixty-five percent (65%) LTV Ratio or greater.

#### III. INTEREST AND FEES.

- 3.1. Interest. Subject to the last sentence of this paragraph, the Obligations shall bear interest at the Interest Rate. Interest on the Term Loan shall be payable in arrears on each Interest Payment Date; provided that all accrued and unpaid interest shall be due and payable at the end of the Term Loan. Upon and after the occurrence of an Event of Default, and during the continuation thereof, at the option of Lender, or in the case of any Event of Default under Article X, immediately and automatically upon the occurrence of any such Event of Default without the requirement of any affirmative action by any party, the Obligations shall bear interest at the lesser of: (a) the Interest Rate, plus six percent (6%); or (b) the Maximum Rate (the "Default Rate").
- 3.2. <u>Origination Fee</u>. Borrower shall pay to Lender an origination fee of \$150,000.00 upon the execution of this Agreement.
- 3.3. <u>Contingent Draw Fee</u>. Borrower shall pay to Lender a contingent draw fee in the amount of \$250,000.00 (the, "Contingent Draw Fee") upon the execution of this Agreement. This Contingent Draw Fee will be refunded by Lender to Borrower if Borrower borrows the Delayed Draw (as outlined in Section 2.1 above) by no later than November 25, 2022.
- 3.4. <u>Computation of Interest and Fees.</u> Interest and fees hereunder shall be computed on the basis of a year of 360 days and for the actual number of days elapsed. If any payment to be made hereunder becomes due and payable on a day other than a Business Day, the due date thereof shall be extended to the next succeeding Business Day and interest thereon shall be payable at the applicable Interest Rate for the outstanding principal balance of the Term Loan during such extension
- 3.5. <u>Maximum Charges</u>. In no event whatsoever shall interest and other charges charged hereunder exceed the highest rate permissible under Applicable Law. In the event interest and other charges as computed hereunder would otherwise exceed the highest rate permitted under Applicable Law: (a) the interest rates hereunder will be reduced to the maximum rate permitted under Applicable Law; (b) such excess amount shall be first applied to any unpaid principal balance owed by Borrower; and (c) if the then remaining excess amount is greater than the previously unpaid principal balance, Lender shall promptly refund such excess amount to Borrower and the provisions hereof shall be deemed amended to provide for such permissible rate.

Notwithstanding anything to the contrary contained in this Agreement or in any Loan Document, all agreements which either now are or which shall become agreements among Loan Parties and Lender are hereby limited so that in no contingency or event whatsoever shall the total liability for payments in the nature of interest, additional interest and other charges exceed the applicable limits imposed by any applicable usury laws. If any payments in the nature of interest, additional interest and other charges made under this Agreement or any Loan Document are held to be in excess of the limits imposed by any applicable usury laws, it is agreed that any such amount held to be in excess shall be considered payment of principal hereunder, and the indebtedness evidenced hereby shall be reduced by such amount so that the total liability for payments in the nature of interest, additional interest and other charges shall not exceed the applicable limits imposed by any applicable usury laws, in compliance with the desires of Loan Parties and Lender. The foregoing provisions shall never be superseded or waived and shall control every other provision of this Agreement or any Loan Document and all agreements among Borrower and Lender, or their respective successors and assigns. If the applicable state or federal law is amended in the future to allow a greater rate of interest to be charged under this Agreement than is presently allowed by applicable state or federal law, then the limitation of interest hereunder shall be increased to the maximum rate of interest allowed by applicable state or federal law as amended, which increase shall be effective hereunder on the effective date of such amendment, and all interest charges owing to Lender by reason thereof shall be payable in accordance with Section 3.1 of this Agreement.

- 3.6. <u>Increased Costs.</u> In the event that any Applicable Law or any Change in Law or compliance by Lender with any request or directive (whether or not having the force of law) from any central bank or other financial, monetary or other authority, shall:
- (a) subject Lender to any tax of any kind whatsoever with respect to this Agreement, or change the basis of taxation of payments to Lender in respect;
- (b) impose, modify or deem applicable any reserve, special deposit, assessment, special deposit, compulsory loan, insurance charge or similar requirement against assets held by, or deposits in or for the account of, advances or loans by, or other credit extended by, any office of Lender, including pursuant to Regulation D of the Board of Governors of the Federal Reserve System; or
- (c) impose on Lender any other condition, loss or expense (other than Taxes) affecting this Agreement or any Loan Document or the Term Loan made by Lender, and the result of any of the foregoing is to increase the cost to Lender of making, converting to, continuing, renewing or maintaining the Term Loan hereunder by an amount that Lender deems to be material or to reduce the amount of any payment (whether of principal, interest or otherwise) in respect of the Term Loan by an amount Lender deems to be material, then, in any case Borrower shall promptly pay Lender, upon its demand, such additional amount as will compensate Lender for such additional cost or such reduction, as the case may be, provided that the foregoing shall not apply to increased costs which are reflected in the Base Rate, as the case may be. Lender shall certify the amount of such additional cost or reduced amount to Borrower, and such certification shall be conclusive absent manifest error. For purposes of this Section 3.6, the Dodd-Frank Wall Street Reform and Consumer Protection Act and any and all rules, regulations, orders, requests, guidelines, and directives adopted, promulgated or implemented in connection therewith are

deemed to have been introduced and adopted after the date of this Agreement. Any such surcharge shall be limited to the direct increase in taxes or other fees caused thereby, and no convenience fee or other surcharges may be added by Lender.

3.7. Application of Payments. So long as no Event of Default then exists, all payments from Borrower received by Lender shall first be applied to accrued but unpaid interest; second, to the principal balance outstanding; and third, to any late charges and costs of collection. Upon the occurrence of an Event of Default, Lender may apply payments in such order and amounts as Lender elects, in its reasonable discretion.

#### 3.8. Taxes

- (a) Any and all payments by or on account of any Obligations hereunder or under any Loan Document shall be made free and clear of and without reduction or withholding for any Indemnified Taxes or Other Taxes; <a href="provided">provided</a> that if Borrower shall be required by Applicable Law to deduct any Indemnified Taxes (including any Other Taxes) from such payments, then (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section) Lender receives an amount equal to the sum it would have received had no such deductions been made, (ii) Borrower shall make such deductions and (iii) Borrower shall timely pay the full amount deducted to the relevant Governmental Body in accordance with Applicable Law.
- (b) Without limiting the provisions of Section 3.8(a) above, Borrower shall timely pay any Other Taxes to the relevant Governmental Body in accordance with Applicable Law.
- (c) Borrower shall indemnify Lender, within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section) paid by Lender and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Body. A certificate as to the amount of such payment or liability delivered to Borrower by Lender shall be conclusive absent manifest error.
- (d) As soon as practicable after any payment of Indemnified Taxes or Other Taxes by Borrower to a Governmental Body, Borrower shall deliver to Lender the original or a certified copy of a receipt issued by such Governmental Body evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to Lender.
- (e) If Lender determines, in its reasonable discretion, that it has received a refund of any Indemnified Taxes or Other Taxes as to which it has been indemnified by Borrower or with respect to which Borrower have paid additional amounts pursuant to this Section, it shall pay to Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by Borrower under this Section with respect to the Indemnified Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses of the Lender and without interest (other than any interest paid by the relevant Governmental Body with respect

to such refund); provided that Borrower, upon the request of Lender, agrees to repay the amount paid over to Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Body) to Lender in the event Lender is required to repay such refund to such Governmental Body. This Section shall not be construed to require Lender to make available its tax returns (or any other information relating to its taxes that it deems confidential) to Borrower or any other Person.

### IV. COLLATERAL: GENERAL TERMS.

- 4.1. <u>Security Interest in the Collateral</u>. To secure the prompt payment and performance to Lender of the Obligations, Borrower hereby assigns, pledges and grants to Lender, a continuing security interest in and to and Lien on all of the Collateral, whether now owned or existing or hereafter created, acquired or arising and wheresoever located.
- Perfection of Security Interest. Borrower shall take all action that may be necessary or commercially reasonably desirable, or that Lender may commercially reasonably request, so as at all times to maintain the validity, perfection, enforceability and priority of Lender's security interest in and Lien on the Collateral or to enable Lender to protect, exercise or enforce its rights hereunder and in the Collateral, including, but not limited to, (a) immediately discharging all Liens on the Collateral other than Lender's security interest, (b) entering into custodial arrangements satisfactory to Lender, and (c) executing and delivering financing statements, control agreements, instruments of pledge, notices and assignments, in each case in form and substance satisfactory to Lender, relating to the creation, validity, perfection, maintenance or continuation of Lender's security interest and Lien under the UCC or other Applicable Law. By its signature hereto, Borrower hereby authorizes Lender to file against such Borrower, one or more financing, continuation or amendment statements pursuant to the UCC in form and substance satisfactory to Lender, on the condition that the description of the collateral in any such statement only is limited to only the Collateral. Subject to any expense limitation set forth in Section 13.9 all charges, expenses and fees Lender may incur in doing any of the foregoing, and any local taxes relating thereto, shall be charged to Borrower's Account, and added to the Obligations, or, at Lender's option, shall be paid by Borrower to Lender immediately upon demand.
- 4.3. Preservation of Collateral. Following the occurrence of a Default or Event of Default, in addition to the rights and remedies set forth in Section 11.1 hereof, Lender: (a) may at any time take such steps as Lender deems necessary to protect Lender's interest in and to preserve the Collateral, including the hiring of cyber and security consultants or implementing other security protection measures as Lender may deem appropriate; (b) may purchase or lease cold storage vaults to preserve the Collateral and to otherwise do all acts necessary to protect Lender's interests in the Collateral; and (c) may acquire one or more hardware wallets where Lender may store the private keys associated with all or part of the Collateral. Borrower shall cooperate fully with all of Lender's efforts to preserve the Collateral and will take such actions to preserve the Collateral as Lender may direct. Borrower shall reimburse Lender all of Lender's expenses of preserving the Collateral, including any expenses relating to the bonding or insuring of a custodian, which reimbursement obligation shall be part of the Obligations. Nothing contained in this Section 4.3 is intended to impose any burden or obligation on Lender to preserve or hold any portion of the Collateral, and Lender may immediately, with or without notice, liquidate some or all of the Collateral following an Event of Default.

- 4.4. <u>Ownership and Location of Collateral</u>. With respect to the Collateral, at the time the Collateral becomes subject to Lender's security interest, Borrower shall be the sole owner of and fully authorized and able to sell, transfer, pledge and/or grant a first priority security interest in each and every item of its respective Collateral to Lender, and the Collateral shall be free and clear of all Liens whatsoever.
- 4.5. <u>Defense of Lender's Interests.</u> Until (a) payment and performance in full of all of the Obligations and (b) termination of this Agreement, Lender's interests in the Collateral shall continue in full force and effect. During such period Borrower shall not, without Lender's prior written consent, pledge, sell, assign, transfer, create or suffer to exist a Lien upon or encumber or allow or suffer to be encumbered in any way, any part of the Collateral. Borrower shall defend Lender's interests in the Collateral against any and all Persons whatsoever. At any time, following demand by Lender for payment of all Obligations during the continuance of an Event of Default, Lender shall have the right to take control of the Collateral in whatever manner stored, including hardware wallets, software wallets, whether encrypted or unencrypted, and all medium upon which the private keys associated with the Collateral are retrievable. In addition, with respect to all Collateral, Lender shall be entitled to all of the rights and remedies set forth herein and further provided by the Uniform Commercial Code or other Applicable Law.
- 4.6. <u>Financing Statements</u>. Except with respect to the financing statements filed by Lender, no financing statement covering any of the Collateral or any proceeds thereof is or will be on file in any public office.

### 4.7 Custodian

- (a) Lender shall have no liability to Borrower or other person for Custodian's failure to comply with any of its contractual obligations with respect to the Collateral or any other acts or omissions of Custodian, including, without limitation, any losses or damages arising out of the theft or misappropriation of the Collateral or claims that Custodian failed to properly safeguard the Collateral. Any diligence performed by Lender with respect to the qualifications and suitability of Custodian is solely for Lender's benefit. Lender makes no representations or statements about the Custodian's experience or qualifications to perform its obligations under the Collateral Covenants. Borrower is engaged in commercial activities involving virtual currency and digital assets, has significant experience buying, selling and storing virtual currency and other digital assets, including bitcoin, and has done its own evaluation and diligence concerning Custodian. By entering into this Agreement and accepting the Term Loan, Borrower represents and warrants to Lender that it has performed independent diligence on the Custodian and determined Custodian to be qualified to custody the Collateral. Borrower is under no obligation to enter into the transactions evidenced by this Agreement and does so only after making the determination in the immediately preceding sentence.
- (b) In the event Borrower suffers any loss or damages as a result of Custodian's acts or omissions, including without limitation, any theft or misappropriation of the Collateral, Borrower shall look solely to Custodian for any such loss or damages. Borrower hereby waives and releases Lender from any and all claims, liabilities, damages, losses or otherwise caused by Custodian or Custodian's employees, agents or contractors, including, without limitation, any claims or causes of action based on the theory that Borrower sustained damages or losses as a result of Lender

conditioning the making of the Term Loan on the requirement that a Lender-approved custodian control the Collateral during the term of the Term Loan.

- (c) Without limiting the generality of the provisions in subparagraph (a) above, Borrower also acknowledges that it shall have no right to offset any amounts owed under this Agreement or any Loan Documents by any claims against Custodian, including, without limitation, claims arising out of the theft or misappropriation of any of the Collateral, it being understood that Borrower's obligations hereunder are absolute and irrespective of any such losses or claims.
- (d) Borrower acknowledges that Lender may have entered into separate agreements with Custodian and/or one or more Affiliates of Custodian pursuant to which Lender compensates Custodian or Affiliates of Custodian, including, without limitation, for performing certain loan servicing functions with respect to the Collateral and/or the Term Loan. The obligations of the Custodian or the Affiliates of Custodian under those agreements to service certain aspects of the Term Loan and provide trade execution services are for the sole benefit of Lender and Borrower is not an intended third-party beneficiary under any such agreements.
- (e) Lender shall not be liable to Borrower for any delays in the return of any Collateral to Borrower resulting from any delays caused by the Custodian or any other cause outside the reasonable control of Lender, including, without limitation, any disruption of the bitcoin network or governmental directive preventing the lawful transfer of the Collateral to Borrower.
- (f) Custodian shall constitute a "securities intermediary" as that term is defined in Article 8 of the Uniform Commercial Code of the State of New York ("Article 8"), or an agent thereof. Borrower and Lender acknowledge and agree that (i) the Collateral Account is considered a "securities account" under Article 8, and (iii) the Collateral in the Collateral Account shall be considered "financial assets" under Article 8. The treatment of Collateral as a financial asset under Article 8 does not determine the characterization or treatment of those assets under any other law or rule.
- Lender may from time to time require Borrower to establish the Collateral Account with a substitute Custodian (each, a "Replacement Custodian"). Upon Borrower's receipt of Lender's written request to establish a Collateral Account with a substitute Custodian, Borrower shall, within thirty (30) Business Days after receiving Lender's request, (i) cause such replacement Collateral Account to be established, (ii) transfer the Collateral to the replacement Collateral Account, and (iii) enter into, and cause the substitute Custodian to enter into, an account control agreement in form and substance acceptable to Lender. Borrower's failure to perform the foregoing obligations within that thirty (30) Business Days period shall entitle Lender to immediately (x) cause the liquidation of some or all of the Collateral and the application of the proceeds to repay all outstanding Obligations, and/or (y) direct the Custodian to transfer some or all of the Collateral to another account under the exclusive control of the Lender, which other account may be maintained with another custodian or subject to self-custody (the actions in clauses (x) and (y), collectively, "Protective Measures". Notwithstanding anything to the contrary contained herein, in the event Lender determines a material adverse change has occurred with respect to the then current Custodian, Lender may immediately take such Protective Measures as Lender deems necessary or prudent, in its reasonable discretion, without any prior notice to, or consent from, Borrower. In the event Lender exercises any right to liquidate Collateral under this

Section 4.7(g), Borrower shall remain liable for any remaining outstanding principal balance of the Term Loan after application of the proceeds from the liquidated Collateral. Subject to Borrower's timely compliance with the foregoing provisions of this Section 4.7(g), in the event Lender previously caused the Collateral to be transferred to another custodian or to self-custody the Collateral, Borrower may thereafter request that Lender custody the Collateral with a substitute Custodian mutually acceptable to both Lender and Borrower. If, after thirty (30) days from the date Borrower provides such written request to Lender, Borrower and Lender are unable to agree upon a substitute Custodian, Borrower may, without any prepayment fees, repay the outstanding principal balance of the Term Loan, together with all accrued and unpaid interest and all other amounts due under the Loan Documents. Borrower acknowledges that the provisions of this Section 4.7 are not intended to condition the effectiveness of Lender's designation of a substitute Custodian on Borrower's consent or approval.

#### V. REPRESENTATIONS AND WARRANTIES.

Borrower represents and warrants as follows:

- Authority. Borrower has full power, authority and legal right to enter into this Agreement and the Loan Documents to which it is a party and to perform all its respective Obligations hereunder and thereunder. This Agreement and the Loan Documents to which it is a party have been duly executed and delivered by Borrower, and this Agreement and the Loan Documents to which it is a party constitute the legal, valid and binding obligation of such Borrower enforceable in accordance with their terms, except as such enforceability may be limited by any applicable bankruptcy, insolvency, moratorium or similar laws affecting creditors' rights generally. The execution, delivery and performance of this Agreement and of the Loan Documents to which it is a party (a) are within such Borrower's corporate or company powers, as applicable, have been duly authorized by all necessary corporate or company action, as applicable, are not in contravention of law or the terms of such Borrower's Organizational Documents or to the conduct of such Borrower's business or of any material contract or undertaking to which such Borrower is a party or by which such Borrower is bound, (b) will not conflict with or violate any law or regulation, or any judgment, order or decree of any Governmental Body, (c) will not require the Consent of any Governmental Body, any party to a material contract or any other Person, and (d) will not conflict with, nor result in any breach in any of the provisions of or constitute a default under or result in the creation of any Lien upon any asset of such Borrower (other than the liens created hereunder in favor of Lender) under the provisions of any agreement, instrument, or other document to which such Borrower is a party or by which it or its property is a party or by which it may be bound.
- 5.2. <u>Formation and Qualification</u>. Borrower is duly incorporated or formed, as applicable, and in good standing under the laws of the state in which it is organized and is qualified to do business and is in good standing in the states where it conducts business which constitute all states in which qualification and good standing are necessary for Borrower to conduct its business and own its property and where the failure to so qualify could reasonably be expected to have a Material Adverse Effect on such Borrower. Borrower has delivered to Lender true and complete copies of its Organizational Documents and will promptly notify Lender of any amendment or changes thereto.

- 5.3. <u>Survival of Representations and Warranties</u>. All representations and warranties of such Borrower contained in this Agreement and the Loan Documents to which it is a party shall be true at the time of such Borrower's execution of this Agreement and the Loan Documents to which it is a party, and shall survive the execution, delivery and acceptance thereof by the parties thereto and the closing of the transactions described therein or related thereto.
- 5.4. Tax Returns. Borrower has filed all federal, state and local tax returns, to the extent required, and other reports each is required by law to file and has paid all taxes, assessments, fees and other governmental charges that are due and payable. The provision for taxes on the books of Borrower is adequate for all years not closed by applicable statutes, and for its current fiscal year, and no Borrower has any knowledge of any deficiency or additional assessment in connection therewith not provided for on its books.
- 5.5. <u>Financial Statements</u>. The December 31, 2020, December 31, 2021 and year-to-date March 31, 2022 balance sheets of Borrower (the "<u>Balance Sheet</u>") furnished to Lender are accurate, complete and correct in all material respects and fairly reflects the financial condition of Borrower as of the Closing Date, and has been prepared in accordance with GAAP, consistently applied. The Balance Sheet has been certified as accurate, complete and correct in all material respects by the Chief Financial Officer of Borrower. All financial statements referred to in this section 5.5, including the related schedules and notes thereto, have been prepared in accordance with GAAP, except as may be disclosed in such financial statements.
- 5.6. <u>Entity Names</u>. Except as disclosed to Lender, in writing, during the application process, Borrower has not been known by any other company or corporate name, as applicable, in the past five (5) years, nor has Borrower been the surviving corporation or company, as applicable, of a merger or consolidation or acquired all or substantially all of the assets of any Person during the preceding five (5) years.

### 5.7. Solvency; No Litigation, Violation, Indebtedness or Default; ERISA Compliance.

- (a) (i) Borrower is, and after giving effect to the Term Loan, Borrower will be solvent, able to pay its debts as they mature, and has, and after giving effect to the Term Loan, will have capital sufficient to carry on its business and all businesses in which it is about to engage, (ii) as of the Closing Date, the fair present saleable value of its assets, calculated on a going concern basis, is in excess of the amount of its liabilities, and (iii) subsequent to the Closing Date, the fair saleable value of its assets (calculated on a going concern basis) will be in excess of the amount of its liabilities.
- (b) Except as disclosed by Borrower to Lender in writing prior to the Closing Date, no Borrower has any pending or threatened litigation, arbitration, actions or proceedings. No Borrower has any outstanding Indebtedness other than the Obligations, except for (i) Indebtedness reflected on the Borrower's financial statements delivered to Lender, and (ii) Indebtedness incurred in the ordinary course of Borrower's business that is not secured by any lien, security interest other charge in or against the Collateral.
- (c) Borrower is not in violation of any applicable statute, law, rule, regulation or ordinance in any respect which could reasonably be expected to have a Material Adverse Effect,

nor is Borrower in violation of any order of any court, Governmental Body or arbitration board or tribunal

- (d) The Borrower and each member of the Controlled Group have fulfilled their obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and have not incurred any liability to the PBGC or a Plan under Title IV of ERISA in connection with the termination of a Plan, and each Plan maintained by the Borrower is in compliance with the presently applicable provisions of ERISA and the Code.
- 5.8. <u>Licenses and Permits</u>. Borrower (a) is in compliance with and (b) has procured and is now in possession of, all material licenses or permits required by any applicable federal, state or local law, rule or regulation for the operation of its business in each jurisdiction wherein it is now conducting or proposes to conduct business and where the failure to procure such licenses or permits could reasonably be expected to have a Material Adverse Effect.
- 5.9. <u>Default of Indebtedness</u>. No Borrower is in default in the payment of the principal of or interest on any Indebtedness or under any instrument or agreement under or subject to which any Indebtedness has been issued and no event has occurred under the provisions of any such instrument or agreement which with or without the lapse of time or the giving of notice, or both, constitutes or would constitute an event of default thereunder.
- 5.10. No Default. No Borrower is in default in the payment or performance of any of its contractual obligations and no Default or Event of Default has occurred.
- 5.11. <u>Disclosure</u>. No representation or warranty made by Borrower in this Agreement or in any financial statement, report, certificate or any other document furnished in connection herewith contains any untrue statement of a material fact or omits to state any material fact necessary to make the statements herein or therein not misleading. There is no fact known to Borrower or which reasonably should be known to such Borrower which such Borrower has not disclosed to Lender in writing with respect to the transactions contemplated by or this Agreement which could reasonably be expected to have a Material Adverse Effect.
- 5.12. <u>Business and Property of Borrower</u>. Upon and after the Closing Date, Borrower does not propose to engage in any business other than the business it currently operates as of the Effective Date, and activities incidental thereto or otherwise necessary to conduct the foregoing. On the Closing Date, Borrower will own all the property and possess all of the rights and Consents necessary for the conduct of the business of such Borrower.
- 5.13. <u>Certificate of Beneficial Ownership</u>. The Certificate of Beneficial Ownership executed and delivered to Lender for Borrower on or prior to the Closing Date, as updated from time to time in accordance with this Agreement, is accurate, complete and correct as of the date hereof and as of the date any such update is delivered. The Borrower acknowledges and agrees that the Certificate of Beneficial Ownership is one of the Loan Documents.
- 5.14 <u>Eligible Contract Participant</u>. Borrower qualifies as an Eligible Contract Participant.

### 5.15 Intentionally omitted.

### VI. AFFIRMATIVE COVENANTS

Borrower shall, until payment in full of the Obligations and termination of this Agreement:

- 6.1. <u>Compliance with Laws</u>. Comply with all Applicable Laws with respect to the Collateral or any part thereof or to the operation of such Borrower's business the non-compliance with which could reasonably be expected to have a Material Adverse Effect (except to the extent any separate provision of this Agreement shall expressly require compliance with any particular Applicable Law(s) pursuant to another standard). Borrower may, however, contest or dispute any Applicable Laws in any reasonable manner, provided that any related Lien is inchoate or stayed and sufficient reserves are established to the reasonable satisfaction of Lender to protect Lender's Lien on or security interest in the Collateral.
- 6.2. Conduct of Business and Maintenance of Existence and Assets. (a) Conduct continuously and operate actively its business according to good business practices; (b) keep in full force and effect its existence and comply in all material respects with the laws and regulations governing the conduct of its business where the failure to do so could reasonably be expected to have a Material Adverse Effect; and (c) make all such reports and pay all such franchise and other taxes and license fees and do all such other acts and things as may be lawfully required to maintain its rights, licenses, leases, powers and franchises under the laws of the United States or any political subdivision thereof where the failure to do so could reasonably be expected to have a Material Adverse Effect.
- 6.3. <u>Books and Records</u>. Keep proper books of record and account in which full, true and correct entries will be made of all dealings or transactions of or in relation to its business and affairs (including without limitation accruals for taxes), all in accordance with, or as required by, GAAP consistently applied in the opinion of such independent public accountant as shall then be regularly engaged by Borrower.
- 6.4. Payment of Taxes. Pay, when due, all taxes, assessments and other charges lawfully levied or assessed upon such Borrower or any of the Collateral, including personal property taxes, assessments and charges and all franchise, income, employment, social security benefits, withholding, and sales taxes. If any tax by any Governmental Body is or may be imposed on or as a result of any transaction between Borrower and Lender which Lender may be required to withhold or pay or if any taxes, assessments, or other charges remain unpaid after the date fixed for their payment, or if any claim shall be made which, in Lender's 's opinion, may possibly create a valid Lien on the Collateral, Lender may without notice to Borrower pay the taxes, assessments or other charges and Borrower hereby indemnifies and holds Lender harmless in respect thereof. The amount of any payment by Lender under this Section 6.4 shall be charged to Borrower' Account as an Advance and added to the Obligations and, until Borrower shall furnish Lender with an indemnity therefor (or supply Lender with evidence satisfactory to Lender that due provision for the payment thereof has been made), Lender may hold without interest any balance standing to Borrower' credit and Lender shall retain its security interest in and Lien on any and all Collateral held by Lender.

### 6.5. Financial Covenants

- (a) Adjusted Net Worth. Borrower shall maintain at all times an Adjusted Net Worth in an amount not less than \$350,000,000.00. Lender shall determine compliance quarterly based on Borrower's consolidated financial statements filed with the United States Securities and Exchange Commission (SEC). As used herein, the term "Adjusted Net Worth" means for Borrower the excess of its total assets over its total liabilities, with consideration for bitcoin at fair market value based on the date of such financial statement.
- (b) <u>Minimum Liquidity</u>. Borrower shall maintain minimum unrestricted and unencumbered cash of at least \$25,000,000.00, which shall be verified quarterly based on consolidated financial statements filed with the SEC for the preceding quarter.

### 6.6. Insurance.

- (a) (i) Borrower shall maintain, at a minimum, commercial general liability insurance in an amount satisfactory to Lender, and such other insurance coverages, as is customary in the case of companies engaged in businesses similar to Borrower, and having deductibles consistent with customary practice. Borrower shall require the Custodian to maintain coverage that insures against the theft, or the accidental or intentional loss or destruction of, the private keys associated with the Collateral, subject to certain exclusions from coverage that are generally applicable to all digital assets in the custody of the Custodian (the "Required Digital Assets Coverage"). During the time in which the Collateral remains in the custody of the Custodian and the Custodian maintains the Required Digital Asset Coverage, Borrower is not required to maintain separate coverage for losses covered by the Required Digital Assets Coverage.
- (b) With respect to any Required Digital Assets Coverage maintained by a Custodian, Borrower acknowledges that such coverages may not be sufficient to cover the entire loss in the event some or all of the Collateral is lost or stolen. Lender's acceptance of such insurance should not be considered as a representation, opinion or indication that the coverages provided by the Custodian are suitable for Borrower. In some cases, the amount of proceeds available on account of a loss may only be sufficient to repay the Obligations and may not be sufficient to cover some or all of Borrower's equity in the Collateral. As such, Borrower should evaluate with its risk management personnel what insurance coverage it should maintain independent of those maintained by the Custodian.
- 6.7. <u>Payment of Indebtedness</u>. Pay, discharge or otherwise satisfy at or before maturity (subject, where applicable, to specified grace periods) all its Indebtedness, except when the failure to do so would not reasonably be expected to have a Material Adverse Effect.
- 6.8. <u>Standards of Financial Statements</u>. Cause all financial statements referred to in Section 9.2 as to which GAAP is applicable to be complete and correct in all material respects (subject, in the case of interim financial statements, to normal year-end audit adjustments) and to be prepared in reasonable detail and in accordance with GAAP applied consistently throughout the periods reflected therein (except as disclosed therein and agreed to by such reporting accountants or officer, as applicable).

- 6.9. <u>Execution of Supplemental Instruments</u>. Execute and deliver to Lender from time to time, upon demand, such supplemental agreements, statements, assignments and transfers, or instructions or documents relating to the Collateral, and such other instruments as Lender may request, in order that the full intent of this Agreement may be carried into effect.
- 6.10. Certificate of Beneficial Ownership and Other Additional Information. Provide to Lender: (i) confirmation of the accuracy of the information set forth in the most recent Certificate of Beneficial Ownership provided to the Lender; (ii) a new Certificate of Beneficial Ownership, in form and substance satisfactory to Lender when the individual(s) to be identified as a Beneficial Owner have changed; and (iii) such other information and documentation as may reasonably be requested by Lender from time to time for purposes of compliance by Lender with applicable laws (including without limitation the USA PATRIOT ACT and other "know your customer" and antimoney laundering rules and regulations), and any policy or procedure implemented by Lender or such Lender to comply therewith.
- 6.11 Operational Security. Borrower shall follow and maintain operational security best practices throughout the Term of the Term Loan, including, without limitation, ensuring that all communications initiated by Borrower to Lender and/or Custodian involving private or public keys or other sensitive information are accomplished through secure means, such as encrypted email communication. Borrower shall employ adequate security measures to safeguard any passwords, personal identification numbers and other credentials that can be used to access information about the Term Loan, the Collateral and any deposits and/or withdrawals of digital assets involving the Term Loan

### VII. NEGATIVE COVENANTS.

Borrower shall not, until satisfaction in full of the Obligations and termination of this Agreement:

- 7.1. <u>Sale of Collateral</u>. Sell, lease, transfer or otherwise dispose of any Collateral in violation of this Agreement.
- 7.2. <u>Creation of Liens</u>. Create or suffer to exist any Lien or transfer upon or against any of the Collateral in violation of this Agreement.
- 7.3. <u>Indebtedness</u>. Create, incur, assume or suffer to exist any Indebtedness, except (i) in the Ordinary Course of Business and (ii) \$747,500,000.00 principal amount of 1.0% Convertible Senior Notes due 2026 and any restructurings thereof.
- 7.4. <u>Nature of Business</u>. Substantially change the nature of the business in which it is presently engaged, nor except as specifically permitted hereby purchase or invest, directly or indirectly, in any assets or property other than in the Ordinary Course of Business for assets or property which are useful in, necessary for and are to be used in its business as presently conducted.
- 7.5. <u>Fiscal Year and Accounting Changes</u>. Change its fiscal year end or make any change (a) in accounting treatment and reporting practices except as required by GAAP or (b) in

tax reporting treatment except as required by law or with the prior consent of Lender, which will not be unreasonably withheld or delayed.

- 7.6. <u>Pledge of Credit</u>. Now or hereafter pledge Lender's credit on any purchases, commitments or contracts or for any purpose whatsoever or use any portion of any Advance in or for any business other than such Borrower's business operations as conducted on the Closing Date.
- 7.7. <u>Compliance with ERISA</u>. Cause, or permit any member of the Controlled Group to cause, a representation or warranty in Section 5.7(d) to cease to be true and correct.

### VIII. CONDITIONS PRECEDENT.

- 8.1 The agreement of Lender to make the Initial Draw on the Closing Date is subject to the satisfaction, or waiver by Lender, immediately prior to or concurrently with the making of the Initial Draw, of the following conditions precedent:
- (a) <u>This Agreement</u>. Lender shall have received this Agreement duly executed and delivered by an authorized officer of Borrower;
- (b)  $\underline{\text{Note}}$ . Lender shall have received the Note duly executed and delivered by an authorized officer of Borrower;
- (c) <u>Loan Documents</u>. Lender shall have received each of the executed Loan Documents, as applicable, including, without limitation, and the Account Control Agreement in form and substance satisfactory to Lender;
- (d) <u>Collateral Value</u>. Lender shall have received evidence from Custodian that the aggregate amount of Collateral in the Collateral Account is sufficient in value and amount, at or prior to closing, to support the amount of the Term Loan on the Closing Date;
- (e) Net Worth and Liquidity. Lender shall have received satisfactory evidence that Borrower meets the Net Worth and Minimum Liquidity covenants;
- (f) Filings, Registrations and Recordings. Each document (including any UCC financing statement) required by this Agreement, any related agreement or under law or reasonably requested by Lender to be filed, registered or recorded in order to create, in favor of Lender, a perfected security interest in or lien upon the Collateral shall have been properly filed, registered or recorded in each jurisdiction in which the filing, registration or recordation thereof is so required or requested, and Lender shall have received an acknowledgment copy, or other evidence satisfactory to it, of each such filing, registration or recordation and satisfactory evidence of the payment of any necessary fee, tax or expense relating thereto;
- (g) <u>Secretary's Certificates, Authorizing Resolutions and Good Standings of Borrower</u>. Lender shall have received a certificate of the Secretary or Assistant Secretary (or other equivalent officer, partner or manager) of Borrower in form and substance satisfactory to Lender dated as of the Closing Date which shall certify (i) copies of resolutions in form and substance reasonably satisfactory to Lender, of the board of directors (or other equivalent governing body, member or partner) of such Borrower authorizing (x) the execution, delivery and performance of

this Agreement, the Notes and each Loan Document to which such Borrower is a party (including authorization of the incurrence of indebtedness and borrowing of the Term Loan), and (y) the granting by such Borrower of the security interests in and liens upon the Collateral to secure all of the Obligations of Borrower (and such certificate shall state that such resolutions have not been amended, modified, revoked or rescinded as of the date of such certificate), (ii) the incumbency and signature of the officers of such Borrower authorized to execute this Agreement and the Loan Documents, (iii) copies of the Organizational Documents of such Borrower as in effect on such date, complete with all amendments thereto, and (iv) the good standing (or equivalent status) of such Borrower in its jurisdiction of organization and each applicable jurisdiction where the conduct of such Borrower's business activities or the ownership of its properties necessitates qualification, as evidenced by good standing certificate(s) (or the equivalent thereof issued by any applicable jurisdiction) dated not more than thirty (30) days prior to the Closing Date, issued by the Secretary of State or other appropriate official of each such jurisdiction;

- (h) <u>Legal Opinion</u>. Lender shall have received in form and substance satisfactory to Lender, a legal opinion from Borrower's legal counsel as to good standing, due authorization, enforceability as such other matters as customarily required for transaction in the nature of the Term Loan;
- (i) No Litigation. No litigation, investigation or proceeding before or by any arbitrator or Governmental Body shall be continuing or threatened against Borrower or against the officers or directors of Borrower (A) in connection with this Agreement, the Note, the Term Loan Documents or any of the transactions contemplated thereby and which, in the reasonable opinion of Lender, is deemed material or (B) which could, in the reasonable opinion of Lender, have a Material Adverse Effect; and no injunction, writ, restraining order or other order of any nature materially adverse to Borrower or the conduct of its business or inconsistent with the due consummation of the transactions contemplated by this Agreement shall have been issued by any Governmental Body:
- (j) <u>Fees.</u> Lender shall have received all fees payable to Lender on or prior to the Closing Date hereunder, including pursuant to Article III hereof;
- (k) <u>Insurance</u>. Subject to the provisions of Section 6.6(a), Lender shall have received in form and substance satisfactory to Lender, (i) evidence that adequate insurance required to be maintained under this Agreement is in full force and effect, (ii) insurance certificates issued by Borrower's insurance broker containing such information regarding Borrower's insurance policies as Lender shall request;
- (l) <u>Other.</u> All corporate and other proceedings, and all documents, instruments and other legal matters in connection with the transactions contemplated by this Agreement shall be satisfactory in form and substance to Lender and its counsel; and
- (m) Representations and Warranties. Each of the representations and warranties made by Borrower in or pursuant to this Agreement, the Loan Documents and any related agreements to which it is a party, and each of the representations and warranties contained in any certificate, document or financial or other statement furnished at any time under or in connection with this Agreement, the Loan Documents or any related agreement shall be true and correct in all

respects on and as of such date as if made on and as of such date (except to the extent any such representation or warranty expressly relates only to any earlier and/or specified date).

- 8.2. <u>Conditions to Delayed Draw.</u> In addition to the conditions precedent set forth in Section 8.1, the agreement of Lender to make the Delayed Draw is subject to the satisfaction of the following conditions precedent as of the date such Delayed Draw is made:
- (a) <u>Request for Delayed Draw.</u> A properly completed Request for Delayed Draw received on or before the Delayed Draw Deadline;
- (b) No Default. No Event of Default or Default shall have occurred and be continuing on such date, or would exist after giving effect to the disbursement requested to be made, on such date; provided, however that Lender, in its sole discretion, may make the Delayed Draw notwithstanding the existence of an Event of Default or Default and that the disbursement of the Delayed Draw made shall not be deemed a waiver of any such Event of Default or Default;
- (c) <u>Maximum Disbursements</u>. After giving effect to the requested disbursement, the aggregate amount of disbursements shall not exceed the amount of the Term Commitment; and
- (d) <u>Collateral Value</u>. Lender shall have received evidence from Custodian that the aggregate amount of Collateral in the Collateral Account is sufficient in value and amount, at or prior to the disbursement of the Delayed Draw, to support the outstanding principal balance of the Term Loan on the day of disbursement.

Borrower's request for the Delayed Draw hereunder shall constitute a representation and warranty by Borrower as of the date of such disbursement of the Delayed Draw that the conditions contained in this Section shall have been satisfied.

### IX. INFORMATION AS TO BORROWER.

Borrower shall, until satisfaction in full of the Obligations and the termination of this Agreement:

- 9.1. <u>Disclosure of Material Matters</u>. Immediately upon learning thereof, report to Lender all matters materially affecting the value, enforceability or collectability of any portion of the Collateral, or any Lien other than Lender's security interest, placed upon or asserted against Borrower or any Collateral.
- 9.2. <u>Financial Statements.</u> Borrower shall, in accordance with Section 13 or 15(d) of the Securities and Exchange Act of 1934, file with the SEC: (a) an annual financial report on Form 10-K within ninety (90) days after the end of each of its fiscal years, and (b) quarterly financial reports on Form 10-Q forty-five (45) days after the end of each of its first three fiscal quarters, each of the preceding dates a "Reporting Due Date." The Borrower's Reporting Due Date for any Form 10-K or Form 10-Q may be extended for only such time as permitted by the SEC if the Borrower shall file with the SEC an accurate and complete SEC Form NT 10-K or NT 10-Q, as

applicable, prior to each respective Reporting Due Date and provide a copy of such Form NT to Lender immediately after filing.

#### 9.3. <u>Intentionally Omitted.</u>

9.4. <u>Collateral Account Reports</u>. At or prior to 1:00 p.m. PT each day during an Availability Period, the Borrower shall make available a report to the Lender that sets forth the total amount of bitcoin subject to the Collateral Account, in each case as of 1:00 p.m. PT on the immediately preceding day. Notwithstanding the foregoing, Borrower will not be required to submit such reports for so long as Lender does not provide written notice to Borrower that the applicable Custodian is no longer providing such reports directly to Lender.

#### 9.5. Intentionally Omitted.

- 9.6. <u>Additional Information</u>. Furnish Lender with such additional information as Lender shall reasonably request in order to enable Lender to determine whether the terms, covenants, provisions and conditions of this Agreement and the Notes have been complied with by Borrower including, without the necessity of any request by Lender, copies of all security audits and reviews.
- 9.7. <u>Notice of Suits, Adverse Events.</u> Furnish Lender with prompt written notice of (a) any lapse or other termination of any Consent issued to Borrower by any Governmental Body or any other Person that is material to the operation of Borrower's business, (b) any refusal by any Governmental Body or any other Person to renew or extend any such Consent; and (c) copies of any periodic or special reports filed by Borrower with any Governmental Body or Person, if such reports indicate any material change in the business, operations, affairs or condition of Borrower, or if copies thereof are requested by Lender, and (d) copies of any material notices and other communications from any Governmental Body or Person which specifically relate to Borrower.
- 9.8. <u>Financial Disclosure</u>. Borrower hereby irrevocably authorizes and directs all accountants and auditors employed by such Borrower at any time during the Term to exhibit and deliver to Lender copies of any of such Borrower's financial statements, trial balances or other accounting records of any sort in the accountant's or auditor's possession, and to disclose to Lender any information such accountants may have concerning such Borrower's financial status and business operations. Borrower hereby authorizes all Governmental Bodies to furnish to Lender copies of reports or examinations relating to such Borrower, whether made by such Borrower or otherwise; however, Lender will attempt to obtain such information or materials directly from such Borrower prior to obtaining such information or materials from such accountants or Governmental Bodies.

#### X. EVENTS OF DEFAULT.

The occurrence of any one or more of the following events shall constitute an "Event of Default":

10.1. Nonpayment. Failure by Borrower to pay (a) within ten (10) days of the date due, any scheduled installment of principal or interest on the Obligations, or (b) on or before the date

due, any other fee, charge, amount or liability provided for herein, including, without limitation, any amounts owed under Section 2.4, or in any Loan Document, in each case whether at maturity, by reason of acceleration pursuant to the terms of this Agreement, by notice of intention to prepay or by required prepayment.

- 10.2. <u>Breach of Representation</u>. Any representation or warranty made or deemed made by Borrower in this Agreement, any Loan Document or any related agreement or in any certificate, document or financial or other statement furnished at any time in connection herewith or therewith shall prove to have been incorrect or misleading in any material respect on the date when made or deemed to have been made; provided that, in each case, such inaccuracies, to the extent capable of being corrected, are not corrected within ten (10) days following notice from Lender.
- 10.3. <u>Financial Information</u>. Failure by Borrower to (a) file financial reports in accordance with Section 9.2, or (b) provide Lender with financial information it may request within three (3) Business Days of such requests or (c) permit the inspection of its books or records in accordance with this Agreement;
- 10.4. <u>Judicial Actions</u>. Issuance of a notice of Lien, levy, assessment, injunction or attachment (a) against any Collateral or (b) against a material portion of Borrower's other property which is not stayed or lifted within thirty (30) days;
- 10.5. <u>Noncompliance</u>. Except as otherwise provided for in Sections 10.1, 10.2, 10.3 and 10.4, (a) failure or neglect of Borrower, or any Person to perform, keep or observe any term, provision, condition, covenant herein contained, or contained in any Loan Document or any other agreement or arrangement, now or hereafter entered into between Borrower, or such Person, and Lender beyond any applicable cure and/or grace period;
- 10.6. <u>Judgments</u>. Any (a) judgment or judgments, writ(s), order(s) or decree(s) for the payment of money are rendered against Borrower for an aggregate amount in excess of \$500,000.00 and (b) (i) action shall be legally taken by any judgment creditor to levy upon assets or properties of Borrower to enforce any such judgment, (ii) such judgment shall remain undischarged for a period of thirty (30) consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, shall not be in effect, or (iii) any Liens arising by virtue of the rendition, entry or issuance of such judgment upon assets or properties of Borrower shall be senior to any Liens in favor of Lender on such assets or properties;
- 10.7. <u>Bankruptcy</u>. Borrower, any Subsidiary or Affiliate of Borrower shall (a) apply for, consent to or suffer the appointment of, or the taking of possession by, a receiver, custodian, trustee, liquidator or similar fiduciary of itself or of all or a substantial part of its property, (b) admit in writing its inability, or be generally unable, to pay its debts as they become due or cease operations of its present business, (c) make a general assignment for the benefit of creditors, (d) commence a voluntary case under any state or federal bankruptcy or receivership laws (as now or hereafter in effect), (e) be the subject of an Insolvency Event or otherwise be adjudicated a bankrupt or insolvent (including by entry of any order for relief in any involuntary bankruptcy or insolvency proceeding commenced against it), (f) file a petition seeking to take advantage of any other law providing for the relief of debtors, (g) acquiesce to, or fail to have dismissed, within

thirty (30) days, any petition filed against it in any involuntary case under such bankruptcy laws, or (h) take any action for the purpose of effecting any of the foregoing;

- 10.8. <u>Material Adverse Effect</u>. The occurrence of any event or development which could reasonably be expected to have a Material Adverse Effect;
- 10.9. <u>Lien Priority</u>. Any Lien created hereunder or provided for hereby or under any related agreement for any reason ceases to be or is not a valid and perfected Lien having a first priority interest in the Collateral;
- 10.10. Cross Default. Either (a) any specified "event of default" under any Indebtedness (other than the Obligations) of Borrower with a then-outstanding principal balance (or, in the case of any Indebtedness not so denominated, with a then-outstanding total obligation amount) of \$500,000.00 or more, or any other event or circumstance which would permit the holder of any such Indebtedness of Borrower to accelerate such Indebtedness (and/or the obligations of Borrower thereunder) prior to the scheduled maturity or termination thereof, shall occur (regardless of whether the holder of such Indebtedness shall actually accelerate, terminate or otherwise exercise any rights or remedies with respect to such Indebtedness), (b) a default of the obligations of Borrower under any other agreement to which it is a party shall occur which has or is reasonably likely to have a Material Adverse Effect, or (c) Lender declares a default or "event of default" under any other loan or extension of credit or other account agreement to Borrower or an Affiliate of Borrower;

#### 10.11. Intentionally Omitted.

- 10.12. <u>Change of Control</u>. Any Change of Control shall occur without Borrower's prior written notice to Lender and Lender's prior written approval thereof;
- 10.13. <u>Invalidity</u>. Any material provision of this Agreement or any Loan Document shall, for any reason, cease to be valid and binding on Borrower, or Borrower shall so claim in writing to Lender or Borrower challenges the validity of or its liability under this Agreement or any Loan Document;
- 10.14. <u>Seizures</u>. Any portion of the Collateral shall be seized, subject to garnishment or taken by a Governmental Body; and
- 10.15. <u>Anti-Money Laundering/International Trade Law Compliance</u>. Any representation or warranty contained in Section 13.16 is or becomes false or misleading at any time.

#### XI. LENDER'S RIGHTS AND REMEDIES AFTER DEFAULT.

#### 11.1. Rights and Remedies.

(a) During the continuance of: (i) an Event of Default pursuant to Section 10.7 (other than Section 10.7(g)), all Obligations shall be immediately due and payable accruing interest at the Default Rate, and (ii) any of the other Events of Default and at any time thereafter, at the option of Lender, shall be immediately due and payable. Upon the occurrence of any Event of Default, Lender shall have the right to exercise any and all rights and remedies provided for herein,

under the Loan Documents, under the Uniform Commercial Code and at law or equity generally, including the right to foreclose the security interests granted herein and to realize upon any Collateral by any available judicial or non-judicial procedure and/or to take possession of and sell any or all of the Collateral with or without judicial process. With or without having the Collateral at the time or place of sale, Lender may sell the Collateral, or any part thereof, at public or private sale, at any time or place, in one or more sales, at such price or prices, and upon such terms, either for cash, credit or future delivery, as Lender may elect. The cash proceeds realized from the sale of any Collateral shall be applied to the Obligations in the order determined by Lender in its reasonable discretion. Noncash proceeds will only be applied to the Obligations as they are converted into cash. If any deficiency shall arise, Borrower shall remain liable to Lender therefor. Notwithstanding anything to the contrary contained herein, Lender may sell, or direct any Custodian to sell, any bitcoin, Forked Assets or other digital assets in the Collateral Account without (i) any notice to Borrower, (ii) any publication of notice about a sale of such assets, it being acknowledged by Borrower that highly liquid and efficient markets exist for the sale of bitcoin, or (iii) any obligation to account to Borrower for any losses caused by Lender's decision to liquidate such Collateral.

To the extent that Applicable Law imposes duties on Lender to exercise remedies in a commercially reasonable manner, Borrower acknowledges and agrees that it is not commercially unreasonable for Lender, within customary and usual practices in the commercial banking industry in the U.S.: (i) to fail to incur expenses reasonably deemed significant by Lender to safeguard the Collateral pending disposition; (ii) to fail to obtain third party consents for access to Collateral to be disposed of, or to obtain or, if not required by other law, to fail to obtain governmental or third party consents for the collection or disposition of Collateral to be collected or disposed of; (iii) to fail to remove Liens on or any adverse claims against Collateral; (iv) if required under the UCC, to advertise dispositions of Collateral through publications or media of general circulation, whether or not the Collateral is of a specialized nature; (v) to contact other Persons, whether or not in the same business as Borrower, for expressions of interest in acquiring all or any portion of such Collateral; (vi) to hire one or more market makers and/or brokers specializing in digital asset transactions to assist in the disposition of Collateral; (vii) to dispose of Collateral by utilizing internet sites that provide for the auction of assets of the types included in the Collateral or that have the reasonable capacity of doing so, or that match buyers and sellers of assets; (viii) to dispose of assets in wholesale rather than retail markets; (ix) to disclaim disposition warranties, such as title, possession or quiet enjoyment, (x) to purchase insurance or credit enhancements to insure Lender against risks of loss, collection or disposition of Collateral or to provide to Lender a guaranteed return from the collection or disposition of Collateral; or (xi) to the extent deemed appropriate by the Lender, to obtain the services of other brokers, investment bankers, consultants and other professionals to assist Lender in the collection or disposition of any of the Collateral. Borrower acknowledges that the purpose of this Section 11.1(b) is to provide non-exhaustive indications of what actions or omissions by Lender would not be commercially unreasonable in Lender's exercise of remedies against the Collateral and that other actions or omissions by Lender shall not be deemed commercially unreasonable solely on account of not being indicated in this Section 11.1(b). Upon the occurrence of any Event of Default, Lender shall have the right to exercise any and all rights and remedies provided for herein, under the Loan Documents, under the Uniform Commercial Code and at law or equity generally, including the right to foreclose the security interests granted herein and to realize upon any Collateral by any

available judicial procedure and/or to take possession of and sell any or all of the Collateral with or without judicial process. Without limitation upon the foregoing, nothing contained in this Section 11.1(b) shall be construed to grant any rights to Borrower or to impose any duties on Lender that would not have been granted or imposed by this Agreement or by Applicable Law in the absence of this Section 11.1(b).

- 11.2. <u>Lender's Discretion</u>. Lender shall have the right in its commercially reasonable discretion to determine which rights, Liens, security interests or remedies Lender may at any time pursue, relinquish, subordinate, or modify, which procedures, timing and methodologies to employ, and what any other action to take with respect to any or all of the Collateral and in what order, thereto and such determination will not in any way modify or affect any of Lender's rights hereunder as against Borrower or each other.
- 11.3. <u>Setoff</u>. In addition to any other rights which Lender may have under Applicable Law, upon the occurrence of an Event of Default hereunder, Lender shall have a right, immediately and without notice of any kind, to apply Borrower's property held by Lender or any of its Affiliates to reduce the Obligations and to exercise any and all rights of setoff which may be available to Lender with respect to any deposits held by Lender.
- 11.4. <u>Liquidation without Notice</u>. Borrower acknowledges that the price of bitcoin is (a) volatile and thus may decline speedily in value, and (b) bitcoin is a type of asset customarily sold on a recognized market. Accordingly, Borrower acknowledges and agrees that it would not be necessary under Section 9-611(b) of the UCC to give notice of any proposed disposition of the Collateral
- 11.5. <u>Rights and Remedies not Exclusive</u>. The enumeration of the foregoing rights and remedies is not intended to be exhaustive and the exercise of any rights or remedy shall not preclude the exercise of any other right or remedies provided for herein or otherwise provided by law, all of which shall be cumulative and not alternative.

#### XII. WAIVERS AND JUDICIAL PROCEEDINGS.

- 12.1. Waiver of Notice. Borrower hereby waives notice of non-payment of any indebtedness, demand, presentment, protest and notice thereof with respect to any and all instruments, notice of acceptance hereof, notice of loans or advances made, credit extended, Collateral received or delivered, or any other action taken in reliance hereon, and all other demands and notices of any description, except such as are expressly provided for herein.
- 12.2. <u>Delay</u>. No delay or omission on Lender's 's part in exercising any right, remedy or option shall operate as a waiver of such or any other right, remedy or option or of any Default or Event of Default.

#### XIII. EFFECTIVE DATE AND TERMINATION.

13.1. <u>Term.</u> This Agreement, which shall inure to the benefit of and shall be binding upon the respective successors and permitted assigns of Borrower and Lender shall become

effective on the date hereof and shall continue in full force and effect until August 5, 2024 (the "Term") unless sooner terminated as herein provided.

- Termination. The termination of the Agreement shall not affect Lender's rights, or any of the Obligations having their inception prior to the effective date of such termination or any Obligations which pursuant to the terms hereof continue to accrue after such date, and the provisions hereof shall continue to be fully operative until all transactions entered into, rights or interests created and Obligations have been fully and indefeasibly paid, disposed of, concluded or liquidated. The security interests, Liens and rights granted to Lender hereunder and the financing statements filed hereunder shall continue in full force and effect, notwithstanding the termination of this Agreement or the fact that Borrower's account may from time to time be temporarily in a zero or credit position, until all of the Obligations of Borrower have been indefeasibly paid. Accordingly, Borrower waives any rights which it may have under the Uniform Commercial Code to demand the filing of termination statements with respect to the Collateral, and Lender shall not be required to send such termination statements to Borrower, or to file them with any filing office, unless and until this Agreement shall have been terminated in accordance with its terms and all Obligations have been indefeasibly paid in full in immediately available funds. All representations, warranties, covenants, waivers and agreements contained herein shall survive termination hereof until all Obligations are indefeasibly paid and performed in full.
- 13.3. Governing Law. This Agreement and each Loan Document (unless and except to the extent expressly provided otherwise in any such Loan Document), and all matters relating hereto or thereto or arising herefrom or therefrom (whether arising under contract law, tort law or otherwise) shall be governed by and construed in accordance with the laws of the State of New York applied to contracts to be performed wholly within the State of New York. Any judicial proceeding brought by or against Borrower with respect to any of the Obligations, this Agreement, the Loan Documents or any related agreement may be brought in any court of competent iurisdiction in the State of California or New York, United States of America, and, by execution and delivery of this Agreement, Borrower accepts for itself and in connection with its properties, generally and unconditionally, the non-exclusive jurisdiction of the aforesaid courts, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement. Borrower hereby waives personal service of any and all process upon it and consents that all such service of process may be made by certified or registered mail (return receipt requested) directed to Borrower at its address set forth in this Agreement and service so made shall be deemed completed five (5) days after the same shall have been so deposited in the mails of the United States of America, or, at Lender's option, by service upon Borrower which Borrower irrevocably appoints as such Borrower's agent for the purpose of accepting service within the State of New York or California, as applicable. Nothing herein shall affect the right to serve process in any manner permitted by law or shall limit the right of Lender to bring proceedings against Borrower in the courts of any other jurisdiction. Borrower waives any objection to jurisdiction and venue of any action instituted hereunder and shall not assert any defense based on lack of jurisdiction or venue or based upon forum non conveniens. Borrower waives the right to remove any judicial proceeding brought against such Borrower in any state court to any federal court. Any judicial proceeding by Borrower against Lender involving, directly or indirectly, any matter or claim in any way arising out of, related to or connected with this Agreement or any related

agreement, shall be brought only in a federal or state court located in New York or California, at Lender's discretion.

#### 13.4. Entire Understanding.

THIS AGREEMENT AND THE DOCUMENTS EXECUTED CONCURRENTLY HEREWITH CONTAIN THE ENTIRE UNDERSTANDING BETWEEN BORROWER AND LENDER AND SUPERSEDES ALL PRIOR AGREEMENTS AND UNDERSTANDINGS, IF ANY, RELATING TO THE SUBJECT MATTER OF THIS AGREEMENT. ANY PROMISES, REPRESENTATIONS, WARRANTIES OR GUARANTEES NOT HEREIN CONTAINED AND HEREINAFTER MADE SHALL HAVE NO FORCE AND EFFECT UNLESS IN WRITING, SIGNED BY BORROWER'S AND LENDER'S RESPECTIVE OFFICERS. Neither this Agreement nor any portion or provisions hereof may be changed, modified, amended, waived, supplemented, discharged, cancelled or terminated orally or by any course of dealing, or in any manner other than by an agreement in writing, signed by the party to be charged. Notwithstanding the foregoing, Lender may modify this Agreement or any of the Loan Documents for the purposes of completing missing content or correcting erroneous content of an administrative nature, without the need for a written amendment, provided that the Lender shall send a copy of any such modification to Borrower (which copy may be provided by electronic mail). Borrower acknowledges that it has been advised by counsel in connection with the execution of this Agreement and Loan Documents and is not relying upon oral representations or statements inconsistent with the terms and provisions of this Agreement.

#### 13.5. Successors and Assigns; Participations.

- (a) This Agreement shall be binding upon and inure to the benefit of Borrower and Lender, all future holders of the Obligations and their respective successors and assigns, except that Borrower may not assign or transfer any of its rights or obligations under this Agreement.
- (b) Borrower acknowledges that in the regular course of commercial banking business Lender may at any time and from time to time sell participating interests in the Obligations to other Persons. Each Participant may exercise all rights of payment (including rights of set-off) with respect to the portion of such Obligations held by it or other Obligations payable hereunder as fully as if such Participant were the direct holder thereof provided that (i) Borrower shall not be required to pay to any Participant more than the amount which it would have been required to pay to Lender which granted an interest in the Obligations payable hereunder to such Participant had Lender retained such interest in the Obligations payable hereunder unless the sale of the participation to such Participant is made with Borrower's prior written consent, and (ii) in no event shall Borrower be required to pay any such amount arising from the same circumstances and with respect to the other Obligations payable hereunder to both Lender and such Participant. Borrower hereby grants to any Participant a continuing security interest in any deposits, moneys or other property actually or constructively held by such Participant as security for the Participant's interest in the Obligations.
- 13.6. <u>Notice</u>. Any notice or request hereunder may be given to Borrower or to Lender at their respective addresses set forth below or at such other address as may hereafter be specified in

a notice designated as a notice of change of address under this Section. Any notice, request, demand, direction or other communication (for purposes of this Section 13.6 only, a "Notice") to be given to or made upon any party hereto under any provision of this Agreement shall be given or made by telephone or in writing (which includes by means of electronic transmission (i.e., "e-mail") or facsimile transmission or by setting forth such Notice on a website to which Borrower are directed (an "Internet Posting") if Notice of such Internet Posting (including the information necessary to access such site) has previously been delivered to the applicable parties hereto by another means set forth in this Section 13.6) in accordance with this Section 13.6. Any such Notice must be delivered to the applicable parties hereto at the addresses and numbers set forth under their respective names in this Section 13.6 hereof or in accordance with any subsequent unrevoked Notice from any such party that is given in accordance with this Section 13.6. Any Notice shall be effective:

- (a) In the case of hand-delivery, when delivered;
- (b) If given by mail, four (4) days after such Notice is deposited with the United States Postal Service, with first-class postage prepaid, return receipt requested;
- (c) In the case of a telephonic Notice, when a party is contacted by telephone, if delivery of such telephonic Notice is confirmed no later than the next Business Day by hand delivery, a facsimile or electronic transmission, an Internet Posting or an overnight courier delivery of a confirmatory Notice (received at or before noon on such next Business Day);
- (d) In the case of a facsimile transmission, when sent to the applicable party's facsimile machine's telephone number, if the party sending such Notice receives confirmation of the delivery thereof from its own facsimile machine;
  - (e) In the case of electronic transmission, when actually received;
- (f) In the case of an Internet Posting, upon delivery of a Notice of such posting (including the information necessary to access such site) by another means set forth in this Section 13.6; and
- (g) If given by any other means (including by overnight courier), when actually received.
  - (A) If to Lender at:

Silvergate Bank 4250 Executive Square, Suite 300 La Jolla, Ca 92037

Attention: Loan Servicing Telephone: (858) 362-6300 Facsimile: (858) 362-6323

Email: <u>loanservicing@silvergate.com</u>

(B) If to Borrower:

Marathon Digital Holdings, Inc. 1180 North Town Center Drive, Suite 100 Las Vegas, NV 89144 Attention: Fred Thiel Telephone: (725) 218-3415 Email: fred@marathondh.com

- 13.7. <u>Survival</u>. The obligations of Borrower under this Agreement shall survive termination of this Agreement and the Loan Documents and payment in full of the Obligations.
- 13.8. <u>Severability</u>. If any part of this Agreement is contrary to, prohibited by, or deemed invalid under Applicable Laws, such provision shall be inapplicable and deemed omitted to the extent so contrary, prohibited or invalid, but the remainder hereof shall not be invalidated thereby and shall be given effect so far as possible.
- 13.9. Expenses. Borrower shall pay all out-of-pocket expenses incurred by Lender and its Affiliates (including the reasonable fees, charges and disbursements of counsel for Lender), in connection with the preparation, negotiation, execution, delivery and administration of this Agreement and the Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated). Notwithstanding the forgoing, Borrower shall only pay out-of-pocket expenses in excess of \$10,000.00.
- 13.10. <u>Injunctive Relief.</u> Borrower recognizes that, in the event Borrower fails to perform, observe or discharge any of its obligations or liabilities under this Agreement, or threatens to fail to perform, observe or discharge such obligations or liabilities, any remedy at law may prove to be inadequate relief to Lender; therefor, Lender, if Lender so requests, shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving that actual damages are not an adequate remedy.

#### 13.11 Limitations of Liability.

- (a) Neither Lender, nor any agent or attorney for Lender, shall be liable to Borrower (or any Affiliate of any such Person) for indirect, punitive, exemplary or consequential damages arising from any breach of contract, tort or other wrong relating to the establishment, administration or collection of the Obligations or as a result of any transaction contemplated under this Agreement or any Loan Document.
- (b) Notwithstanding anything to the contrary contained in this Agreement, Lender's liability will not exceed the greater of (i) the fair market value of the amount of the Collateral at the time in which the events giving rise to the liability occurred and (ii) the fair market value of the amount of Collateral at the time that Lender notifies Borrower in writing or Borrower otherwise has knowledge of the events giving rise to the liability. The fair market value of each digital asset held as Collateral will be determined by Lender according to the valuation policy used by the Custodian, which may differ from the way the Borrower values its digital asset holdings.

- 13.12. <u>Captions</u>. The captions at various places in this Agreement are intended for convenience only and do not constitute and shall not be interpreted as part of this Agreement.
- 13.13. <u>Counterparts; Facsimile Signatures</u>. This Agreement may be executed in any number of and by different parties hereto on separate counterparts, all of which, when so executed, shall be deemed an original, but all such counterparts shall constitute one and the same agreement. Any signature delivered by a party by facsimile or other form of electronic transmission (including email transmission of a PDF image or the use of a third-party platform, including DocuSign) shall be deemed to be an original signature hereto.
- 13.14. <u>Construction</u>. The parties acknowledge that each party and its counsel have reviewed this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendments, schedules or exhibits thereto.
- 13.15. <u>Publicity</u>. Borrower hereby authorizes Lender to make appropriate announcements of the financial arrangement entered into among Borrower and Lender, including announcements which are commonly known as tombstones, in such publications and to such selected parties as Lender shall in its reasonable discretion deem appropriate.

#### 13.16. Anti-Terrorism Laws.

- (a) Borrower represents and warrants that (i) no Covered Entity is a Sanctioned Person and (ii) no Covered Entity, either in its own right or through any third party, (A) has any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Terrorism Law; (B) does business in or with, or derives any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law; or (C) engages in any dealings or transactions prohibited by any Anti-Terrorism Law.
- (b) Borrower covenants and agrees that (i) no Covered Entity will become a Sanctioned Person, (ii) no Covered Entity, either in its own right or through any third party, will (A) have any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Terrorism Law; (B) do business in or with, or derive any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law; (C) engage in any dealings or transactions prohibited by any Anti-Terrorism Law or (D) use the proceeds of the Term Loan to fund any operations in, finance any investments or activities in, or, make any payments to, a Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law, (iii) the funds used to repay the Obligations will not be derived from any unlawful activity, (iv) each Covered Entity shall comply with all Anti-Terrorism Laws and (v) the Borrower shall promptly notify the Lender in writing upon the occurrence of a Reportable Compliance Event.

#### 13.17 Enforcement Waivers

TO THE FULLEST EXTENT PERMITTED BY LAW, BORROWER HEREBY KNOWINGLY, INTENTIONALLY, AND VOLUNTARILY, WITH AND UPON THE

ADVICE OF COMPETENT COUNSEL, AGREES NOT TO ELECT A TRIAL BY JURY OF ANY ISSUE TRIABLE OF RIGHT BY JURY, AND WAIVES ANY RIGHT TO TRIAL BY JURY FULLY TO THE EXTENT THAT ANY SUCH RIGHT SHALL NOW OR HEREAFTER EXIST WITH REGARD TO THIS AGREEMENT, OR THE OTHER LOAN DOCUMENTS, OR ANY CLAIM, COUNTERCLAIM, OR OTHER ACTION ARISING IN CONNECTION THEREWITH. THIS WAIVER OF RIGHT TO TRIAL BY JURY IS INTENDED TO ENCOMPASS INDIVIDUALLY EACH INSTANCE AND EACH ISSUE AS TO WHICH THE RIGHT TO A TRIAL BY JURY WOULD OTHERWISE ACCRUE. LENDER IS HEREBY AUTHORIZED TO FILE A COPY OF THIS SECTION IN ANY PROCEEDING AS CONCLUSIVE EVIDENCE OF THIS WAIVER BY BORROWER.

Notwithstanding the foregoing to the contrary, in the event that the jury trial waiver contained herein shall be held or deemed to be unenforceable, Borrower hereby agrees that any controversy, dispute, or claim between the parties arising out of or relating to this Agreement shall be resolved by a reference proceeding in California in accordance with the provisions of California Code of Civil Procedure § 638. The referee shall be a retired California state court judge selected by mutual written agreement of the parties. If the parties are unable to agree upon a referee within ten (10) calendar days after one party serves a written notice of its intent to commence a judicial reference proceeding on the other party, then the referee will be selected by the court in accordance with California Code of Civil Procedure § 640(b). The referee shall be appointed to sit as a temporary judge, with all of the powers of a temporary judge, as authorized by law, and upon selection should take and subscribe to the oath of office as provided for in Rule 244 of the California Rules of Court (or any subsequently enacted Rule). The referee shall determine the manner in which the reference proceeding is conducted including the time and place of all hearings, the order of presentation of evidence, and all other questions that arise with respect to the course of the reference proceeding. The referee shall render a written statement of decision and shall conduct the proceedings in accordance with the California Code of Civil Procedure, the California Rules of Court, and the California Evidence Code, except as otherwise specifically agreed by the parties and approved by the referee. The referee's statement of decision shall set forth findings of fact and conclusions of law. The referee's decision shall be entered as a judgment in the court in accordance with the provisions of California Code of Civil Procedure §§ 644 and 645, and shall be appealable in accordance with California law.

Nothing in this Agreement shall be deemed to apply to or limit Lender's right to: (i) exercise self-help remedies such as (but not limited to) setoff; (ii) foreclose judicially or nonjudicially against any real or personal property collateral, or to exercise judicial or nonjudicial power of sale rights; (iii) obtain from a court provisional or ancillary remedies (including, without limitation, injunctive relief, a writ of possession, prejudgment attachment, a protective order, or the appointment of a receiver); or (iv) pursue its rights against any Person in a third-party proceeding in any action brought against Lender (including, without limitation, actions in bankruptcy court). Neither the exercise of any self-help remedies nor the institution or maintenance of an action for foreclosure or provisional or ancillary remedies, or the opposition to any such provisional remedies, shall constitute a waiver of the right of any party, including, without limitation, the claimant in any such action, to require submission to judicial reference the merits of the dispute occasioning resort to such remedies. No provision in the Loan Documents regarding submission to jurisdiction or venue

in any court is intended to or shall be construed to be in derogation of the foregoing general judicial reference.

The foregoing judicial reference procedure constitutes a full and complete waiver of the right to a trial by jury that the parties may otherwise have and this waiver and judicial reference procedure is a material consideration to each party hereto.

If the enabling legislation which provides for appointment of a referee is repealed (and no successor statute is enacted), any dispute between the parties that would otherwise be determined by the reference procedure herein described will be resolved and determined by arbitration. The arbitration will be conducted by a retired California state court judge, in accordance with the California Arbitration Act, California Code of Civil Procedure §§ 1280 through 1294.2 as amended from time to time.

[Remainder of page intentionally left blank]

BORROWER:	
	MARATHON DIGITAL HOLDINGS, INC A NEVADA CORPORATION
LENDER:	By: Name: Fred Thiel Title: Chief Executive Officer
	SILVERGATE BANK
	By: Name: Title:

Each of the parties has signed this Agreement as of the day and year first above written.

Signature Page to Term Credit and Security Agreement Ref. No.  $7470\,$ 

### Exhibit A

### Form of Request for Delayed Draw

	Tomi of Request for Delayed Di		
To: SILVER	GATE BANK	[ ], 20[_]	J
To Whom It	May Concern:		
Silvergate Ba	e Term Credit and Security Agreement dated as of ank and Marathon Digital Holdings, Inc., a No we the same meanings in this Request for Delayed	evada corporation. Terms defined in the	
	ection 2.1 of the Agreement, the Borrower hereb 0,000,000.00 on, 20	y requests a Draw from the Lender in the	÷
Borro	ower hereby requests funds of the Delayed Draw s	nould be credited to Borrower as follows:	
	Deposited into Borrower's account #	with Lender; or	
	Wired to Borrower's account as follows:		
	Bank Name:  Bank ABA;  Credit Account Name:  Credit Account Number:		
	Reference:		
The Borrower hereby represents and certifies to the Lender that as of the date of this Draw, no Event of Default exists and each of the conditions to the requested Draw set forth in the Agreement, including Section 8.2, has been satisfied (or, with respect to Sections 8.2(a) and 8.2(b) will be satisfied) or otherwise waived by the Lender.			
		Very truly yours,	
		Marathon Digital Holdings, Inc., a Nevada corporation	
		By:	_
		Name: Title:	

Term Credit and Security Agreement Ref. No. 7470 \$100,000,000.00 July 28, 2022

FOR VALUE RECEIVED, Marathon Digital Holdings, Inc., a Nevada corporation (the "Borrower"), promises to pay to the order of SILVERGATE BANK, California-chartered commercial bank (the "Lender"), on the Maturity Date as provided in that certain Term Credit and Security Agreement dated as of the date hereof (as the same may be amended, supplemented or restated from time to time, the "Loan Agreement"), by and among Borrower and Lender, in lawful money of the United States of America and in immediately available funds, the principal sum of up to ONE HUNDRED MILLION AND NO/100THS DOLLARS (\$100,000,000.00) or, if less, the aggregate unpaid balance amount of all draws made by the Lender to the Borrower under the Loan Agreement (collectively, the "Draws"), together with interest from the date hereof until this Note is fully paid on the principal amount hereunder remaining unpaid from time to time, computed in the manner, and at the rates specified in the Loan Agreement. The principal hereof and interest accruing thereon shall be due and payable as provided in the Loan Agreement.

This Note evidences the Draws. This Note is a Loan Document under the Loan Agreement and is entitled to the benefits and security, and is subject to the terms and conditions, of the Loan Agreement, including, without limitation, acceleration upon the terms provided therein and in the other Loan Documents. All capitalized terms used herein which are defined in the Loan Agreement and not otherwise defined herein shall have the meanings given in the Loan Agreement.

This Note is subject to voluntary and mandatory prepayment, in full or in part, in accordance with, and subject to the terms of, the Loan Agreement. All payments of principal and interest under this Note shall be made in lawful money of the United States of America in immediately available funds at the office of the Lender or at such other place as may be designated by the Lender to the Borrower in writing. Borrower waives any rights pursuant to California Civil Code Sections 1479 and 2822 (and any amendments or successors thereto), to designate how payments will be applied, and acknowledges and agrees that Lender shall have the right in its sole discretion to determine the order and method of the application of payments on this Note or any other Loan Document

Upon the occurrence of an Event of Default, the outstanding principal balance hereunder, together with any accrued but unpaid interest and together with all of the other Obligations, may be accelerated and become immediately due and payable at the option of the Lender and without demand or notice of every kind (which are hereby expressly waived by the Borrower).

The Borrower agrees to pay all costs of collection, including attorneys' fees, all as provided in the Loan Agreement, if this Note is not paid when due, whether or not legal proceedings are commenced.

Presentment or other demand for payment, notice of dishonor and protest are expressly waived.

THIS NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (BUT OTHERWISE WITHOUT REGARD TO CONFLICT OF LAWS PROVISIONS) OF THE STATE OF NEW YORK.

The Borrower irrevocably and unconditionally agrees that it will not commence any action, litigation, or proceeding of any kind or description, whether in law or equity, whether in contract or in tort or otherwise, against the Lender in any way relating to this Note or any other Loan Document or the transactions relating hereto or thereto, in any forum other than the U.S. Federal or California state courts sitting in San Diego County, California or in New York County, New York, and the Borrower irrevocably and unconditionally submits to the jurisdiction of such courts and agrees that all claims in respect of any such action, litigation or proceeding may be heard and determined in such California State or New York State, or, to the extent permitted by law, in such Federal court, pursuant to the Loan Agreement. The Borrower agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Nothing in this Note or any other Loan

NYDIG Ref. No. 7470 Document shall affect any right that the Lender may otherwise have to bring any action or proceeding against the Borrower or its properties in the courts of any other jurisdiction.

The Borrower irrevocably and unconditionally waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any suit, action or proceeding arising out of or relating to this Note or any other Loan Document in any court referred to in the preceding paragraph. The Borrower irrevocably waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court.

The Borrower irrevocably consents to service of process in the manner provided for notices in the Loan Agreement. Nothing in this Note or any other Loan Document will affect the right of the Borrower or the Lender to serve process in any other manner permitted by law.

TO THE FULLEST EXTENT PERMITTED BY LAW, BORROWER HEREBY KNOWINGLY, INTENTIONALLY, AND VOLUNTARILY, WITH AND UPON THE ADVICE OF COMPETENT COUNSEL, AGREES NOT TO ELECT A TRIAL BY JURY OF ANY ISSUE TRIABLE OF RIGHT BY JURY, AND WAIVES ANY RIGHT TO TRIAL BY JURY FULLY TO THE EXTENT THAT ANY SUCH RIGHT SHALL NOW OR HEREAFTER EXIST WITH REGARD TO THIS NOTE, THE SECURITY INSTRUMENT, OR THE OTHER LOAN DOCUMENTS, OR ANY CLAIM, COUNTERCLAIM, OR OTHER ACTION ARISING IN CONNECTION THEREWITH. THIS WAIVER OF RIGHT TO TRIAL BY JURY IS INTENDED TO ENCOMPASS INDIVIDUALLY EACH INSTANCE AND EACH ISSUE AS TO WHICH THE RIGHT TO A TRIAL BY JURY WOULD OTHERWISE ACCRUE. LENDER IS HEREBY AUTHORIZED TO FILE A COPY OF THIS SECTION IN ANY PROCEEDING AS CONCLUSIVE EVIDENCE OF THIS WAIVER BY BORROWER.

Notwithstanding the foregoing to the contrary, in the event that the jury trial waiver contained herein shall be held or deemed to be unenforceable, Borrower hereby agrees that any controversy, dispute, or claim between the parties arising out of or relating to this Note shall be resolved by a reference proceeding in California in accordance with the provisions of California Code of Civil Procedure § 638. The referee shall be a retired California state court judge selected by mutual written agreement of the parties. If the parties are unable to agree upon a referee within ten (10) calendar days after one party serves a written notice of its intent to commence a judicial reference proceeding on the other party, then the referee will be selected by the court in accordance with California Code of Civil Procedure § 640(b). The referee shall be appointed to sit as a temporary judge, with all of the powers of a temporary judge, as authorized by law, and upon selection should take and subscribe to the oath of office as provided for in Rule 244 of the California Rules of Court (or any subsequently enacted Rule). The referee shall determine the manner in which the reference proceeding is conducted including the time and place of all hearings, the order of presentation of evidence, and all other questions that arise with respect to the course of the reference proceeding. The referee shall render a written statement of decision and shall conduct the proceedings in accordance with the California Code of Civil Procedure, the California Rules of Court, and the California Evidence Code, except as otherwise specifically agreed by the parties and approved by the referee. The referee's statement of decision shall set forth findings of fact and conclusions of law. The referee's decision shall be entered as a judgment in the court in accordance with the provisions of California Code of Civil Procedure §§ 644 and 645, and shall be appealable in accordance with California law.

Nothing in this Note shall be deemed to apply to or limit Lender's right to: (i) exercise self-help remedies such as (but not limited to) setoff; (ii) foreclose judicially or nonjudicially against any real or personal property collateral, or to exercise judicial or nonjudicial power of sale rights; (iii) obtain from a court provisional or ancillary remedies (including, without limitation, injunctive relief, a writ of possession, prejudgment attachment, a protective order, or the appointment of a receiver); or (iv) pursue its rights against any Person in a third-party proceeding in any action brought against Lender (including, without limitation, actions in bankruptcy court). Neither the exercise of any self-help remedies nor the institution or maintenance of an action for foreclosure or provisional or ancillary remedies, or the opposition to any such provisional remedies, shall constitute a waiver of the right of any party, including, without limitation, the claimant in any such action, to require submission to judicial reference the merits of the dispute

NYDIG Ref. No. 7470 occasioning resort to such remedies. No provision in the Loan Documents regarding submission to jurisdiction or venue in any court is intended to or shall be construed to be in derogation of the foregoing general judicial reference.

The foregoing judicial reference procedure constitutes a full and complete waiver of the right to a trial by jury that the parties may otherwise have and this waiver and judicial reference procedure is a material consideration to each party hereto.

If the enabling legislation which provides for appointment of a referee is repealed (and no successor statute is enacted), any dispute between the parties that would otherwise be determined by the reference procedure herein described will be resolved and determined by arbitration. The arbitration will be conducted by a retired California state court judge, in accordance with the California Arbitration Act, California Code of Civil Procedure §§ 1280 through 1294.2 as amended from time to time.

Delivery of an executed counterpart of a signature page to this Note by facsimile or in electronic (i.e., "pdf" or "tif") format shall be effective as delivery of a manually executed counterpart of this Note.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its duly authorized officer as of the day and year and at the place set forth above.

MARATHON DIGITAL HOLDINGS, INC., A NEVADA CORPORATION

By:\_\_\_\_\_\_ Name: Fred Thiel

Its: Chief Executive Officer

(Signature page to Term Credit Note)

NYDIG Ref. No. 7470

## CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER AND PRINCIPAL FINANCIAL AND ACCOUNTING OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

#### I, Fred Thiel, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Marathon Digital Holdings, Inc.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal controls over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material
    information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly for the period in
    which this quarterly report is being prepared;
  - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation;
  - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent function):
  - a) all significant deficiencies and material weaknesses in the design or operation of internal controls which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) any fraud, whether material, that involves management or other employees who have a significant role in the registrant's internal controls over financial reporting.

Dated: August 9, 2022 By: /s/ Fred Thiel

Fred Thiel

Chief Executive Officer (Principal Executive Officer)

## CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER AND PRINCIPAL FINANCIAL AND ACCOUNTING OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

#### I, Hugh Gallagher certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Marathon Digital Holdings, Inc.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal controls over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material
    information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly for the period in
    which this quarterly report is being prepared;
  - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation;
  - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent function):
  - a) all significant deficiencies and material weaknesses in the design or operation of internal controls which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) any fraud, whether material, that involves management or other employees who have a significant role in the registrant's internal controls over financial reporting.

Dated: August 9, 2022

By: /s/ Hugh Gallagher

Hugh Gallagher

Chief Financial Officer (Principal Financial and Accounting Officer)

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

About the Quarterly Report of Marathon Digital Holdings, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2022 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Fred Thiel, Chief Executive Officer (Principal Executive Officer) of the Company, certifies, pursuant to 18 U.S.C. section 1350 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 9, 2022 By: /s/ Fred Thiel

Fred Thiel

Chief Executive Officer (Principal Executive Officer)

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Marathon Digital Holdings, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2022 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Hugh Gallagher, Chief Financial Officer, Secretary and Director (Principal Financial and Accounting Officer) of the Company, certifies, pursuant to 18 U.S.C. section 1350 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 9, 2022

By: /s/ Hugh Gallagher

Hugh Gallagher

Chief Financial Officer (Principal Financial and Accounting Officer)

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.